

PURCHASE & SALE AGREEMENTS

CONTRACTS INCLUDED

I. PURCHASE & SALE AGREEMENTS (PSA)

- 1. Offer, Agreement and Escrow Instructions for Purchase of Real Estate
- 2. Purchase and Sale Agreement Comparison Sheet
- 3. Offer, Agreement and Escrow Instructions for Purchase of Real Estate Vacant Land
- 4. Offer, Agreement and Escrow Instructions for Purchase of Residential Income Properties
 - a. 4.1 Chart for Authority
 - b. 4.2 Chart for Contingency

II. PSA ADDENDUM

- 5. Buyer's Notice Disapproving of Contingencies
- 6. Seller's Response to Buyer's Requests
- 7. Buyer's Reply to Seller's Response
- 8. Confidentiality Agreement
- 9. Counter Offer Seller
- 10. Counter Offer Buyer
- 11. Addendum to Purchase Agreement
- 12. Amendment to Purchase and Sale Agreement
- 13. Seller's Mandatory Disclosure Statement
- 14. Property Information Sheet
- 15. Lead-Based Paint Hazards Addendum
- 16. Estoppel Certificate By Lessee
- 17. Estoppel Certificate By Lessor
- 18. Release of Funds Addendum
- 19. Assignment and Assumption of Lessor's Interest in Lease



Dated: ___

STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS FOR PURCHASE OF REAL ESTATE

(Non-Residential)

1.	Buyer.	/"P	ar") haraby offers to	nurchase the real	nronorty horoinaf	for described from the	owner thereof (" Seller ") (colle	ctivaly the "Barties" or
indiv				•		· ·	ction of the Buyer's Continger	• •
Clos and	i ng Date ") conditions	to be held by set forth in this a	("Escrow greement ("Agreem	Holder") whose ace	ldress is have the right to a	, Phone No	, Facsimile No under, but any such assignme	upon the terms
subs	1.2 The equent co	term " Date of Ag unteroffer thereto		erein shall be the d	ate when by exec	• •	fined in paragraph 20.2) of the	
2.	Property	d by both Parties.						
	2.1 The	real property ("P					tion) is located	l in the County of
	2.2 If th	ne legal descriptio	n of the Property is	not complete or is i	naccurate, this Ag	reement shall not be inv	alid and the legal description). shall be completed or
part duct cond	2.3 The of the pro ing, condulitioning ed	Property include: perty, as well as tl its, disconnects, li	s, at no additional control of the following items, it is the following fixtures); tele it is are spring are spring the spring s	ost to Buyer, the pe f any, owned by Sel ephone distribution	rmanent improver ler and at present systems (lines, ja	located on the Property cks and connections only	tter described. those items which pursuant electrical distribution system s	ns (power panel, bus utilating, air
	2.4 The	fire sprinkler mo	nitor: is owned	by Seller and includ	ded in the Purchas	e Price, is leased by	Seller, and Buyer will need to	negotiate a new lease
	2.5 Exc		Paragraph 2.3, the		_	there is no fire sprin 's personal property, fu	kler monitor. Initure and furnishings, and	all of
3.	Purchase	Price.						
		purchase price (" ike any not applic		be paid by Buyer to	Seller for the Pro	perty shall be	, payable as follows:	
	(a)	Cash down payr	ment, including the	Deposit as defined	in paragraph 4.3 (or if an all cash transaction	on, the Purchase Price):	
	(b)	Amount of "Nev	w Loan" as defined i	n paragraph 5.1, if a	any:			
	(c)	Trust") securing	the existing promis	sory note(s) (" Exist i	ing Note(s)"):	ing existing deed(s) of the Closing of approxi	rust ("Existing Deed(s) of mately:	
		until paid	(and/or the entire u	npaid balance is du	e on	erest at the rate of		
		Said Secor	nd Note is payable a	t per	month, including	interest at the rate of	% per annum	
	(d)	Buyer shall give		npaid balance is du st (" Purchase Mon o	e on ey Deed of Trust")	_). on the property, to secu		
		of Buyer to Selle	er described in para	graph 6 (" Purchase	Money Note") in	the amount of:		
Tota	Purchase	Price:						
	es includir	ng, but not limited		ng fees, and apprai	sal fees as a condi		of trust permits the beneficia e Property, Buyer agrees to pa	
4.	Deposits							
	4.1						oe delivered by Broker to Escr	
		1 \		=		-	en delivered to Escrow Holde en delivered to Escrow Holde	
term Agre	inate this ement. Sl	transaction by giv	ring written notice o	f such election to E	scrow Holder whe	eupon neither Party sha	e period then Seller may elec all have any further liability to nall, upon request by Buyer, b	the other under this
Buye	4.2 Add	litional deposits: Within 5 busine	ess days after the Da	te of Agreement, B	uyer shall deposit	with Escrow Holder the	additional sum of	to be applied to the
Purc	hase Price	at the Closing.	•	_			proved or waived, Buyer shall	
Hold	er the add	litional sum of	to be ap	plied to the Purcha	se Price at the Clo	sing.	eller may notify Buyer, Escrov	·
	ng that, ui						ce, the Escrow shall be deeme	
	tered bank	in an interest be	aring account whose	term is appropriat	e and consistent v	vith the timing requirem	collectively the " Deposit "), in a ents of this transaction. The applicable instrument is rede	interest therefrom shall
spec	ified matu		eral Tax Identification				unt cannot be opened until Bu	
_								
	IALS				INIT	IALS		
© 2	U19 AIR (RE. All Rights F	keserved.				Last Edited: 10/	/13/2022 1:44 PM

4.4 Notwithstanding the foregoing, within 5 days after Escrow Holder receives the monies described in paragraph 4.1 above, Escrow Holder shall release \$100 of said monies to Seller as and for independent consideration for Seller's' execution of this Agreement and the granting of the contingency period to Buyer as herein provided. Such independent consideration is non-refundable to Buyer but shall be credited to the Purchase Price in the event that the purchase of the Property is completed. 4.5 Upon waiver of all of Buyer's contingencies the Deposit shall become non-refundable but applicable to the Purchase Price except in the event of a Seller breach, or in the event that the Escrow is terminated pursuant to the provisions of Paragraph 9.1(n) (Destruction, Damage or Loss) or 9.1(o) (Material Change). Financing Contingency. (Strike if not applicable) 5.1 This offer is contingent upon Buyer obtaining from an insurance company, financial institution or other lender, a commitment to lend to Buyer a sum equal % of the Purchase Price, on terms acceptable to Buyer. Such loan ("New Loan") shall be secured by a first deed of trust or mortgage on the to at least Property. If this Agreement provides for Seller to carry back junior financing, then Seller shall have the right to approve the terms of the New Loan. Seller shall have 7 days following receipt of the commitment setting forth the proposed terms of the New Loan to approve or disapprove of such proposed terms. If Seller fails to notify Escrow Holder, in writing, of the disapproval within said 7 days it shall be conclusively presumed that Seller has approved the terms of the New Loan. 5.2 If Buyer shall fail to notify its Broker, Escrow Holder and Seller, in writing within days following the Date of Agreement, that the New Loan has not been obtained, it shall be conclusively presumed that Buyer has either obtained said New Loan or has waived this New Loan contingency. 5.3 If Buyer shall notify its Broker, Escrow Holder and Seller, in writing, within the time specified in paragraph 5.2 hereof, that Buyer has not obtained said New Loan, this Agreement shall be terminated, and Buyer shall be entitled to the prompt return of the Deposit, plus any interest earned thereon, less only Escrow Holder and Title Company cancellation fees and costs, which Buyer shall pay. Seller Financing. (Purchase Money Note). (Strike if not applicable) 6.1 If Seller approves Buyer's financials (see paragraph 6.5) the Purchase Money Note shall provide for interest on unpaid principal at the rate of % per annum, with principal and interest paid as follows: _. The Purchase Money Note and Purchase Money Deed of Trust shall be on the current forms nmonly used by Escrow Holder, and be junior and subordinate only to the Existing Note(s) and/or the New Loan expressly called for by this Agreement 6.2 The Purchase Money Note and/or the Purchase Money Deed of Trust shall contain provisions regarding the following (see also paragraph 10.3 (b)) Prepayment. Principal may be prepaid in whole or in part at any time without penalty, at the option of the Buyer. Late Charge. A late charge of 6% shall be payable with respect to any payment of principal, interest, or other charges, not made within 10 days after (b) it is due. Due On Sale. In the event the Buyer sells or transfers title to the Property or any portion thereof, then the Seller may, at Seller's option, require the (c) entire unpaid balance of said Note to be paid in full. 6.3 If the Purchase Money Deed of Trust is to be subordinate to other financing, Escrow Holder shall, at Buyer's expense prepare and record on Seller's behalf a request for notice of default and/or sale with regard to each mortgage or deed of trust to which it will be subordinate. 6.4 WARNING: CALIFORNIA LAW DOES NOT ALLOW DEFICIENCY JUDGEMENTS ON SELLER FINANCING. IF BUYER ULTIMATELY DEFAULTS ON THE LOAN, SELLER'S SOLE REMEDY IS TO FORECLOSE ON THE PROPERTY. 6.5 Seller's obligation to provide financing is contingent upon Seller's reasonable approval of Buyer's financial condition. Buyer to provide a current financial statement and copies of its Federal tax returns for the last 3 years to Seller within 10 days following the Date of Agreement. Seller has 10 days following receipt of such documentation to satisfy itself with regard to Buyer's financial condition and to notify Escrow Holder as to whether or not Buyer's financial condition is acceptable. If Seller fails to notify Escrow Holder, in writing, of the disapproval of this contingency within said time period, it shall be conclusively presumed that Seller has approved Buyer's financial condition. If Seller is not satisfied with Buyer's financial condition or if Buyer fails to deliver the required documentation then Seller may notify Escrow Holder in writing that Seller Financing will not be available, and Buyer shall have the option, within 10 days of the receipt of such notice, to either terminate this transaction or to purchase the Property without Seller financing. If Buyer fails to notify Escrow Holder within said time period of its election to terminate this transaction then Buyer shall be conclusively presumed to have elected to purchase the Property without Seller financing. If Buyer elects to terminate, Buyer's Deposit shall be refunded less Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. 7.1 Each Party acknowledges receiving a Disclosure Regarding Real Estate Agency Relationship, confirms and consents to the following agency relationships in this transaction with the following real estate broker(s) ("Brokers") and/or their agents ("Agent(s)"): is the broker of (check one): Lathe Seller; or both the Buyer and Seller (dual agent). License No. Seller's Brokerage Firm License No. the Seller's Agent (salesperson or broker associate); or both the Seller's Agent is (check one): Seller's Agent and the Buyer's Agent (dual agent). is the broker of (check one): $\ \ \ \ \$ the Buyer; or $\ \ \ \ \$ both the Buyer and Seller (dual agent). Buyer's Brokerage Firm License No. the Buyer's Agent (salesperson or broker associate); or 🔲 both the Buyer's Agent is (check one): Buyer's Agent License No. and the Seller's Agent (dual agent). The Parties acknowledge that other than the Brokers and Agents listed above, there are no other brokers or agents representing the Parties or due any fees and/or commissions under this Agreement. Buyer shall use the services of Buyer's Broker exclusively in connection with any and all negotiations and offers with respect to the Property for a period of 1 year from the date inserted for reference purposes at the top of page 1. 7.2 Buyer and Seller each represent and warrant to the other that he/she/it has had no dealings with any person, firm, broker, agent or finder in connection with the negotiation of this Agreement and/or the consummation of the purchase and sale contemplated herein, other than the Brokers and Agents named in paragraph 7.1, and no broker, agent or other person, firm or entity, other than said Brokers and Agents is/are entitled to any commission or finder's fee in connection with this transaction as the result of any dealings or acts of such Party. Buyer and Seller do each hereby agree to indemnify, defend, protect and hold the other harmless from and against any costs, expenses or liability for compensation, commission or charges which may be claimed by any broker, agent, finder or other similar party, other than said named Brokers and Agents by reason of any dealings or act of the indemnifying Party. 8.1 Upon acceptance hereof by Seller, this Agreement, including any counteroffers incorporated herein by the Parties, shall constitute not only the agreement of purchase and sale between Buyer and Seller, but also instructions to Escrow Holder for the consummation of the Agreement through the Escrow. Escrow Holder shall not prepare any further escrow instructions restating or amending the Agreement unless specifically so instructed by the Parties or a Broker herein. Subject to the reasonable approval of the Parties, Escrow Holder may, however, include its standard general escrow provisions. In the event that there is any conflict between the provisions of the Agreement and the provisions of any additional escrow instructions the provisions of the Agreement shall prevail as to the Parties and the Escrow Holden 8.2 As soon as practical after the receipt of this Agreement and any relevant counteroffers, Escrow Holder shall ascertain the Date of Agreement as defined in paragraphs 1.2 and 20.2 and advise the Parties and Brokers, in writing, of the date ascertained. Escrow Holder is hereby authorized and instructed to conduct the Escrow in accordance with this Agreement, applicable law and custom and practice of the community in which Escrow Holder is located, including any reporting requirements of the Internal Revenue Code. In the event of a conflict between the law of the state where the Property is located and the law of the state where the Escrow Holder is located, the law of the state where the Property is located shall prevail. 8.4 Subject to satisfaction of the contingencies herein described, Escrow Holder shall close this escrow (the "Closing") by recording a general warranty deed (a grant deed in California) and the other documents required to be recorded, and by disbursing the funds and documents in accordance with this Agreement. 8.5 Buyer and Seller shall each pay one-half of the Escrow Holder's charges and Seller shall pay the usual recording fees and any required documentary transfer taxes. Seller shall pay the premium for a standard coverage owner's or joint protection policy of title insurance. (See also paragraph 11.)

taxes. Seller shall pay the premium for a standard coverage owner's or joint protection policy of title insurance. (See also paragraph 11.)

8.6 Escrow Holder shall verify that all of Buyer's contingencies have been satisfied or waived prior to Closing. The matters contained in paragraphs 9.1 subparagraphs (b), (c), (d), (e), (g), (i), (n), and (o), 9.4, 12, 13, 14, 16, 18, 20, 21, 22, and 24 are, however, matters of agreement between the Parties only and are not

subparagraphs (b), (c), (d), (e), (g), (i), (n), and (o), 9.4, 12, 13, 14, 16, 18, 20, 21, 22, and 24 are, however, matters of agreement between the Parties only and are not instructions to Escrow Holder.

8.7 If this transaction is terminated for non-satisfaction and non-waiver of a Buyer's Contingency, as defined in paragraph 9.2 or disapproval of any other matter subject to Buyer's approval, then neither of the Parties shall thereafter have any liability to the other under this Agreement, except to the extent of a breach of any affirmative covenant or warranty in this Agreement. In the event of such termination, Buyer shall, subject to the provisions of paragraph 8.10, be promptly refunded all funds deposited by Buyer with Escrow Holder, less only the \$100 provided for in paragraph 4.4 and the Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. If this transaction is terminated as a result of Seller's breach of this Agreement then Seller shall pay the Title Company and Escrow Holder cancellation fees and costs.

8.8 The Closing shall occur on the Expected Closing Date, or as soon thereafter as the Escrow is in condition for Closing; provided, however, that if the Closing

INITIALS

© 2019 AIR CRE. All Rights Reserved.

does not occur by the Expected Closing Date and said Date is not extended by mutual instructions of the Parties, a Party not then in default under this Agreement may notify the other Party, Escrow Holder, and Brokers, in writing that, unless the Closing occurs within 5 business days following said notice, the Escrow shall be deemed terminated without further notice or instructions.

8.9 Except as otherwise provided herein, the termination of Escrow shall not relieve or release either Party from any obligation to pay Escrow Holder's fees and costs or constitute a waiver, release or discharge of any breach or default that has occurred in the performance of the obligations, agreements, covenants or warranties contained therein.

8.10 If this Escrow is terminated for any reason other than Seller's breach or default, then as a condition to the return of Buyer's deposit, Buyer shall within 5 days after written request deliver to Seller, at no charge, copies of all surveys, engineering studies, soil reports, maps, master plans, feasibility studies and other similar items prepared by or for Buyer that pertain to the Property.

9. Contingencies to Closing.

9.1 IF, BEFORE EXPIRATION OF THE APPLICABLE TIME, BUYER FAILS TO PROVIDE ESCROW HOLDER WRITTEN NOTICE OF BUYER'S DISAPPROVAL OF ANY OF
BUYER'S CONTINGENCIES OR ANY OTHER MATTER THAT IS SUBJECT TO BUYER'S APPROVAL IN THIS AGREEMENT, THEN BUYER SHALL BE CONCLUSIVELY DEEMED TO HAVE SATISFIED SUCH BUYER'S CONTINGENCIES AND/OR APPROVED OF SUCH OTHER MATTERS. If a number of days is completed in any of the optional spaces
in subparagraphs 9.1 (a) through (m), then such number shall apply and override the pre-printed number, even if the pre-printed number is not stricken. The Closing
of this transaction is contingent upon the satisfaction or waiver of the following contingencies: (a) Disclosure. Seller shall make to Buyer, through Escrow, all of the applicable disclosures required by law (See AIR CRE ("AIR") standard form entitled "Seller's
Mandatory Disclosure Statement") and provide Buyer with a completed Property Information Sheet ("Property Information Sheet") concerning the Property, duly
executed by or on behalf of Seller in the current form or equivalent to that published by the AIR within 10 or days following the Date of Agreement.
Buyer has 10 days from the receipt of said disclosures to approve or disapprove the matters disclosed.
(b) Physical Inspection. Buyer has 10 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever
is later, to satisfy itself with regard to the physical aspects and size of the Property.
(c) Hazardous Substance Conditions Report. Buyer has 30 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever is later, to satisfy itself with regard to the environmental aspects of the Property. Seller recommends that Buyer obtain a Hazardous Substance
Conditions Report concerning the Property and relevant adjoining properties. Any such report shall be paid for by Buyer. A "Hazardous Substance" for purposes of
this Agreement is defined as any substance whose nature and/or quantity of existence, use, manufacture, disposal or effect, render it subject to Federal, state or local
regulation, investigation, remediation or removal as potentially injurious to public health or welfare. A "Hazardous Substance Condition" for purposes of this
Agreement is defined as the existence on, under or relevantly adjacent to the Property of a Hazardous Substance that would require remediation and/or removal
under applicable Federal, state or local law. (d) Soil Inspection. Buyer has 30 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever is
later, to satisfy itself with regard to the condition of the soils on the Property. Seller recommends that Buyer obtain a soil test report. Any such report shall be paid for
by Buyer. Seller shall provide Buyer copies of any soils report that Seller may have within 10 days following the Date of Agreement.
(e) Governmental Approvals. Buyer has 30 or days following the Date of Agreement to satisfy itself with regard to approvals and permits
from governmental agencies or departments which have or may have jurisdiction over the Property and which Buyer deems necessary or desirable in connection with
its intended use of the Property, including, but not limited to, permits and approvals required with respect to zoning, planning, building and safety, fire, police,
handicapped and Americans with Disabilities Act requirements, transportation and environmental matters. (f) Conditions of Title. Escrow Holder shall cause a current commitment for title insurance ("Title Commitment") concerning the Property issued by the
Title Company, as well as legible copies of all documents referred to in the Title Commitment (" Underlying Documents "), and a scaled and dimensioned plot showing
the location of any easements to be delivered to Buyer within 10 or days following the Date of Agreement. Buyer has 10 days from the receipt of the
Title Commitment, the Underlying Documents and the plot plan to satisfy itself with regard to the condition of title. The disapproval by Buyer of any monetary
encumbrance, which by the terms of this Agreement is not to remain against the Property after the Closing, shall not be considered a failure of this contingency, as
Seller shall have the obligation, at Seller's expense, to satisfy and remove such disapproved monetary encumbrance at or before the Closing. (g) Survey. Buyer has 30 or days following the receipt of the Title Commitment and Underlying Documents to satisfy itself with regard to
any ALTA title supplement based upon a survey prepared to American Land Title Association ("ALTA") standards for an owner's policy by a licensed surveyor, showing
the legal description and boundary lines of the Property, any easements of record, and any improvements, poles, structures and things located within 10 feet of either
side of the Property boundary lines. Any such survey shall be prepared at Buyer's direction and expense. If Buyer has obtained a survey and approved the ALTA title
supplement, Buyer may elect within the period allowed for Buyer's approval of a survey to have an ALTA extended coverage owner's form of title policy, in which even
Buyer shall pay any additional premium attributable thereto.
(h) Existing Leases and Tenancy Statements. Seller shall within 10 or days following the Date of Agreement provide both Buyer and Escrow
Holder with legible copies of all leases, subleases or rental arrangements (collectively, "Existing Leases") affecting the Property, and with a tenancy statement ("Estoppel Certificate") in the latest form or equivalent to that published by the AIR, executed by Seller and/or each tenant and subtenant of the Property. Seller shall
use its best efforts to have each tenant complete and execute an Estoppel Certificate. If any tenant fails or refuses to provide an Estoppel Certificate then Seller shall
complete and execute an Estoppel Certificate for that tenancy. Buyer has 10 days from the receipt of said Existing Leases and Estoppel Certificates to satisfy itself with
regard to the Existing Leases and any other tenancy issues.
(i) Owner's Association. Seller shall within 10 or days following the Date of Agreement provide Buyer with a statement and transfer package from any owner's association servicing the Property. Such transfer package shall at a minimum include: copies of the association's bylaws, articles of
incorporation, current budget and financial statement. Buyer has 10 days from the receipt of such documents to satisfy itself with regard to the association.
(j) Other Agreements. Seller shall within 10 or days following the Date of Agreement provide Buyer with legible copies of all other
agreements ("Other Agreements") known to Seller that will affect the Property after Closing. Buyer has 10 days from the receipt of said Other Agreements to satisfy
itself with regard to such Agreements.
(k) Financing. If paragraph 5 hereof dealing with a financing contingency has not been stricken, the satisfaction or waiver of such New Loan contingency.
(I) Existing Notes. If paragraph 3.1(c) has not been stricken, Seller shall within 10 or days following the Date of Agreement provide Buyer with legible copies of the Existing Notes, Existing Deeds of Trust and related agreements (collectively, "Loan Documents") to which the Property will remain subject
after the Closing. Escrow Holder shall promptly request from the holders of the Existing Notes a beneficiary statement ("Beneficiary Statement") confirming: (1) the
amount of the unpaid principal balance, the current interest rate, and the date to which interest is paid, and (2) the nature and amount of any impounds held by the
beneficiary in connection with such loan. Buyer has 10 or days following the receipt of the Loan Documents and Beneficiary Statements to satisfy itself
with regard to such financing. Buyer's obligation to close is conditioned upon Buyer being able to purchase the Property without acceleration or change in the terms
of any Existing Notes or charges to Buyer except as otherwise provided in this Agreement or approved by Buyer, provided, however, Buyer shall pay the transfer fee
referred to in paragraph 3.2 hereof. Likewise if Seller is to carry back a Purchase Money Note then Seller shall within 10 or days following the Date of
Agreement provide Buyer with a copy of the proposed Purchase Money Note and Purchase Money Deed of Trust. Buyer has 10 or days following the receipt of such documents to satisfy itself with regard to the form and content thereof.
(m) Personal Property. In the event that any personal property is included in the Purchase Price, Buyer has 10 or days following the Date of
Agreement to satisfy itself with regard to the title condition of such personal property. Seller recommends that Buyer obtain a UCC-1 report. Any such report shall be
paid for by Buyer. Seller shall provide Buyer copies of any liens or encumbrances affecting such personal property that it is aware of within 10 or days
following the Date of Agreement.
(n) Destruction, Damage or Loss. Subsequent to the Date of Agreement and prior to Closing there shall not have occurred a destruction of, or damage or
loss to, the Property or any portion thereof, from any cause whatsoever, which would cost more than \$10,000.00 to repair or cure. If the cost of repair or cure is \$10,000.00 or less, Seller shall repair or cure the loss prior to the Closing. Buyer shall have the option, within 10 days after receipt of written notice of a loss costing
more than \$10,000.00 to repair or cure, to either terminate this Agreement or to purchase the Property notwithstanding such loss, but without deduction or offset
against the Purchase Price. If the cost to repair or cure is more than \$10,000.00, and Buyer does not elect to terminate this Agreement, Buyer shall be entitled to any
insurance proceeds applicable to such loss. Unless otherwise notified in writing, Escrow Holder shall assume no such destruction, damage or loss has occurred prior
to Closing. (a) Material Change, Buyer shall have 10 days following receipt of written notice of a Material Change within which to satisfy itself with regard to such
(o) Material Change. Buyer shall have 10 days following receipt of written notice of a Material Change within which to satisfy itself with regard to such change. "Material Change" shall mean a substantial adverse change in the use, occupancy, tenants, title, or condition of the Property that occurs after the date of this
offer and prior to the Closing. Unless otherwise notified in writing, Escrow Holder shall assume that no Material Change has occurred prior to the Closing.
(p) Seller Performance. The delivery of all documents and the due performance by Seller of each and every undertaking and agreement to be performed
by Seller under this Agreement.
INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
OFA-20.30, Revised 10-13-2022

- (q) Brokerage Fee. Payment at the Closing of such brokerage fee as is specified in this Agreement or later written instructions to Escrow Holder executed by Seller and Brokers ("Brokerage Fee"). It is agreed by the Parties and Escrow Holder that Brokers are a third party beneficiary of this Agreement insofar as the Brokerage Fee is concerned, and that no change shall be made with respect to the payment of the Brokerage Fee specified in this Agreement, without the written consent of Brokers.
- 9.2 The contingencies specified in subparagraphs 9.1(a) through (m) are for the benefit of, and may be waived by, Buyer, and are referred to collectively as "Buyer's Contingencies" and individually as a "Buyer's Contingency."
- 9.3 Buyer's timely and written disapproval or conditional approval of a Buyer's Contingency or any other matter that is subject to Buyer's approval in this Agreement shall constitute disapproval thereof ("Disapproved Item(s)"). Concurrent with notice of a Disapproved Item, Buyer may make a request to Seller regarding such Disapproved Item ("Buyer's Request"). If Buyer fails to make a timely and written Buyer's Request, then this Agreement shall terminate due to the non-satisfaction and non-waiver of a contingency. Seller may respond to a Buyer's Request within 10 days following Seller's receipt thereof ("Seller's Response"). Seller's acceptance of a Buyer's Request shall amend this Agreement accordingly. If Seller fails to provide a timely and written Seller's Response, then Seller's Response shall be deemed to be a rejection of Buyer's Request. Buyer may, within 10 days following the earlier of Buyer's receipt of a Seller's Response (which is not an acceptance of Buyer's Request) or the date of Seller's deemed rejection of a Buyer's Request ("Buyer's Reply Period"), reply to a Seller's Response ("Buyer's Reply") and elect to (i) terminate this Agreement due to the non-satisfaction and non-waiver of the applicable contingency, (ii) accept the Seller's Response in which event this Agreement shall be amended accordingly, or (iii) withdraw Buyer's Request and waive the Disapproved Item in which event Buyer shall accept the Property subject to the Disapproved Item. If Buyer fails to provide a timely and written Buyer's Reply, then Buyer shall be deemed to have elected to terminate this Agreement as of the end of the Buyer's Reply Period. The date Buyer accepts a Seller's Response or withdraws a Buyer's Request and waives a Disapproved Item shall be the date of Buyer's approval of the Disapproved Item. A Party shall provide to Escrow Holder copy of all notices of a Disapproved Item, Buyer's Request, Seller's Response and Buyer's Reply and Escrow Holder shall promptly provide copies thereof to the other Party. Unless the Parties in writing agree otherwise, if the Expected Closing Date is a specific calendar date and a Buyer's Reply Period expires after such specific calendar date, then notwithstanding paragraph 1.1, the Expected Closing Date shall be extended to be 3 business days after the earlier of the date Buyer withdraws a Buyer's Request and waives the applicable Disapproved Item or Buyer accepts the applicable Seller's Response.
- 9.4 The Parties acknowledge that extensive local, state and Federal legislation establish broad liability upon owners and/or users of real property for the investigation and remediation of Hazardous Substances. The determination of the existence of a Hazardous Substance Condition and the evaluation of the impact of such a condition are highly technical and beyond the expertise of Brokers. The Parties acknowledge that they have been advised by Brokers to consult their own technical and legal experts with respect to the possible presence of Hazardous Substances on the Property or adjoining properties, and Buyer and Seller are not relying upon any investigation by or statement of Brokers with respect thereto. The Parties hereby assume all responsibility for the impact of such Hazardous Substances upon their respective interests herein.

10. Documents and Other Items Required at or Before Closing.

- 10.1 Five days prior to the Closing date Escrow Holder shall obtain an updated Title Commitment concerning the Property from the Title Company and provide copies thereof to each of the Parties.
 - 10.2 Seller shall deliver to Escrow Holder in time for delivery to Buyer at the Closing:
 - (a) Grant or general warranty deed, duly executed and in recordable form, conveying fee title to the Property to Buyer.
 - (b) If applicable, the Beneficiary Statements concerning Existing Note(s).
- (c) If applicable, the Existing Leases and Other Agreements together with duly executed assignments thereof by Seller and Buyer. The assignment of Existing Leases shall be on the most recent Assignment and Assumption of Lessor's Interest in Lease form published by the AIR or its equivalent.
- (d) An affidavit executed by Seller to the effect that Seller is not a "foreign person" within the meaning of Internal Revenue Code Section 1445 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Internal Revenue Service such sum as is required by applicable Federal law with respect to purchases from foreign sellers.
- (e) If the Property is located in California, an affidavit executed by Seller to the effect that Seller is not a "nonresident" within the meaning of California Revenue and Tax Code Section 18662 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Franchise Tax Board such sum as is required by such statute.
 - (f) If applicable, a bill of sale, duly executed, conveying title to any included personal property to Buyer.
 - (g) If the Seller is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the sale of the Property.

 10.3 Buyer shall deliver to Seller through Escrow:
- (a) The cash portion of the Purchase Price and such additional sums as are required of Buyer under this Agreement shall be deposited by Buyer with Escrow Holder, by federal funds wire transfer, or any other method acceptable to Escrow Holder in immediately collectable funds, no later than 2:00 P.M. on the business day prior to the Expected Closing Date provided, however, that Buyer shall not be required to deposit such monies into Escrow if at the time set for the deposit of such monies Seller is in default or has indicated that it will not perform any of its obligations hereunder. Instead, in such circumstances in order to reserve its rights to proceed Buyer need only provide Escrow with evidence establishing that the required monies were available.
- (b) If a Purchase Money Note and Purchase Money Deed of Trust are called for by this Agreement, the duly executed originals of those documents, the Purchase Money Deed of Trust being in recordable form, together with evidence of fire insurance on the improvements in the amount of the full replacement cost naming Seller as a mortgage loss payee, and a real estate tax service contract (at Buyer's expense), assuring Seller of notice of the status of payment of real property taxes during the life of the Purchase Money Note.
 - (c) The Assignment and Assumption of Lessor's Interest in Lease form specified in paragraph 10.2(c) above, duly executed by Buyer.
 - (d) Assumptions duly executed by Buyer of the obligations of Seller that accrue after Closing under any Other Agreements.
 - (e) If applicable, a written assumption duly executed by Buyer of the loan documents with respect to Existing Notes.
 - (f) If the Buyer is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the purchase of the Property.
- 10.4 At Closing, Escrow Holder shall cause to be issued to Buyer a standard coverage (or ALTA extended, if elected pursuant to 9.1(g)) owner's form policy of title insurance effective as of the Closing, issued by the Title Company in the full amount of the Purchase Price, insuring title to the Property vested in Buyer, subject only to the exceptions approved by Buyer. In the event there is a Purchase Money Deed of Trust in this transaction, the policy of title insurance shall be a joint protection policy insuring both Buyer and Seller.

IMPORTANT: IN A PURCHASE OR EXCHANGE OF REAL PROPERTY, IT MAY BE ADVISABLE TO OBTAIN TITLE INSURANCE IN CONNECTION WITH THE CLOSE OF ESCROW SINCE THERE MAY BE PRIOR RECORDED LIENS AND ENCUMBRANCES WHICH AFFECT YOUR INTEREST IN THE PROPERTY BEING ACQUIRED. A NEW POLICY OF TITLE INSURANCE SHOULD BE OBTAINED IN ORDER TO ENSURE YOUR INTEREST IN THE PROPERTY THAT YOU ARE ACQUIRING.

11. Prorations and Adjustments.

- 11.1 Taxes. Applicable real property taxes and special assessment bonds shall be prorated through Escrow as of the date of the Closing, based upon the latest tax bill available. The Parties agree to prorate as of the Closing any taxes assessed against the Property by supplemental bill levied by reason of events occurring prior to the Closing. Payment of the prorated amount shall be made promptly in cash upon receipt of a copy of any supplemental bill.
- 11.2 Insurance. WARNING: Any insurance which Seller may have maintained will terminate on the Closing. Buyer is advised to obtain appropriate insurance to cover the Property.
- 11.3 Rentals, Interest and Expenses. Scheduled rentals, interest on Existing Notes, utilities, and operating expenses shall be prorated as of the date of Closing. The Parties agree to promptly adjust between themselves outside of Escrow any rents received after the Closing.
 - 11.4 Security Deposit. Security Deposits held by Seller shall be given to Buyer as a credit to the cash required of Buyer at the Closing.
- 11.5 Post Closing Matters. Any item to be prorated that is not determined or determinable at the Closing shall be promptly adjusted by the Parties by appropriate cash payment outside of the Escrow when the amount due is determined.
- 11.6 Variations in Existing Note Balances. In the event that Buyer is purchasing the Property subject to an Existing Deed of Trust(s), and in the event that a Beneficiary Statement as to the applicable Existing Note(s) discloses that the unpaid principal balance of such Existing Note(s) at the closing will be more or less than the amount set forth in paragraph 3.1(c) hereof ("Existing Note Variation"), then the Purchase Money Note(s) shall be reduced or increased by an amount equal to such Existing Note Variation. If there is to be no Purchase Money Note, the cash required at the Closing per paragraph 3.1(a) shall be reduced or increased by the amount of such Existing Note Variation.
- 11.7 Variations in New Loan Balance. In the event Buyer is obtaining a New Loan and the amount ultimately obtained exceeds the amount set forth in paragraph 5.1, then the amount of the Purchase Money Note, if any, shall be reduced by the amount of such excess.
- 11.8 Owner's Association Fees. Escrow Holder shall: (i) bring Seller's account with the association current and pay any delinquencies or transfer fees from Seller's proceeds, and (ii) pay any up front fees required by the association from Buyer's funds.

12. Representations and Warranties of Seller and Disclaimers.

12.1 Seller's warranties and representations shall survive the Closing and delivery of the deed for a period of 3 years, and any lawsuit or action based upon them

INITIALS

© 2019 AIR CRE. All Rights Reserved.

must be commenced within such time period. Seller's warranties and representations are true, material and relied upon by Buyer and Brokers in all respects. Seller hereby makes the following warranties and representations to Buyer and Brokers:

- (a) Authority of Seller. Seller is the owner of the Property and/or has the full right, power and authority to sell, convey and transfer the Property to Buyer as provided herein, and to perform Seller's obligations hereunder.
- (b) Maintenance During Escrow and Equipment Condition At Closing. Except as otherwise provided in paragraph 9.1(n) hereof, Seller shall maintain the Property until the Closing in its present condition, ordinary wear and tear excepted.
- (c) Hazardous Substances/Storage Tanks. Seller has no knowledge, except as otherwise disclosed to Buyer in writing, of the existence or prior existence on the Property of any Hazardous Substance, nor of the existence or prior existence of any above or below ground storage tank.
- (d) Compliance. Except as otherwise disclosed in writing, Seller has no knowledge of any aspect or condition of the Property which violates applicable laws, rules, regulations, codes or covenants, conditions or restrictions, or of improvements or alterations made to the Property without a permit where one was required, or of any unfulfilled order or directive of any applicable governmental agency or casualty insurance company requiring any investigation, remediation, repair, maintenance or improvement be performed on the Property.
- (e) Changes in Agreements. Prior to the Closing, Seller will not violate or modify any Existing Lease or Other Agreement, or create any new leases or other agreements affecting the Property, without Buyer's written approval, which approval will not be unreasonably withheld.
- (f) Possessory Rights. Seller has no knowledge that anyone will, at the Closing, have any right to possession of the Property, except as disclosed by this Agreement or otherwise in writing to Buyer.
 - (g) Mechanics' Liens. There are no unsatisfied mechanics' or materialmens' lien rights concerning the Property.
- (h) Actions, Suits or Proceedings. Seller has no knowledge of any actions, suits or proceedings pending or threatened before any commission, board, bureau, agency, arbitrator, court or tribunal that would affect the Property or the right to occupy or utilize same.
- (i) Notice of Changes. Seller will promptly notify Buyer and Brokers in writing of any Material Change (see paragraph 9.1(o)) affecting the Property that becomes known to Seller prior to the Closing.
- (j) No Tenant Bankruptcy Proceedings. Seller has no notice or knowledge that any tenant of the Property is the subject of a bankruptcy or insolvency proceeding.
 - (k) No Seller Bankruptcy Proceedings. Seller is not the subject of a bankruptcy, insolvency or probate proceeding.
- (I) Personal Property. Seller has no knowledge that anyone will, at the Closing, have any right to possession of any personal property included in the Purchase Price nor knowledge of any liens or encumbrances affecting such personal property, except as disclosed by this Agreement or otherwise in writing to Buyer.
- 12.2 Buyer hereby acknowledges that, except as otherwise stated in this Agreement, Buyer is purchasing the Property in its existing condition and will, by the time called for herein, make or have waived all inspections of the Property Buyer believes are necessary to protect its own interest in, and its contemplated use of, the Property. The Parties acknowledge that, except as otherwise stated in this Agreement, no representations, inducements, promises, agreements, assurances, oral or written, concerning the Property, or any aspect of the occupational safety and health laws, Hazardous Substance laws, or any other act, ordinance or law, have been made by either Party or Brokers, or relied upon by either Party hereto.
- 12.3 In the event that Buyer learns that a Seller representation or warranty might be untrue prior to the Closing, and Buyer elects to purchase the Property anyway then, and in that event, Buyer waives any right that it may have to bring an action or proceeding against Seller or Brokers, regarding said representation or warranty.
- 12.4 Any environmental reports, soils reports, surveys, and other similar documents which were prepared by third party consultants and provided to Buyer by Seller or Seller's representatives, have been delivered as an accommodation to Buyer and without any representation or warranty as to the sufficiency, accuracy, completeness, and/or validity of said documents, all of which Buyer relies on at its own risk. Seller believes said documents to be accurate, but Buyer is advised to retain appropriate consultants to review said documents and investigate the Property.

13. Possession.

Possession of the Property shall be given to Buyer at the Closing subject to the rights of tenants under Existing Leases.

14. Buyer's Entry

At any time during the Escrow period, Buyer, and its agents and representatives, shall have the right at reasonable times and subject to rights of tenants, to enter upon the Property for the purpose of making inspections and tests specified in this Agreement. No destructive testing shall be conducted, however, without Seller's prior approval which shall not be unreasonably withheld. Following any such entry or work, unless otherwise directed in writing by Seller, Buyer shall return the Property to the condition it was in prior to such entry or work, including the re-compaction or removal of any disrupted soil or material as Seller may reasonably direct. All such inspections and tests and any other work conducted or materials furnished with respect to the Property by or for Buyer shall be paid for by Buyer as and when due and Buyer shall indemnify, defend, protect and hold harmless Seller and the Property of and from any and all claims, liabilities, losses, expenses (including reasonable attorneys' fees), damages, including those for injury to person or property, arising out of or relating to any such work or materials or the acts or omissions of Buyer, its agents or employees in connection therewith.

15. Further Documents and Assurances.

The Parties shall each, diligently and in good faith, undertake all actions and procedures reasonably required to place the Escrow in condition for Closing as and when required by this Agreement. The Parties agree to provide all further information, and to execute and deliver all further documents, reasonably required by Escrow Holder or the Title Company.

16. Attorneys' Fees.

If any Party or Broker brings an action or proceeding (including arbitration) involving the Property whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees and costs. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred.

17. Prior Agreements/Amendments.

- 17.1 This Agreement supersedes any and all prior agreements between Seller and Buyer regarding the Property.
- 17.2 Amendments to this Agreement are effective only if made in writing and executed by Buyer and Seller.

18. Broker's Rights.

18.1 If this sale is not consummated due to the default of either the Buyer or Seller, the defaulting Party shall be liable to and shall pay to Brokers the Brokerage Fee that Brokers would have received had the sale been consummated. If Buyer is the defaulting party, payment of said Brokerage Fee is in addition to any obligation with respect to liquidated or other damages.

18.2 Upon the Closing, Brokers are authorized to publicize the facts of this transaction.

19. Notices.

19.1 Whenever any Party, Escrow Holder or Brokers herein shall desire to give or serve any notice, demand, request, approval, disapproval or other communication, each such communication shall be in writing and shall be delivered personally, by messenger, or by mail, postage prepaid, to the address set forth in this agreement or by facsimile transmission, electronic signature, digital signature, or email.

19.2 Service of any such communication shall be deemed made on the date of actual receipt if personally delivered, or transmitted by facsimile transmission, electronic signature, digital signature, or email. Any such communication sent by regular mail shall be deemed given 48 hours after the same is mailed. Communications sent by United States Express Mail or overnight courier that guarantee next day delivery shall be deemed delivered 24 hours after delivery of the same to the Postal Service or courier. If such communication is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.

19.3 Any Party or Broker hereto may from time to time, by notice in writing, designate a different address to which, or a different person or additional persons to whom, all communications are thereafter to be made.

tion of Offer.
tion of Offer

20.1 If this offer is not accepted by Seller on or before 5:00 P.M. according to the time standard applicable to the city of	on the date of
, it shall be deemed automatically revoked.	

20.2 The acceptance of this offer, or of any subsequent counteroffer hereto, that creates an agreement between the Parties as described in paragraph 1.2, shall be deemed made upon delivery to the other Party or either Broker herein of a duly executed writing unconditionally accepting the last outstanding offer or counteroffer.

INITIALS
© 2019 AIR CRE. All Rights Reserved.

21. LIQUIDATED DAMAGES. (This Liquidated Damages paragraph is applicable only if initialed by both Parties). THE PARTIES AGREE THAT IT WOULD BE IMPRACTICABLE OR EXTREMELY DIFFICULT TO FIX, PRIOR TO SIGNING THIS AGREEMENT, THE ACTUAL DAMAGES WHICH WOULD BE SUFFERED BY SELLER IF BUYER FAILS TO PERFORM ITS OBLIGATIONS UNDER THIS AGREEMENT. THEREFORE, IF, AFTER THE SATISFACTION OR WAIVER OF ALL CONTINGENCIES PROVIDED FOR THE BUYER'S BENEFIT, BUYER BREACHES THIS AGREEMENT, SELLER SHALL BE ENTITLED TO LIQUIDATED DAMAGES IN THE AMOUNT OF UPON PAYMENT OF SAID SUM TO SELLER, BUYER SHALL BE RELEASED FROM ANY FURTHER LIABILITY TO SELLER, AND ANY ESCROW CANCELLATION FEES AND TITLE COMPANY CHARGES SHALL BE PAID BY SELLER. BUYER'S InitialS Seller'S InitialS						
22.1 ANY CONTROVERSY AS SHALL BE DETERMINED BY BINDIN ITS COMMERCIAL ARBITRATION RUSUCH CONTROVERSY SHALL BE AR EXPERIENCE IN THE TYPE OF REAL ACCORDANCE WITH APPLICABLE LAGREEMENT AND ANY AMENDME PERMITTED IN ACCORDANCE WITH AWARD WITHIN 30 DAYS AFTER THE PARAGRAPH 16 HEREOF AND SHAIL OF THE DEPOSITS FOR ARBITRATO CROSS-EXAMINE WITNESSES, BUT AND LEGAL ARGUMENT AS THE AR COMPETENT JURISDICTION NOTW 22.2 BUYER'S RESORT TO OR BUYER FOR DAMAGES AND/OR SPEWHICH EVENT SUCH AWARD SHALL 22.3 NOTICE: BY INITIALING IN OF DISPUTES" PROVISION DECIDED HAVE THE DISPUTE LITIGATED IN A APPEAL, UNLESS SUCH RIGHTS ARE AGREEING TO THIS PROVISION, 10 TO THIS ARBITRATION PROVISION IN	22. ARBITRATION OF DISPUTES. (This Arbitration of Disputes paragraph is applicable only if initialed by both Parties.) 22.1 ANY CONTROVERSY AS TO WHETHER SELLER IS ENTITLED TO LIQUIDATED DAMAGES AND/OR BUYER IS ENTITLED TO THE RETURN OF THE DEPOSIT SHALL BE DETERMINED BY BINDING ARBITRATION ADMINISTERED BY THE JUDICIAL ARBITRATION & MEDIATION SERVICES, INC. ("JAMS") IN ACCORDANCE WITH ITS COMMERCIAL ARBITRATION RULES ("COMMERCIAL RULES"). ARBITRATION HEARINGS SHALL BE HELD IN THE COUNTY WHERE THE PROPERTY IS LOCATED. SUCH CONTROVERSY SHALL BE ARBITRATED BY A SINGLE ARBITRATOR, APPOINTED UNDER THE COMMERCIAL RULES WHO HAS HAD AT LEAST 5 YEARS OF EXPERIENCE IN THE TYPE OF REAL ESTATE THAT IS THE SUBJECT OF THIS AGREEMENT. THE ARBITRATOR SHALL HEAR AND DETERMINE SAID CONTROVERSY IN ACCORDANCE WITH APPLICABLE LAW OF THE JURISDICTION WHERE THE PROPERTY IS LOCATED, THE INTENTION OF THE PARTIES AS EXPRESSED IN THIS AGREEMENT AND ANY AMENDMENTS THERETO, AND UPON THE EVIDENCE PRODUCED AT AN ARBITRATION HEARING. PRE-ARBITRATION DISCOVERY SHALL BE PERMITTED IN ACCORDANCE WITH THE COMMERCIAL RULES OR STATE LAW APPLICABLE TO ARBITRATION PROCEEDINGS. THE ARBITRATION SHALL RENDER AN AWARD WITHIN 30 DAYS AFTER THE CONCLUSION OF THE HEARING, WHICH MAY INCLUDE ATTORNEYS' FEES AND COSTS TO THE PREVAILING PARTY PER PARAGRAPH 16 HEREOF AND SHALL BE ACCOMPANIED BY A REASONED OPINION. THE FAILURE OR REFUSAL OF A PARTY TO PAY SUCH PARTY'S REQUIRED SHARE OF THE DEPOSITS FOR ARBITRATOR COMPENSATION OR ADMINISTRATIVE CHARGES SHALL CONSTITUTE A WAIVER BY SUCH PARTY TO PRESENT EVIDENCE OR CROSS-EXAMINE WITNESSES, BUT SUCH WAIVER SHALL NOT ALLOW FOR A DEFAULT JUDGMENT MAY BE ENTERED ON THE AWARD IN ANY COURT OF COMPETENT JURISDICTION NOTWITHSTANDING THE FAILURE OF A PARTY DULY NOTHIFIED OF THE ARBITRATION HEARING TO APPEAR THEREAT. 22.2 BUYER'S RESORT TO OR PARTICIPATION IN SUCH ARBITRATION PROCEEDINGS SHALL NOT BAR SUIT IN A COURT OF COMPETENT JURISDICTION BY THE BUYER FOR DAMAGES AND/OR SPECIFIC PERFORMANCE. 22.3 NOTICE: BY INITIALING					
WE HAVE READ AND UNDERSTAND PROVISION TO NEUTRAL ARBITRATI	THE FOREGOING AND AGREE TO SUBMIT D ON.	ISPUTES ARISING OUT OF THE M.	ATTERS INCLUDED IN THE "ARBIT	RATION OF DISPUTES"		
		A	~			
	Buyer's Initials	26	ller's Initials			
Paragraphs 21 and 22 are each inco Agreement accomplished by means 23.2 Applicable Law. This Agreement accomplished by means 23.2 Applicable Law. This Agreement are of Essence. Time in 23.4 Counterparts. This Agreement is combined the signed signature paragraph 23.5 Waiver of Jury Trial. THE PROPERTY OR ARISING OUT OF THE 23.6 Conflict. Any conflict be typewritten or handwritten provision 23.7 1031 Exchange. Both Separty initiating an exchange shall be exchanging Party in the event that the state of the sta	eement shall be binding on the Parties with reporated into this Agreement only if initiale to of electronic signature or similar technology reement shall be governed by, and paragrag in the Parties hereto concerning this Agreement of the essence of this Agreement. The ement may be executed by Buyer and Sellentstrument. Escrow Holder, after verifying the ges on one of the counterparts, which shall the parties Hereby WAIVE THEIR RESPECTIVES AGREEMENT. The tween the printed provisions of this Agreements. Seller and Buyer must initial any and a seller and costs of such exchange. The cooperate she sale is delayed and/or that the sale other specifically indicated to the contrary, the workship in the sale of the sale is delayed and/or that the sale of the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed indicated to the contrary, the workship is the sale is delayed and/or that the sale of the sale is delayed and/or that the sale of the sale is delayed and in the contrary, the workship is the sale is delayed and in the contrary, the workship is the sale is delayed and in the contrary in the sale of the sale is delayed and in the contrary in the sale of the sale is delayed and in the contrary in the sale of the sale is delayed and in the sale of the sale is delayed and in the sale of the sale of the sale is delayed and in the sale of the sale is delayed and in the sale of the sale is delayed and in the sale of the sale is delayed and in the sale of the sale is delayed and in the sale of the sale is delayed and in the sale of the sale is delayed and in the sale of the sale is delayed and in the sale of the s	d by both Parties at the time that sy shall be legal and binding. The shall be legal and binding. The shall be initiated in the countries of t	the Agreement is executed. Signal laws of the state in which the Property is locate shall be deemed an original, and except for the signatures, is authory ACTION OR PROCEEDING IN written provisions shall be controlled by the control of the signal with the provisions shall be controlled by the control of t	Property is located. d. all of which together corized and instructed VOLVING THE blled by the B1 exchange. Any chages to the		
	sture of a Real Estate Agency Relationship.	unad butha principles set forth i	n the applicable coefficies of the C	alifornia Civil Codo ac		
summarized in paragraph 24.2. 24.2 When entering into a disof agency relationship or represent transaction, as follows: (a) Seller's Agent. A S following affirmative obligations: (1 Seller: a. Diligent exercise of reason all facts known to the agent material Parties. An agent is not obligated to	agree that their relationship(s) shall be gove scussion with a real estate agent regarding a ation it has with the agent or agents in the teller's agent under a listing agreement with) To the Seller: A fiduciary duty of utmost canable skills and care in performance of the a ally affecting the value or desirability of the preveal to either Party any confidential info	real estate transaction, a Buyer ransaction. Buyer and Seller ack the Seller acts as the agent for the re, integrity, honesty, and loyalty gent's duties. b. A duty of hones property that are not known to, or	or Seller should from the outset of nowledge being advised by the Bone Seller only. A Seller's agent or in dealings with the Seller. (2) Total and fair dealing and good faith. Or within the diligent attention ar	understand what type rokers in this subagent has the othe Buyer and the c. A duty to disclose and observation of, the		
agent, even if by agreement the age the following affirmative obligation the Seller: a. Diligent exercise of rea disclose all facts known to the agen observation of, the Parties. An age affirmative duties set forth above. (c) Agent Representing	elling agent can, with a Buyer's consent, agrent may receive compensation for services r. (1) To the Buyer: A fiduciary duty of utmosonable skills and care in performance of the materially affecting the value or desirability is not obligated to reveal to either Party and Both Seller and Buyer. A real estate agent yer in a transaction, but only with the know	endered, either in full or in part is st care, integrity, honesty, and lo se agent's duties. b. A duty of ho sy of the property that are not kn my confidential information obtain, either acting directly or through	from the Seller. An agent acting of yalty in dealings with the Buyer. nest and fair dealing and good fa own to, or within the diligent atte ined from the other Party which one or more associate licensees.	only for a Buyer has (2) To the Buyer and ith. c. A duty to ention and does not involve the , can legally be the		
agent has the following affirmative either Seller or the Buyer. b. Other both Seller and Buyer, the agent manot limited to, facts relating to eithe including Seller's willingness to according to the self-buyer.	obligations to both the Seller and the Buyer duties to the Seller and the Buyer aduties to the Seller and the Buyer as stated by not, without the express permission of the Buyer's or Seller's financial position, motiept a price less than the listing price or Buyer on trelieve a Seller or Buyer from the response	: a. A fiduciary duty of utmost ca above in their respective sections e respective Party, disclose to the vations, bargaining position, or o r's willingness to pay a price grea	re, integrity, honesty and loyalty s (a) or (b) of this paragraph 24.2. e other Party confidential informa ther personal information that m ater than the price offered. (3) Th	in the dealings with (2) In representing ation, including, but ay impact price, ne above duties of the		

INITIALS

agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Buyer has the duty to exercise reasonable care to protect Buyer, including as to those facts about the Property which are known to Buyer or within Buyer's diligent attention and observation. Both Seller and Buyer should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.

- (d) Further Disclosures. Throughout this transaction Buyer and Seller may receive more than one disclosure, depending upon the number of agents assisting in the transaction. Buyer and Seller should each read its contents each time it is presented, considering the relationship between them and the real estate agent in this transaction and that disclosure. Buyer and Seller each acknowledge receipt of a disclosure of the possibility of multiple representation by the Broker representing that principal. This disclosure may be part of a listing agreement, buyer representation agreement or separate document. Buyer understands that Broker representing Buyer may also represent other potential buyers, who may consider, make offers on or ultimately acquire the Property. Seller understands that Broker representing Seller may also represent other sellers with competing properties that may be of interest to this Buyer. Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this transaction may be brought against Broker more than one year after the Date of Agreement and that the liability (including court costs and attorneys' fees), of any Broker with respect to any breach of duty, error or omission relating to this Agreement shall not exceed the fee received by such Broker pursuant to this Agreement; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such
- 24.3 Confidential Information. Buyer and Seller agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- Construction of Agreement. In construing this Agreement, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Agreement. Whenever required by the context, the singular shall include the plural and vice versa. This Agreement shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.

26. Additional Provisions.	
Additional provisions of this offer, if any, are as follows or are attached hereto by an addendum or addenda consisting of paragraphs	through
(If there are no additional provisions write "NONE".)	

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS AGREEMENT OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS AGREEMENT.
- RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PROPERTY. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PROPERTY, THE INTEGRITY AND CONDITION OF ANY STRUCTURES AND OPERATING SYSTEMS, AND THE SUITABILITY OF THE PROPERTY FOR BUYER'S INTENDED USE.

WARNING: IF THE PROPERTY IS LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THIS AGREEMENT MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PROPERTY IS LOCATED.

NOTE:

- THIS FORM IS NOT FOR USE IN CONNECTION WITH THE SALE OF RESIDENTIAL PROPERTY.

2. IF EITHER PARTY IS A CORPORATION, IT IS RECOMMENDED THAT THIS A	AGREEMENT BE SIGNED BY TWO CORPORATE OFFICERS.
The undersigned Buyer offers and agrees to buy the Property on the terms and con	ditions stated and acknowledges receipt of a copy hereof.
•	Date
BROKER	BUYER
Attn:	Dec
Title:	By: Name Printed:
Address:	Title:
Phone:	Phone: Fax:
Fax:	Email:
Email: Federal ID No.:	
Broker DRE License #:	Ву:
Agent DRE License #:	Name Printed:
	Title:
	Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
27. Acceptance.	
	grees to sell the Property to Buyer on the terms and conditions therein specified.
27.2 In consideration of real estate brokerage service rendered by Brokers, Se	Seller's Broker % and Buyer's Broker %. This
Agreement shall serve as an irrevocable instruction to Escrow Holder to pay such Br	·
Closing.	σ · · · · · · · · · · · · · · · · · · ·
27.3 Seller acknowledges receipt of a copy hereof and authorizes Brokers to o	
NOTE: A PROPERTY INFORMATION SHEET IS REQUIRED TO BE DELIVERED TO BUYE	ER BY SELLER UNDER THIS AGREEMENT.
	Date:
BROKER	SELLER
	
Attn:	By:
Title:	Name Printed:
Address:	

INITIALS

© 2019 AIR CRE. All Rights Reserved. OFA-20.30, Revised 10-13-2022

INITIALS

Dhana	Title.
Phone: Fax:	Title: Phone:
Email:	Fax:
Federal ID No.:	Email:
Broker DRE License #:	
Agent's DRE License #:	By: Name Printed:
	Title:
	Phone:
	Fax: Email:
	Email:
	Address:
	Federal ID No.:
	AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.



STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS

FOR PURCHASE OF REAL ESTATE

The following is a list of some of the major differences between the various purchase agreements published by AIR CRE.

The 'Standard' Purchase Agreement v. Purchase Agreement for Vacant Land

Header changed from (Non-Residential) to (Vacant Land)

- **2.3** Eliminates all references to mechanical systems and fixtures leaving only a generic reference to "improvements" in place.
- **2.4** Eliminates the reference to fire sprinklers.
- **3.1** Gives two choices for determining price: 1. Fixed price or 2. Price per unit, such as price per square foot.
- **9.1e** Adds a warning that past uses may no longer be allowed. Buyer's responsibility to rezone and Seller must cooperate.
- **12.4** Feasibility studies has been added to the list of studies/reports that 3rd parties prepare and are passed from Seller to Buyer.

The 'Standard' Purchase Agreement v. Purchase Agreement for Residential Income Properties

Header changed from (Non-Residential) to (Do Not Use for Properties Containing Less Than Five Units)

- **2.3** List of equipment has been modified to reflect differences between commercial and residential properties.
- **9.1a** Lead based paint has been added as part of the mandatory disclosures in certain circumstances.

- **9.1b** A termite inspection/report to be paid for by Buyer has been added.
- **9.1i** Discusses Seller providing Buyer with statement and transfer package in the event that an Owner's Association exists.
- **10.2i** Adds a requirement that Seller provide the Buyer with all keys to the Property.
- **10.4** Again discusses keys.
- **23.8** Adds a provision regarding the fact that the Property is being sold in compliance with Equal Housing Opportunity laws.

Warning #2 Modifies the standard warning so as to also discuss Rent Control and Termites.

Note #1 This form is NOT to be used with properties containing less than 5 units. Residential transactions involving 1-4 units is governed by a different set of laws. As a consequence, we decided to avoid those issues and let CAR deal with them.

NOTE: The above is not intended to be a complete list of all of the differences between the various purchase agreements.



STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS FOR PURCHASE OF REAL ESTATE

(Vacant Land)

Dated:	
1. Buyer.	reinafter described from the owner thereof ("Caller") (collectively the "Darties" o
1.1, ("Buyer") hereby offers to purchase the real property, he individually, a "Party"), through an escrow ("Escrow") to close 30 or ("Escrow Holder") whose address is ("Escrow Holder") whose address is	
and conditions set forth in this agreement ("Agreement"). Buyer shall have the righ	
Buyer of Buyer's obligations herein unless Seller expressly releases Buyer. 1.2 The term "Date of Agreement" as used herein shall be the date when by	
subsequent counteroffer thereto, Buyer and Seller have reached agreement in writiterms accepted by both Parties.	, , , , ,
2. Property.	
2.1 The real property ("Property") that is the subject of this offer consists of	
, is commonly known as (street address, city, state, zip)	and is legally described as: (APN:
corrected to meet the requirements of ("Title Company"), which sha	Il issue the title policy hereinafter described.
2.3 The Property includes, at no additional cost to Buyer, the permanent impart of the property, as well as the following items, if any, owned by Seller and at property.	provements thereon, including those items which pursuant to applicable law are a resent located on the Property: (collectively, the "Improvements")
2.4 Except as provided in Paragraph 2.3, the Purchase Price does not include	
which shall be removed by Seller prior to Closing.	
3. Purchase Price.	
3.1 The purchase price (" Purchase Price ") to be paid by Buyer to Seller for the will be determined based on a per unit cost instead of a fixed price)	
	re number of units shall be based on a calculation of total area of the Property as
certified to the Parties by a licensed surveyor in accordance with paragraph 9.1(g). It calculation: . The Purchase Price shall be payable as follows:	
(Strike any not applicable)	
(a) Cash down payment, including the Deposit as defined in paragraph	4.3 (or if an all cash transaction, the Purchase Price):
(b) Amount of "New Loan" as defined in paragraph 5.1, if any:	
(c) Buyer shall take title to the Property subject to and/or assume the	following existing deed(s) of trust ("Existing Deed(s) of
Trust") securing the existing promissory note(s) ("Existing Note(s)" (i) An Existing Note ("First Note") with an unpaid principal balan):
Said First Note is payable at per month, includi	ng interest at the rate of % per annum
until paid (and/or the entire unpaid balance is due on	
(ii) An Existing Note ("Second Note") with an unpaid principal ba	lance as of the Closing of approximately:
Said Second Note is payable at per month, incl	uding interest at the rate of% per
annum until paid (and/or the entire unpaid balance is due on).
(d) Buyer shall give Seller a deed of trust ("Purchase Money Deed of T of Buyer to Seller described in paragraph 6 ("Purchase Money Note	
3.2 If Buyer is taking title to the Property subject to, or assuming, an Existing of fees including, but not limited to, points, processing fees, and appraisal fees as a maximum of 1.5% of the unpaid principal balance of the applicable Existing Note.	Deed of Trust and such deed of trust permits the beneficiary to demand paymen condition to the transfer of the Property, Buyer agrees to pay such fees up to a
4. Deposits.	
	payable to Escrow Holder, to be delivered by Broker to Escrow Holder within 2 or
	ne executed Agreement has been delivered to Escrow Holder, or within 2 or
business days after both Parties have executed this Agreement and the	ne executed Agreement has been delivered to Escrow Holder Buyer shall deliver to
Escrow Holder a check in the sum of If said check is not received by terminate this transaction by giving written notice of such election to Escrow Holde	
Agreement. Should Buyer and Seller not enter into an agreement for purchase and	
Buyer. 4.2 Additional deposits:	
	posit with Escrow Holder the additional sum of to be applied to the
Purchase Price at the Closing. (b) Within 5 husiness days after the contingencies discussed in paragram	aph 9.1 (a) through (m) are approved or waived, Buyer shall deposit with Escrow
Holder the additional sum of to be applied to the Purchase Price at the	he Closing.
(c) If an Additional Deposit is not received by Escrow Holder within the writing that, unless the Additional Deposit is received by Escrow Holder within 2 bu.	e time period provided then Seller may notify Buyer, Escrow Holder, and Brokers, siness days following said notice, the Escrow shall be deemed terminated without
further notice or instructions.	•
chartered bank in an interest bearing account whose term is appropriate and consist	_ ·
accrue to the benefit of Buyer, who hereby acknowledges that there may be penalti specified maturity. Buyer's Federal Tax Identification Number is NO	
Identification Number is provided.	
4.4 Notwithstanding the foregoing, within 5 days after Escrow Holder received	es the monies described in paragraph 4.1 above, Escrow Holder shall release \$100
	
INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	Last Edited: 10/13/2022 1:44 PM

of said monies to Seller as and for independent consideration for Seller's' execution of this Agreement and the granting of the contingency period to Buyer as herein provided. Such independent consideration is non-refundable to Buyer but shall be credited to the Purchase Price in the event that the purchase of the Property is completed.

completed.
4.5 Upon waiver of all of Buyer's contingencies the Deposit shall become non-refundable but applicable to the Purchase Price except in the event of a Seller breach, or in the event that the Escrow is terminated pursuant to the provisions of Paragraph 9.1(n) (Destruction, Damage or Loss) or 9.1(o) (Material Change).

5. Financing Contingency. (Strike if not applicable)

- 5.2 If Buyer shall fail to notify its Broker, Escrow Holder and Seller, in writing within ______ days following the Date of Agreement, that the New Loan has not been obtained, it shall be conclusively presumed that Buyer has either obtained said New Loan or has waived this New Loan contingency.
- 5.3 If Buyer shall notify its Broker, Escrow Holder and Seller, in writing, within the time specified in paragraph 5.2 hereof, that Buyer has not obtained said New Loan, this Agreement shall be terminated, and Buyer shall be entitled to the prompt return of the Deposit, plus any interest earned thereon, less only Escrow Holder and Title Company cancellation fees and costs, which Buyer shall pay.

6. Seller Financing. (Purchase Money Note). (Strike if not applicable)

- 6.1 If Seller approves Buyer's financials (see paragraph 6.5) the Purchase Money Note shall provide for interest on unpaid principal at the rate of _________. The Purchase Money Note and Purchase Money Deed of Trust shall be on the current forms commonly used by Escrow Holder, and be junior and subordinate only to the Existing Note(s) and/or the New Loan expressly called for by this Agreement.
 - 6.2 The Purchase Money Note and/or the Purchase Money Deed of Trust shall contain provisions regarding the following (see also paragraph 10.3 (b)):
 - (a) Prepayment. Principal may be prepaid in whole or in part at any time without penalty, at the option of the Buyer.
- (b) Late Charge. A late charge of 6% shall be payable with respect to any payment of principal, interest, or other charges, not made within 10 days after it is due.
- (c) Due On Sale. In the event the Buyer sells or transfers title to the Property or any portion thereof, then the Seller may, at Seller's option, require the entire unpaid balance of said Note to be paid in full.
- 6.3 If the Purchase Money Deed of Trust is to be subordinate to other financing, Escrow Holder shall, at Buyer's expense prepare and record on Seller's behalf a request for notice of default and/or sale with regard to each mortgage or deed of trust to which it will be subordinate.
- 6.4 WARNING: CALIFORNIA LAW DOES NOT ALLOW DEFICIENCY JUDGEMENTS ON SELLER FINANCING. IF BUYER ULTIMATELY DEFAULTS ON THE LOAN, SELLER'S SOLE REMEDY IS TO FORECLOSE ON THE PROPERTY.
- 6.5 Seller's obligation to provide financing is contingent upon Seller's reasonable approval of Buyer's financial condition. Buyer to provide a current financial statement and copies of its Federal tax returns for the last 3 years to Seller within 10 days following the Date of Agreement. Seller has 10 days following receipt of such documentation to satisfy itself with regard to Buyer's financial condition and to notify Escrow Holder as to whether or not Buyer's financial condition is acceptable. If Seller fails to notify Escrow Holder, in writing, of the disapproval of this contingency within said time period, it shall be conclusively presumed that Seller has approved Buyer's financial condition. If Seller is not satisfied with Buyer's financial condition or if Buyer fails to deliver the required documentation then Seller may notify Escrow Holder in writing that Seller Financing will not be available, and Buyer shall have the option, within 10 days of the receipt of such notice, to either terminate this transaction or to purchase the Property without Seller financing. If Buyer fails to notify Escrow Holder within said time period of its election to terminate this transaction then Buyer shall be conclusively presumed to have elected to purchase the Property without Seller financing. If Buyer elects to terminate, Buyer's Deposit shall be refunded less Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation.

7. Real Estate Brokers.

	_	Regarding Real Estate Agency Relationship, confirms and consents to the following agency relationships in
this transaction with the following	g real estate broker(s) (" Br	okers") and/or their agents ("Agent(s)"):
Seller's Brokerage Firm	License No.	is the broker of (check one): the Seller; or both the Buyer and Seller (dual agent).
Seller's Agent	License No.	is (check one): the Seller's Agent (salesperson or broker associate); or both the Seller's Agent
and the Buyer's Agent (dual agent	·).	
Buyer's Brokerage Firm	License No.	is the broker of (check one): the Buyer; or both the Buyer and Seller (dual agent).
Buyer's Agent	License No.	_ is (check one): — the Buyer's Agent (salesperson or broker associate); or — both the Buyer's Agent
and the Seller's Agent (dual agent).	
_		gents listed above, there are no other brokers or agents representing the Parties or due any fees and/or
commissions under this Agreemen	nt. Buyer shall use the se	vices of Buyer's Broker exclusively in connection with any and all negotiations and offers with respect to
the Property for a period of 1 year	r from the date inserted for	or reference purposes at the top of page 1.

7.2 Buyer and Seller each represent and warrant to the other that he/she/it has had no dealings with any person, firm, broker, agent or finder in connection with the negotiation of this Agreement and/or the consummation of the purchase and sale contemplated herein, other than the Brokers and Agents named in paragraph 7.1, and no broker, agent or other person, firm or entity, other than said Brokers and Agents is/are entitled to any commission or finder's fee in connection with this transaction as the result of any dealings or acts of such Party. Buyer and Seller do each hereby agree to indemnify, defend, protect and hold the other harmless from and against any costs, expenses or liability for compensation, commission or charges which may be claimed by any broker, agent, finder or other similar party, other than said named Brokers and Agents by reason of any dealings or act of the indemnifying Party.

8. Escrow and Closing.

- 8.1 Upon acceptance hereof by Seller, this Agreement, including any counteroffers incorporated herein by the Parties, shall constitute not only the agreement of purchase and sale between Buyer and Seller, but also instructions to Escrow Holder for the consummation of the Agreement through the Escrow. Escrow Holder shall not prepare any further escrow instructions restating or amending the Agreement unless specifically so instructed by the Parties or a Broker herein. Subject to the reasonable approval of the Parties, Escrow Holder may, however, include its standard general escrow provisions. In the event that there is any conflict between the provisions of the Agreement and the provisions of any additional escrow instructions the provisions of the Agreement shall prevail as to the Parties and the Escrow Holder.
- 8.2 As soon as practical after the receipt of this Agreement and any relevant counteroffers, Escrow Holder shall ascertain the Date of Agreement as defined in paragraphs 1.2 and 20.2 and advise the Parties and Brokers, in writing, of the date ascertained.
- 8.3 Escrow Holder is hereby authorized and instructed to conduct the Escrow in accordance with this Agreement, applicable law and custom and practice of the community in which Escrow Holder is located, including any reporting requirements of the Internal Revenue Code. In the event of a conflict between the law of the state where the Property is located and the law of the state where the Property is located shall prevail.
- 8.4 Subject to satisfaction of the contingencies herein described, Escrow Holder shall close this escrow (the "Closing") by recording a general warranty deed (a grant deed in California) and the other documents required to be recorded, and by disbursing the funds and documents in accordance with this Agreement.
- 8.5 Buyer and Seller shall each pay one-half of the Escrow Holder's charges and Seller shall pay the usual recording fees and any required documentary transfer taxes. Seller shall pay the premium for a standard coverage owner's or joint protection policy of title insurance. (See also paragraph 11.)
- 8.6 Escrow Holder shall verify that all of Buyer's contingencies have been satisfied or waived prior to Closing. The matters contained in paragraphs 9.1 subparagraphs (b), (c), (d), (e), (g), (i), (n), and (o), 9.4, 12, 13, 14, 16, 18, 20, 21, 22, and 24 are, however, matters of agreement between the Parties only and are not instructions to Escrow Holder.
- 8.7 If this transaction is terminated for non-satisfaction and non-waiver of a Buyer's Contingency, as defined in Paragraph 9.2 or disapproval of any other matter subject to Buyer's approval, then neither of the Parties shall thereafter have any liability to the other under this Agreement, except to the extent of a breach of any affirmative covenant or warranty in this Agreement. In the event of such termination, Buyer shall, subject to the provisions of paragraph 8.10, be promptly refunded all funds deposited by Buyer with Escrow Holder, less only the \$100 provided for in paragraph 4.4 and the Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. If this transaction is terminated as a result of Seller's breach of this Agreement then Seller shall pay the Title Company and Escrow Holder cancellation fees and costs.
- 8.8 The Closing shall occur on the Expected Closing Date, or as soon thereafter as the Escrow is in condition for Closing; provided, however, that if the Closing does not occur by the Expected Closing Date and said Date is not extended by mutual instructions of the Parties, a Party not then in default under this Agreement may

INITIALS

notify the other Party, Escrow Holder, and Brokers, in writing that, unless the Closing occurs within 5 business days following said notice, the Escrow shall be deemed terminated without further notice or instructions.

8.9 Except as otherwise provided herein, the termination of Escrow shall not relieve or release either Party from any obligation to pay Escrow Holder's fees and costs or constitute a waiver, release or discharge of any breach or default that has occurred in the performance of the obligations, agreements, covenants or warranties contained therein.

8.10 If this Escrow is terminated for any reason other than Seller's breach or default, then as a condition to the return of Buyer's deposit, Buyer shall within 5 days after written request deliver to Seller, at no charge, copies of all surveys, engineering studies, soil reports, maps, master plans, feasibility studies and other similar items prepared by or for Buyer that pertain to the Property.

9.1 IF, BEFORE EXPIRATION OF THE APPLICABLE TIME, BUYER FAILS TO PROVIDE ESCROW HOLDER WRITTEN NOTICE OF BUYER'S DISAPPROVAL OF ANY OF

9. Contingencies to Closing.

BUYER'S CONTINGENCIES OR ANY OTHER MATTER THAT IS SUBJECT TO BUYER'S APPROVAL II TO HAVE SATISFIED SUCH BUYER'S CONTINGENCIES AND/OR APPROVED OF SUCH OTHER MA	•
in subparagraphs 9.1 (a) through (m), then such number shall apply and override the pre-printe	
of this transaction is contingent upon the satisfaction or waiver of the following contingencies:	at number, even if the pre-printed number is necessioned.
(a) Disclosure. Seller shall make to Buyer, through Escrow, all of the applicable dis	sclosures required by law (See AIR CRE ("AIR") standard form entitled
"Seller's Mandatory Disclosure Statement") and provide Buyer with a completed Property Info	
duly executed by or on behalf of Seller in the current form or equivalent to that published by the	
Agreement. Buyer has 10 days from the receipt of said disclosures to approve or disapprove the	
(b) Physical Inspection. Buyer has 10 or days following the receipt	of the Property Information Sheet or the Date of Agreement, whichever
is later, to satisfy itself with regard to the physical aspects and size of the Property. (c) Hazardous Substance Conditions Report. Buyer has 30 or days f	following the receipt of the Droporty Information Shoot or the Date of
Agreement, whichever is later, to satisfy itself with regard to the environmental aspects of the Conditions Report concerning the Property and relevant adjoining properties. Any such report this Agreement is defined as any substance whose nature and/or quantity of existence, use, may regulation, investigation, remediation or removal as potentially injurious to public health or we Agreement is defined as the existence on, under or relevantly adjacent to the Property of a Hazunder applicable Federal, state or local law. (d) Soil Inspection. Buyer has 30 or days following the receipt of the later, to satisfy itself with regard to the condition of the soils on the Property. Seller recommendations are considered as any such as a self-self with regard to the condition of the soils on the Property.	Property. Seller recommends that Buyer obtain a Hazardous Substance shall be paid for by Buyer. A "Hazardous Substance" for purposes of anufacture, disposal or effect, render it subject to Federal, state or local effect. A "Hazardous Substance Condition" for purposes of this zardous Substance that would require remediation and/or removal e Property Information Sheet or the Date of Agreement, whichever is
by Buyer. Seller shall provide Buyer copies of any soils report that Seller may have within 10 da (e) Governmental Approvals. Buyer has 30 or days following the Days	lys following the Date of Agreement. ate of Agreement to satisfy itself with regard to approvals and permits
from governmental agencies or departments which have or may have jurisdiction over the Propits intended use of the Property, including, but not limited to, permits and approvals required valuacion handicapped and Americans with Disabilities Act requirements, transportation and environment	with respect to zoning, planning, building and safety, fire, police, ntal matters.
NOTE: Past uses of the Property may no longer be allowed. In the event that the Property mus appropriate government agencies. Seller shall sign all documents Buyer is required to file in co development approvals.	nnection with rezoning, conditional use permits and/or other
(f) Conditions of Title. Escrow Holder shall cause a current commitment for title i Title Company, as well as legible copies of all documents referred to in the Title Commitment ('the location of any easements to be delivered to Buyer within 10 or days following Title Commitment, the Underlying Documents and the plot plan to satisfy itself with regard to the encumbrance, which by the terms of this Agreement is not to remain against the Property after Seller shall have the obligation, at Seller's expense, to satisfy and remove such disapproved mo	"Underlying Documents"), and a scaled and dimensioned plot showing ng the Date of Agreement. Buyer has 10 days from the receipt of the the condition of title. The disapproval by Buyer of any monetary r the Closing, shall not be considered a failure of this contingency, as
(g) Survey. Buyer has 30 or days following the receipt of the Title C	
any ALTA title supplement based upon a survey prepared to American Land Title Association ("A	
the legal description and boundary lines of the Property, any easements of record, and any imposide of the Property boundary lines. Any such survey shall be prepared at Buyer's direction and	d expense. If Buyer has obtained a survey and approved the ALTA title
supplement, Buyer may elect within the period allowed for Buyer's approval of a survey to have Buyer shall pay any additional premium attributable thereto.	e an ALIA extended coverage owner's form of title policy, in which event
(h) Existing Leases and Tenancy Statements. Seller shall within 10 or	days following the Date of Agreement provide both Ruyer and Escrow
Holder with legible copies of all leases, subleases or rental arrangements (collectively, "Existing ("Estoppel Certificate") in the latest form or equivalent to that published by the AIR, executed use its best efforts to have each tenant complete and execute an Estoppel Certificate. If any te complete and execute an Estoppel Certificate for that tenancy. Buyer has 10 days from the recregard to the Existing Leases and any other tenancy issues.	s Leases") affecting the Property, and with a tenancy statement by Seller and/or each tenant and subtenant of the Property. Seller shall nant fails or refuses to provide an Estoppel Certificate then Seller shall eipt of said Existing Leases and Estoppel Certificates to satisfy itself with
(i) Owner's Association. Seller shall within 10 or days following the	
package from any owner's association servicing the Property. Such transfer package shall at a rincorporation, current budget and financial statement. Buyer has 10 days from the receipt of s (j) Other Agreements. Seller shall within 10 or days following the lagreements ("Other Agreements") known to Seller that will affect the Property after Closing. Extractly with regard to such Agreements.	uch documents to satisfy itself with regard to the association. Date of Agreement provide Buyer with legible copies of all other Buyer has 10 days from the receipt of said Other Agreements to satisfy
 (k) Financing. If paragraph 5 hereof dealing with a financing contingency has not (l) Existing Notes. If paragraph 3.1(c) has not been stricken, Seller shall within 10 	- · · · · · · · · · · · · · · · · · · ·
with legible copies of the Existing Notes, Existing Deeds of Trust and related agreements (collection after the Closing. Escrow Holder shall promptly request from the holders of the Existing Notes amount of the unpaid principal balance, the current interest rate, and the date to which interest beneficiary in connection with such loan. Buyer has 10 or days following the rection with regard to such financing. Buyer's obligation to close is conditioned upon Buyer being able of any Existing Notes or charges to Buyer except as otherwise provided in this Agreement or appreferred to in paragraph 3.2 hereof. Likewise if Seller is to carry back a Purchase Money Note to Agreement provide Buyer with a copy of the proposed Purchase Money Note and Purchase Note Purchase Note Purchase Note Purchase Note Purchase Note	ctively, "Loan Documents") to which the Property will remain subject a beneficiary statement ("Beneficiary Statement") confirming: (1) the st is paid, and (2) the nature and amount of any impounds held by the seipt of the Loan Documents and Beneficiary Statements to satisfy itself to purchase the Property without acceleration or change in the terms approved by Buyer, provided, however, Buyer shall pay the transfer fee then Seller shall within 10 or days following the Date of
receipt of such documents to satisfy itself with regard to the form and content thereof.	auys nom tile
(m) Personal Property. In the event that any personal property is included in the P	Purchase Price, Buyer has 10 or days following the Date of
Agreement to satisfy itself with regard to the title condition of such personal property. Seller repaid for by Buyer. Seller shall provide Buyer copies of any liens or encumbrances affecting such	ecommends that Buyer obtain a UCC-1 report. Any such report shall be
following the Date of Agreement.	At Clasicathon shall not be a second of the second
(n) Destruction, Damage or Loss. Subsequent to the Date of Agreement and prior loss to, the Property or any portion thereof, from any cause whatsoever, which would cost mor \$10,000.00 or less, Seller shall repair or cure the loss prior to the Closing. Buyer shall have the more than \$10,000.00 to repair or cure, to either terminate this Agreement or to purchase the against the Purchase Price. If the cost to repair or cure is more than \$10,000.00, and Buyer doc insurance proceeds applicable to such loss. Unless otherwise notified in writing, Escrow Holde to Closing.	e than \$10,000.00 to repair or cure. If the cost of repair or cure is option, within 10 days after receipt of written notice of a loss costing Property notwithstanding such loss, but without deduction or offset es not elect to terminate this Agreement, Buyer shall be entitled to any
(o) Material Change. Buyer shall have 10 days following receipt of written notice change. "Material Change" shall mean a substantial adverse change in the use, occupancy, ten offer and prior to the Closing. Unless otherwise notified in writing, Escrow Holder shall assume	nants, title, or condition of the Property that occurs after the date of this
	-
INITIALS INITIALS	-

INITIALS
© 2019 AIR CRE. All Rights Reserved.
OFAL-15.30, Revised 10-13-2022

- (p) Seller Performance. The delivery of all documents and the due performance by Seller of each and every undertaking and agreement to be performed by Seller under this Agreement.
- (q) Brokerage Fee. Payment at the Closing of such brokerage fee as is specified in this Agreement or later written instructions to Escrow Holder executed by Seller and Brokers ("Brokerage Fee"). It is agreed by the Parties and Escrow Holder that Brokers are a third party beneficiary of this Agreement insofar as the Brokerage Fee is concerned, and that no change shall be made with respect to the payment of the Brokerage Fee specified in this Agreement, without the written consent of Brokers.
- 9.2 The contingencies specified in subparagraphs 9.1(a) through (m) are for the benefit of, and may be waived by, Buyer, and are referred to collectively as "Buyer's Contingencies" and individually as a "Buyer's Contingency."
- 9.3 Buyer's timely and written disapproval or conditional approval of a Buyer's Contingency or any other matter that is subject to Buyer's approval in this Agreement shall constitute disapproval thereof ("Disapproved Item(s)"). Concurrent with notice of a Disapproved Item, Buyer may make a request to Seller regarding such Disapproved Item ("Buyer's Request"). If Buyer fails to make a timely and written Buyer's Request, then this Agreement shall terminate due to the non-satisfaction and non-waiver of a contingency. Seller may respond to a Buyer's Request within 10 days following Seller's receipt thereof ("Seller's Response"). Seller's acceptance of a Buyer's Request shall amend this Agreement accordingly. If Seller fails to provide a timely and written Seller's Response, then Seller's Response shall be deemed to be a rejection of Buyer's Request. Buyer may, within 10 days following the earlier of Buyer's receipt of a Seller's Response (which is not an acceptance of Buyer's Request) or the date of Seller's deemed rejection of a Buyer's Request ("Buyer's Reply Period"), reply to a Seller's Response ("Buyer's Reply") and elect to (i) terminate this Agreement due to the non-satisfaction and non-waiver of the applicable contingency, (ii) accept the Seller's Response in which event this Agreement shall be amended accordingly, or (iii) withdraw Buyer's Request and waive the Disapproved Item in which event Buyer shall accept the Property subject to the Disapproved Item. If Buyer fails to provide a timely and written Buyer's Reply, then Buyer shall be deemed to have elected to terminate this Agreement as of the end of the Buyer's Reply Period. The date Buyer accepts a Seller's Response or withdraws a Buyer's Request and waives a Disapproved Item shall be the date of Buyer's approval of the Disapproved Item. A Party shall provide to Escrow Holder copy of all notices of a Disapproved Item, Buyer's Request, Seller's Response and Buyer's Reply and Escrow Holder shall promptly provide copies thereof to the other Party. Unless the Parties in writing agree otherwise, if the Expected Closing Date is a specific calendar date and a Buyer's Reply Period expires after such specific calendar date, then notwithstanding paragraph 1.1, the Expected Closing Date shall be extended to be 3 business days after the earlier of the date Buyer withdraws a Buyer's Request and waives the applicable Disapproved Item or Buyer accepts the applicable Seller's Response.
- 9.4 The Parties acknowledge that extensive local, state and Federal legislation establish broad liability upon owners and/or users of real property for the investigation and remediation of Hazardous Substances. The determination of the existence of a Hazardous Substance Condition and the evaluation of the impact of such a condition are highly technical and beyond the expertise of Brokers. The Parties acknowledge that they have been advised by Brokers to consult their own technical and legal experts with respect to the possible presence of Hazardous Substances on the Property or adjoining properties, and Buyer and Seller are not relying upon any investigation by or statement of Brokers with respect thereto. The Parties hereby assume all responsibility for the impact of such Hazardous Substances upon their respective interests herein.

10. Documents Required at or Before Closing.

- 10.1 Five days prior to the Closing date Escrow Holder shall obtain an updated Title Commitment concerning the Property from the Title Company and provide copies thereof to each of the Parties.
 - 10.2 Seller shall deliver to Escrow Holder in time for delivery to Buyer at the Closing:
 - (a) Grant or general warranty deed, duly executed and in recordable form, conveying fee title to the Property to Buyer.
 - (b) If applicable, the Beneficiary Statements concerning Existing Note(s).
- (c) If applicable, the Existing Leases and Other Agreements together with duly executed assignments thereof by Seller and Buyer. The assignment of Existing Leases shall be on the most recent Assignment and Assumption of Lessor's Interest in Lease form published by the AIR or its equivalent.
- (d) An affidavit executed by Seller to the effect that Seller is not a "foreign person" within the meaning of Internal Revenue Code Section 1445 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Internal Revenue Service such sum as is required by applicable Federal law with respect to purchases from foreign sellers.
- (e) If the Property is located in California, an affidavit executed by Seller to the effect that Seller is not a "nonresident" within the meaning of California Revenue and Tax Code Section 18662 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Franchise Tax Board such sum as is required by such statute.
 - f) If applicable, a bill of sale, duly executed, conveying title to any included personal property to Buyer.
 - (g) If the Seller is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the sale of the Property.
 - 10.3 Buyer shall deliver to Seller through Escrow:
- (a) The cash portion of the Purchase Price and such additional sums as are required of Buyer under this Agreement shall be deposited by Buyer with Escrow Holder, by federal funds wire transfer, or any other method acceptable to Escrow Holder in immediately collectable funds, no later than 2:00 P.M. on the business day prior to the Expected Closing Date provided, however, that Buyer shall not be required to deposit such monies into Escrow if at the time set for the deposit of such monies Seller is in default or has indicated that it will not perform any of its obligations hereunder. Instead, in such circumstances in order to reserve its rights to proceed Buyer need only provide Escrow with evidence establishing that the required monies were available.
- (b) If a Purchase Money Note and Purchase Money Deed of Trust are called for by this Agreement, the duly executed originals of those documents, the Purchase Money Deed of Trust being in recordable form, together with evidence of fire insurance on the improvements in the amount of the full replacement cost naming Seller as a mortgage loss payee, and a real estate tax service contract (at Buyer's expense), assuring Seller of notice of the status of payment of real property taxes during the life of the Purchase Money Note.
 - (c) The Assignment and Assumption of Lessor's Interest in Lease form specified in paragraph 10.2(c) above, duly executed by Buyer.
 - (d) Assumptions duly executed by Buyer of the obligations of Seller that accrue after Closing under any Other Agreements.
 - (e) If applicable, a written assumption duly executed by Buyer of the loan documents with respect to Existing Notes.
 - (f) If the Buyer is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the purchase of the Property.

10.4 At Closing, Escrow Holder shall cause to be issued to Buyer a standard coverage (or ALTA extended, if elected pursuant to 9.1(g)) owner's form policy of title insurance effective as of the Closing, issued by the Title Company in the full amount of the Purchase Price, insuring title to the Property vested in Buyer, subject only to the exceptions approved by Buyer. In the event there is a Purchase Money Deed of Trust in this transaction, the policy of title insurance shall be a joint protection policy insuring both Buyer and Seller.

IMPORTANT: IN A PURCHASE OR EXCHANGE OF REAL PROPERTY, IT MAY BE ADVISABLE TO OBTAIN TITLE INSURANCE IN CONNECTION WITH THE CLOSE OF ESCROW SINCE THERE MAY BE PRIOR RECORDED LIENS AND ENCUMBRANCES WHICH AFFECT YOUR INTEREST IN THE PROPERTY BEING ACQUIRED. A NEW POLICY OF TITLE INSURANCE SHOULD BE OBTAINED IN ORDER TO ENSURE YOUR INTEREST IN THE PROPERTY THAT YOU ARE ACQUIRING.

11. Prorations and Adjustments.

- 11.1 Taxes. Applicable real property taxes and special assessment bonds shall be prorated through Escrow as of the date of the Closing, based upon the latest tax bill available. The Parties agree to prorate as of the Closing any taxes assessed against the Property by supplemental bill levied by reason of events occurring prior to the Closing. Payment of the prorated amount shall be made promptly in cash upon receipt of a copy of any supplemental bill.
- 11.2 *Insurance*. **WARNING**: Any insurance which Seller may have maintained will terminate on the Closing. Buyer is advised to obtain appropriate insurance to cover the Property.
- 11.3 Rentals, Interest and Expenses. Scheduled rentals, interest on Existing Notes, utilities, and operating expenses shall be prorated as of the date of Closing. The Parties agree to promptly adjust between themselves outside of Escrow any rents received after the Closing.
 - 11.4 Security Deposit. Security Deposits held by Seller shall be given to Buyer as a credit to the cash required of Buyer at the Closing.
- 11.5 Post Closing Matters. Any item to be prorated that is not determined or determinable at the Closing shall be promptly adjusted by the Parties by appropriate cash payment outside of the Escrow when the amount due is determined.
- 11.6 Variations in Existing Note Balances. In the event that Buyer is purchasing the Property subject to an Existing Deed of Trust(s), and in the event that a Beneficiary Statement as to the applicable Existing Note(s) discloses that the unpaid principal balance of such Existing Note(s) at the closing will be more or less than the amount set forth in paragraph 3.1(c) hereof ("Existing Note Variation"), then the Purchase Money Note(s) shall be reduced or increased by an amount equal to such Existing Note Variation. If there is to be no Purchase Money Note, the cash required at the Closing per paragraph 3.1(a) shall be reduced or increased by the amount of such Existing Note Variation.
- 11.7 Variations in New Loan Balance. In the event Buyer is obtaining a New Loan and the amount ultimately obtained exceeds the amount set forth in paragraph 5.1, then the amount of the Purchase Money Note, if any, shall be reduced by the amount of such excess.
- 11.8 Owner's Association Fees. Escrow Holder shall: (i) bring Seller's account with the association current and pay any delinquencies or transfer fees from Seller's proceeds, and (ii) pay any up front fees required by the association from Buyer's funds.

INITIALS		INITIALS
@ 2040 AIR CRE	All Di Li D	

12. Representations and Warranties of Seller and Disclaimers.

- 12.1 Seller's warranties and representations shall survive the Closing and delivery of the deed for a period of 3 years, and any lawsuit or action based upon them must be commenced within such time period. Seller's warranties and representations are true, material and relied upon by Buyer and Brokers in all respects. Seller hereby makes the following warranties and representations to Buyer and Brokers:
- (a) Authority of Seller. Seller is the owner of the Property and/or has the full right, power and authority to sell, convey and transfer the Property to Buyer as provided herein, and to perform Seller's obligations hereunder.
- (b) Maintenance During Escrow and Equipment Condition At Closing. Except as otherwise provided in paragraph 9.1(n) hereof, Seller shall maintain the Property until the Closing in its present condition, ordinary wear and tear excepted.
- (c) Hazardous Substances/Storage Tanks. Seller has no knowledge, except as otherwise disclosed to Buyer in writing, of the existence or prior existence on the Property of any Hazardous Substance, nor of the existence or prior existence of any above or below ground storage tank.
- (d) Compliance. Except as otherwise disclosed in writing, Seller has no knowledge of any aspect or condition of the Property which violates applicable laws, rules, regulations, codes or covenants, conditions or restrictions, or of improvements or alterations made to the Property without a permit where one was required, or of any unfulfilled order or directive of any applicable governmental agency or casualty insurance company requiring any investigation, remediation, repair, maintenance or improvement be performed on the Property.
- (e) Changes in Agreements. Prior to the Closing, Seller will not violate or modify any Existing Lease or Other Agreement, or create any new leases or other agreements affecting the Property, without Buyer's written approval, which approval will not be unreasonably withheld.
- Possessory Rights. Seller has no knowledge that anyone will, at the Closing, have any right to possession of the Property, except as disclosed by this Agreement or otherwise in writing to Buyer.
 - Mechanics' Liens. There are no unsatisfied mechanics' or materialmens' lien rights concerning the Property.
- Actions, Suits or Proceedings. Seller has no knowledge of any actions, suits or proceedings pending or threatened before any commission, board, (h) bureau, agency, arbitrator, court or tribunal that would affect the Property or the right to occupy or utilize same.
- (i) Notice of Changes. Seller will promptly notify Buyer and Brokers in writing of any Material Change (see paragraph 9.1(o)) affecting the Property that becomes known to Seller prior to the Closing.
- No Tenant Bankruptcy Proceedings. Seller has no notice or knowledge that any tenant of the Property is the subject of a bankruptcy or insolvency (j) proceeding.
 - (k) No Seller Bankruptcy Proceedings. Seller is not the subject of a bankruptcy, insolvency or probate proceeding.
- Personal Property. Seller has no knowledge that anyone will, at the Closing, have any right to possession of any personal property included in the Purchase Price nor knowledge of any liens or encumbrances affecting such personal property, except as disclosed by this Agreement or otherwise in writing to Buyer.
- 12.2 Buyer hereby acknowledges that, except as otherwise stated in this Agreement, Buyer is purchasing the Property in its existing condition and will, by the time called for herein, make or have waived all inspections of the Property Buyer believes are necessary to protect its own interest in, and its contemplated use of, the Property. The Parties acknowledge that, except as otherwise stated in this Agreement, no representations, inducements, promises, agreements, assurances, oral or written, concerning the Property, or any aspect of the occupational safety and health laws, Hazardous Substance laws, or any other act, ordinance or law, have been made by either Party or Brokers, or relied upon by either Party hereto.
- 12.3 In the event that Buyer learns that a Seller representation or warranty might be untrue prior to the Closing, and Buyer elects to purchase the Property anyway then, and in that event, Buyer waives any right that it may have to bring an action or proceeding against Seller or Brokers regarding said representation or warranty.
- 12.4 Any environmental reports, soils reports, surveys, feasibility studies, and other similar documents which were prepared by third party consultants and provided to Buyer by Seller or Seller's representatives, have been delivered as an accommodation to Buyer and without any representation or warranty as to the sufficiency, accuracy, completeness, and/or validity of said documents, all of which Buyer relies on at its own risk. Seller believes said documents to be accurate, but Buyer is advised to retain appropriate consultants to review said documents and investigate the Property.

Possession.

Possession of the Property shall be given to Buyer at the Closing subject to the rights of tenants under Existing Leases.

14. Buyer's Entry.

At any time during the Escrow period, Buyer, and its agents and representatives, shall have the right at reasonable times and subject to rights of tenants, to enter upon the Property for the purpose of making inspections and tests specified in this Agreement. No destructive testing shall be conducted, however, without Seller's prior approval which shall not be unreasonably withheld. Following any such entry or work, unless otherwise directed in writing by Seller, Buyer shall return the Property to the condition it was in prior to such entry or work, including the recompaction or removal of any disrupted soil or material as Seller may reasonably direct. All such inspections and tests and any other work conducted or materials furnished with respect to the Property by or for Buyer shall be paid for by Buyer as and when due and Buyer shall indemnify, defend, protect and hold harmless Seller and the Property of and from any and all claims, liabilities, losses, expenses (including reasonable attorneys' fees), damages, including those for injury to person of property, arising out of or relating to any such work or materials or the acts or omissions of Buyer, its agents or employees in connection therewith.

Further Documents and Assurances.

The Parties shall each, diligently and in good faith, undertake all actions and procedures reasonably required to place the Escrow in condition for Closing as and when required by this Agreement. The Parties agree to provide all further information, and to execute and deliver all further documents, reasonably required by Escrow Holder or the Title Company.

If any Party or Broker brings an action or proceeding (including arbitration) involving the Property whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees and costs. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred.

Prior Agreements/Amendments.

- 17.1 This Agreement supersedes any and all prior agreements between Seller and Buyer regarding the Property.
- 17.2 Amendments to this Agreement are effective only if made in writing and executed by Buyer and Seller.

Broker's Rights.

18.1 If this sale is not consummated due to the default of either the Buyer or Seller, the defaulting Party shall be liable to and shall pay to Brokers the Brokerage Fee that Brokers would have received had the sale been consummated. If Buyer is the defaulting party, payment of said Brokerage Fee is in addition to any obligation with respect to liquidated or other damages.

18.2 Upon the Closing, Brokers are authorized to publicize the facts of this transaction.

19.1 Whenever any Party, Escrow Holder or Brokers herein shall desire to give or serve any notice, demand, request, approval, disapproval or other communication, each such communication shall be in writing and shall be delivered personally, by messenger, or by mail, postage prepaid, to the address set forth in this agreement or by facsimile transmission, electronic signature, digital signature, or email.

19.2 Service of any such communication shall be deemed made on the date of actual receipt if personally delivered, or transmitted by facsimile transmission, electronic signature, digital signature, or email. Any such communication sent by regular mail shall be deemed given 48 hours after the same is mailed. Communications sent by United States Express Mail or overnight courier that guarantee next day delivery shall be deemed delivered 24 hours after delivery of the same to the Postal Service or courier. If such communication is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.

19.3 Any Party or Broker hereto may from time to time, by notice in writing, designate a different address to which, or a different person or additional persons to whom, all communications are thereafter to be made.

20.	Duration	of	Offer.

.0.	diation of Office.
	0.1 If this offer is not accepted by Seller on or before 5:00 P.M. according to the time standard applicable to the city of on the date of
	, it shall be deemed automatically revoked.
	0.2 The acceptance of this offer, or of any subsequent counteroffer hereto, that creates an agreement between the Parties as described in paragraph 1.2, shall
	
NIT	LS INITIALS

be deemed made upon delivery to the other Party or either Broker herein of a duly executed w counteroffer.	riting unconditionally accepting the last outstanding offer or
21. LIQUIDATED DAMAGES. (This Liquidated Damages paragraph is applicable only if initials the parties agree that it would be impracticable or extremely difficult to fix, preserved by seller if buyer fails to perform its obligations under this of all contingencies provided for the buyer's benefit, buyer breaches this agree amount of Upon payment of said sum to seller, buyer shall be relecancellation fees and title company charges shall be paid by seller.	IOR TO SIGNING THIS AGREEMENT, THE ACTUAL DAMAGES WHICH AGREEMENT. THEREFORE, IF, AFTER THE SATISFACTION OR WAIVER
Buyer's Initials	Seller's Initials

22. ARBITRATION OF DISPUTES. (This Arbitration of Disputes paragraph is applicable only if initialed by both Parties.)

22.1 ANY CONTROVERSY AS TO WHETHER SELLER IS ENTITLED TO LIQUIDATED DAMAGES AND/OR BUYER IS ENTITLED TO THE RETURN OF THE DEPOSIT SHALL BE DETERMINED BY BINDING ARBITRATION ADMINISTERED BY THE JUDICIAL ARBITRATION & MEDIATION SERVICES, INC. ("JAMS") IN ACCORDANCE WITH ITS COMMERCIAL ARBITRATION RULES ("COMMERCIAL RULES"). ARBITRATION HEARINGS SHALL BE HELD IN THE COUNTY WHERE THE PROPERTY IS LOCATED. SUCH CONTROVERSY SHALL BE ARBITRATED BY A SINGLE ARBITRATOR. APPOINTED UNDER THE COMMERCIAL RULES WHO HAS HAD AT LEAST 5 YEARS OF EXPERIENCE IN THE TYPE OF REAL ESTATE THAT IS THE SUBJECT OF THIS AGREEMENT. THE ARBITRATOR SHALL HEAR AND DETERMINE SAID CONTROVERSY IN ACCORDANCE WITH APPLICABLE LAW OF THE JURISDICTION WHERE THE PROPERTY IS LOCATED. THE INTENTION OF THE PARTIES AS EXPRESSED IN THIS AGREEMENT AND ANY AMENDMENTS THERETO, AND UPON THE EVIDENCE PRODUCED AT AN ARBITRATION HEARING. PRE-ARBITRATION DISCOVERY SHALL BE PERMITTED IN ACCORDANCE WITH THE COMMERCIAL RULES OR STATE LAW APPLICABLE TO ARBITRATION PROCEEDINGS. THE ARBITRATOR SHALL RENDER AN AWARD WITHIN 30 DAYS AFTER THE CONCLUSION OF THE HEARING, WHICH MAY INCLUDE ATTORNEYS' FEES AND COSTS TO THE PREVAILING PARTY PER PARAGRAPH 16 HEREOF AND SHALL BE ACCOMPANIED BY A REASONED OPINION. THE FAILURE OR REFUSAL OF A PARTY TO PAY SUCH PARTY'S REQUIRED SHARE OF THE DEPOSITS FOR ARBITRATOR COMPENSATION OR ADMINISTRATIVE CHARGES SHALL CONSTITUTE A WAIVER BY SUCH PARTY TO PRESENT EVIDENCE OR CROSS-EXAMINE WITNESSES, BUT SUCH WAIVER SHALL NOT ALLOW FOR A DEFAULT JUDGMENT AGAINST THE NON-PAYING PARTY IN THE ABSENCE OF EVIDENCE AND LEGAL ARGUMENT AS THE ARBITRATOR MAY REQUIRE FOR MAKING AN AWARD. JUDGMENT MAY BE ENTERED ON THE AWARD IN ANY COURT OF COMPETENT JURISDICTION NOTWITHSTANDING THE FAILURE OF A PARTY DULY NOTIFIED OF THE ARBITRATION HEARING TO APPEAR THEREAT.

22.2 BUYER'S RESORT TO OR PARTICIPATION IN SUCH ARBITRATION PROCEEDINGS SHALL NOT BAR SUIT IN A COURT OF COMPETENT JURISDICTION BY THE BUYER FOR DAMAGES AND/OR SPECIFIC PERFORMANCE UNLESS AND UNTIL THE ARBITRATION RESULTS IN AN AWARD TO THE SELLER OF LIQUIDATED DAMAGES, IN WHICH EVENT SUCH AWARD SHALL ACT AS A BAR AGAINST ANY ACTION BY BUYER FOR DAMAGES AND/OR SPECIFIC PERFORMANCE.

22.3 NOTICE: BY INITIALING IN THE SPACE BELOW YOU ARE AGREEING TO HAVE ANY DISPUTE ARISING OUT OF THE MATTERS INCLUDED IN THE "ARBITRATION" OF DISPUTES" PROVISION DECIDED BY NEUTRAL ARBITRATION AS PROVIDED BY CALIFORNIA LAW AND YOU ARE GIVING UP ANY RIGHTS YOU MIGHT POSSESS TO HAVE THE DISPUTE LITIGATED IN A COURT OR JURY TRIAL. BY INITIALING IN THE SPACE BELOW YOU ARE GIVING UP YOUR JUDICIAL RIGHTS TO DISCOVERY AND APPEAL, UNLESS SUCH RIGHTS ARE SPECIFICALLY INCLUDED IN THE "ARBITRATION OF DISPUTES" PROVISION. IF YOU REFUSE TO SUBMIT TO ARBITRATION AFTER AGREEING TO THIS PROVISION, YOU MAY BE COMPELLED TO ARBITRATE UNDER THE AUTHORITY OF THE CALIFORNIA CODE OF CIVIL PROCEDURE. YOUR AGREEMENT TO THIS ARBITRATION PROVISION IS VOLUNTARY.

WE HAVE READ AND UNDERSTAND THE FOREGOING AND AGREE TO SUBMIT DISPUTES ARISING OUT OF THE MATTERS INCLUDED IN THE "ARBITRATION OF DISPUTES" PROVISION TO NEUTRAL ARBITRATION.

Buyer's Initials		Seller's Initials	

23. Miscellaneous.

- 23.1 Binding Effect. Buyer and Seller both acknowledge that they have carefully read and reviewed this Agreement and each term and provision contained herein. In addition, this Agreement shall be binding on the Parties without regard to whether or not paragraphs 21 and 22 are initialed by both of the Parties. Paragraphs 21 and 22 are each incorporated into this Agreement only if initialed by both Parties at the time that the Agreement is executed. Signatures to this
- Agreement accomplished by means of electronic signature or similar technology shall be legal and binding.

 23.2 **Applicable Law**. This Agreement shall be governed by, and paragraph 22.3 is amended to refer to, the laws of the state in which the Property is located. Any litigation or arbitration between the Parties hereto concerning this Agreement shall be initiated in the county in which the Property is located.
 - 23.3 Time of Essence. Time is of the essence of this Agreement.
- 23.4 **Counterparts**. This Agreement may be executed by Buyer and Seller in counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument. Escrow Holder, after verifying that the counterparts are identical except for the signatures, is authorized and instructed to combine the signature pages on one of the counterparts, which shall then constitute the Agreement.
- 23.5 Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS AGREEMENT.
- 23.6 Conflict. Any conflict between the printed provisions of this Agreement and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions. Seller and Buyer must initial any and all handwritten provisions.

 23.7 1031 Exchange. Both Seller and Buyer agree to cooperate with each other in the event that either or both wish to participate in a 1031 exchange. Any
- party initiating an exchange shall bear all costs of such exchange. The cooperating Party shall not have any liability (special or otherwise) for damages to the exchanging Party in the event that the sale is delayed and/or that the sale otherwise fails to qualify as a 1031 exchange.
 - 23.8 Days. Unless otherwise specifically indicated to the contrary, the word "days" as used in this Agreement shall mean and refer to calendar days.

Disclosures Regarding The Nature of a Real Estate Agency Relationship.

- 24.1 The Parties and Brokers agree that their relationship(s) shall be governed by the principles set forth in the applicable sections of the California Civil Code, as summarized in paragraph 24.2.
- 24.2 When entering into a discussion with a real estate agent regarding a real estate transaction, a Buyer or Seller should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Buyer and Seller acknowledge being advised by the Brokers in this transaction, as follows:
- (a) Seller's Agent. A Seller's agent under a listing agreement with the Seller acts as the agent for the Seller only. A Seller's agent or subagent has the following affirmative obligations: (1) To the Seller: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Seller. (2) To the Buyer and the Seller: a. Diligent exercise of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith. c. A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set
- Buyer's Agent. A selling agent can, with a Buyer's consent, agree to act as agent for the Buyer only. In these situations, the agent is not the Seller's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Seller. An agent acting only for a Buyer has the following affirmative obligations. (1) To the Buyer: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Buyer. (2) To the Buyer and the Seller: a. Diligent exercise of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith. c. A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set forth above.
- (c) Agent Representing Both Seller and Buyer. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Seller and the Buyer in a transaction, but only with the knowledge and consent of both the Seller and the Buyer. (1) In a dual agency situation, the agent has the following affirmative obligations to both the Seller and the Buyer: a. A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Seller or the Buyer. b. Other duties to the Seller and the Buyer as stated above in their respective sections (a) or (b) of this paragraph 24.2. (2) In representing both Seller and Buyer, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

not limited to, facts relating to either Buyer's or Seller's financial position, motivations, bargaining position, or other personal information that may impact price, including Seller's willingness to accept a price less than the listing price or Buyer's willingness to pay a price greater than the price offered. (3) The above duties of the agent in a real estate transaction do not relieve a Seller or Buyer from the responsibility to protect their own interests. Buyer and Seller should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Buyer has the duty to exercise reasonable care to protect Buyer, including as to those facts about the Property which are known to Buyer or within Buyer's diligent attention and observation. Both Seller and Buyer should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.

(d) Further Disclosures. Throughout this transaction Buyer and Seller may receive more than one disclosure, depending upon the number of agents assisting in the transaction. Buyer and Seller should each read its contents each time it is presented, considering the relationship between them and the real estate agent in this transaction and that disclosure. Buyer and Seller each acknowledge receipt of a disclosure of the possibility of multiple representation by the Broker representing that principal. This disclosure may be part of a listing agreement, buyer representation agreement or separate document. Buyer understands that Broker representing Buyer may also represent other potential buyers, who may consider, make offers on or ultimately acquire the Property. Seller understands that Broker representing Seller may also represent other sellers with competing properties that may be of interest to this Buyer. Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this transaction may be brought against Broker more than one year after the Date of Agreement and that the liability (including court costs and attorneys' fees), of any Broker with respect to any breach of duty, error or omission relating to this Agreement; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.

24.3 Confidential Information. Buyer and Seller agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.

25. Construction of Agreement. In construing this Agreement, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Agreement. Whenever required by the context, the singular shall include the plural and vice versa. This Agreement shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.

26.	Additional Provision	anc

Additional provisions of this offer, if any, are as follows or are attached hereto by an addendum or addenda consisting of paragraphs ______ through _______. (If there are no additional provisions write "NONE".)

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS AGREEMENT OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- 1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS AGREEMENT.
- 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PROPERTY. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PROPERTY, THE INTEGRITY AND CONDITION OF ANY STRUCTURES AND OPERATING SYSTEMS, AND THE SUITABILITY OF THE PROPERTY FOR BUYER'S INTENDED USE.

WARNING: IF THE PROPERTY IS LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THIS AGREEMENT MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PROPERTY IS LOCATED.

NOTE:

- 1. THIS FORM IS NOT FOR USE IN CONNECTION WITH THE SALE OF RESIDENTIAL PROPERTY.
- 2. IF EITHER PARTY IS A CORPORATION, IT IS RECOMMENDED THAT THIS AGREEMENT BE SIGNED BY TWO CORPORATE OFFICERS.

The undersigned Buyer offers and agrees to buy the Property on the terms and conditions stated and acknowledges receipt of a copy hereof.

\	Butt.
BROKER	BUYER
	\
Atta	
Attn:	Ву:
Title:	Name Printed:
Address	Title:
Address:	Phone:
Phone:	Fax:
Fax:	Email:
Email:	
Federal ID No.:	D
Broker DRE License #:	By:
Agent DRE License #:	Name Printed:
	Title:
	Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
_ Y	

27. Acceptance.

27.1 Seller accepts the foregoing offer to purchase the Property and hereby agrees to sell the Property to Buyer on the terms and conditions therein specified.

27.2 In consideration of real estate brokerage service rendered by Brokers, Seller agrees to pay Brokers a real estate Brokerage Fee in a sum equal to

% of the Purchase Price to be divided between the Brokers as follows: Seller's Broker ______ % and Buyer's Broker ______ %. This

Agreement shall serve as an irrevocable instruction to Escrow Holder to pay such Brokerage Fee to Brokers out of the proceeds accruing to the account of Seller at the Closing.

27.3 Seller acknowledges receipt of a copy hereof and authorizes Brokers to deliver a signed copy to Buyer.

NOTE: A PROPERTY INFORMATION SHEET IS REQUIRED TO BE DELIVERED TO BUYER BY SELLER UNDER THIS AGREEMENT.

		Date:
BROKER		SELLER
Attn:		
INITIALS	All Bights Recogned	INITIALS
(C) JOJO AID CDE	All Dights Dosonyod	

Title:	
	Ву:
Address:	Name Printed:
Phone: Fax:	Title:
Email:	Phone: Fax:
Federal ID No.:	Email:
Broker DRE License #:	
Agent's DRE License #:	Ву:
	Name Printed:
	Title: Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
AIR CRE * https://www.aircre.com * NOTICE: No part of these works may be repro	* 213-687-8777 * contracts@aircre.com oduced in any form without permission in writing.
No nezi No part of these works may be repre	National Manager Permission in Manager
	< \
	< / ·
\	
\sim \times	
\ \ \	



Dated:

STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS FOR PURCHASE OF RESIDENTIAL INCOME PROPERTY

(Do Not Use for a Property Which Contains Less than Five Units. Do Not Use for a Condominium or for a Common Interest Development)

1.	Buyer.		/II= II) I						5 (II = 11 II) (
indiv	1.1 idually a				chase the real proper close 30 or						
					der") whose address						
and of Buye	conditions or of Buyer 1.2 The equent co	set fo r's oblig term untero	rth in this agreer gations herein ur " Date of Agree m	nent (" Agreement " less Seller expressl ent " as used hereir). Buyer shall have th	ne right to assig	n Buyer's rights he	ereunder, bu defined in p	t any such assig aragraph 20.2)	nment shall not re	elieve or a
2.	Property										
					ect of this offer consi y, state, zip)					//	/ of
part telep equi	2.2 If the ected to m 2.3 The of the prophone distorment; fillite dishes	ne lega neet the Prope perty, ributio ire spri	I description of t e requirements of erty includes, at r as well as the fol n systems (lines, nkler systems; so and spa equipme	the Property is not of f ("T o additional cost to cowing items, if any jacks and connection courity and fire detern; garage door op	omplete or is inaccur itle Company"), whice Buyer, the permane cons only); built-in appection systems; carpet eners; and	rate, this Agreei ch shall issue th ent improvemen d at present loca pliances; plumb ts; window cov (collective	ment shall not be e title policy herei ts thereon, includ ated on the Propeing fixtures; ceilingerings and screens ly, the "Improvem"	invalid and t inafter descr ling those ite erty: electrica g fans; firepl s; attached w nents").	he legal descrip ibed. ms which pursu I distribution sy aces; heating, v vall and floor co	tion shall be comp ant to applicable I stems and lighting entilating, air cond verings; television	law are a g fixtures; litioning antennas
with				•	e determined during E nase Price does not in			//	/ v	nd a	all of
whic	h shall be	remov	ed by Seller prio	to Closing.							
3.	Purchase	Price.									
		-	ase price (" Purcl / not applicable)	nase Price") to be p	aid by Buyer to Seller	r for the Proper	ty shall be	, paya	ible as follows:		
	,	,	, ,	teal alternative Decre	-11 d-6 d-1						
	(a)	Casn	i down payment,	including the Depo	sit as defined in para	agraph 4.3 (or if	an all cash transa	ction, the Pu	rchase Price):		
	(b)	Amo	ount of "New Loa	n" as defined in par	ragraph 5.1, if any:						
	(c)	Trus	$oldsymbol{t}$ ") securing the $oldsymbol{\epsilon}$	existing promissory	ject to and/or assum note(s) (" Existing No n an unpaid principal	te(s)"):			ting Deed(s) of		
	40	(ii)	until paid (and/ An Existing Not Said Second No until paid (and/	or the entire unpaid ("Second Note") te is payable at or the entire unpaid	per month, iu d balance is due on _ with an unpaid princip per month d balance is due on _). pal balance as o h, including inte).	of the Closing of aperest at the rate of	pproximately	r: % per annum		_
	(d)	•	-	•	Purchase Money Dee 1 6 ("Purchase Mone	•		ecure the pro	omissory note		
Total	Purchase	Price:									
	es includi	ng, but .5% of	not limited to, p	oints, processing fe	to, or assuming, an Exes, and appraisal fees applicable Existing No	s as a condition					
	4.1	Buye	r has delivered to	Broker a check in	the sum of	, payable to	o Escrow Holder, t	o be delivere	ed by Broker to	Escrow Holder wit	hin 2 or
term	ow Holder inate this ement. Si	busine a ched transa hould E	ss days after botl ck in the sum of ction by giving w	n Parties have execu . If sa ritten notice of suc	Ited this Agreement a Ited this Agreement a id check is not receive the election to Escrow I Igreement for purchase	and the execute ed by Escrow H Holder whereup	ed Agreement has older within said to oon neither Party	been delive time period t shall have an	red to Escrow H hen Seller may ly further liabili	older Buyer shall of elect to unilaterally ty to the other und	deliver to ly der this
			•	ys after the Date of	Agreement, Buyer sh	hall deposit with	n Escrow Holder th	he additional	sum of	to be appli	ed to the
Purc	hase Price		•	s after the contine	oncine discussed in	aragraph 0 1 /=	through (m) are	annroyed s-	waived Books	hall donosit with	Eccross
Hold				_	encies discussed in pa I to the Purchase Pric			approved or	waived, Buyer s	snaii deposit with i	:SCrOW
char accru spec	ng that, under notice 4.3 Escape tered bankue to the bified matu	nless the or instance or insta	he Additional De ructions. older shall depos interest bearing of Buyer, who he	oosit is received by t the funds deposit account whose terr ereby acknowledge	oy Escrow Holder with Escrow Holder within ed with it by Buyer p in is appropriate and o is that there may be p inber is	n 2 business day oursuant to para consistent with penalties or inte	ys following said n graphs 4.1 and 4.2 the timing requir rest forfeitures if t	otice, the Es 2 (collectively ements of th the applicabl	crow shall be do y the "Deposit" is transaction. le instrument is	eemed terminated), in a State or Fed The interest theref redeemed prior to	without erally from shall o its
							_				
INIT	IALS					INITIAL	<u></u>				
		CRE. A	All Rights Reser	ved.			-		Last Edited	: 10/12/2022 3:0)4 PM

ROFA-11.30, Revised 10-12-2022

4.4 Notwithstanding the foregoing, within 5 days after Escrow Holder receives the monies described in paragraph 4.1 above, Escrow Holder shall release \$100 of said monies to Seller as and for independent consideration for Seller's' execution of this Agreement and the granting of the contingency period to Buyer as herein provided. Such independent consideration is non-refundable to Buyer but shall be credited to the Purchase Price in the event that the purchase of the Property is completed. 4.5 Upon waiver of all of Buyer's contingencies the Deposit shall become non-refundable but applicable to the Purchase Price except in the event of a Seller breach, or in the event that the Escrow is terminated pursuant to the provisions of Paragraph 9.1(n) (Destruction, Damage or Loss) or 9.1(o) (Material Change). Financing Contingency. (Strike if not applicable) 5.1 This offer is contingent upon Buyer obtaining from an insurance company, financial institution or other lender, a commitment to lend to Buyer a sum equal % of the Purchase Price, on terms acceptable to Buyer. Such loan ("New Loan") shall be secured by a first deed of trust or mortgage on the to at least Property. If this Agreement provides for Seller to carry back junior financing, then Seller shall have the right to approve the terms of the New Loan. Seller shall have 7 days from receipt of the commitment setting forth the proposed terms of the New Loan to approve or disapprove of such proposed terms. If Seller fails to notify Escrow Holder, in writing, of the disapproval within said 7 days it shall be conclusively presumed that Seller has approved the terms of the New Loan. 5.2 If Buyer shall fail to notify its Broker, Escrow Holder and Seller, in writing within days following the Date of Agreement, that the New Loan has not been obtained, it shall be conclusively presumed that Buyer has either obtained said New Loan or has waived this New Loan contingency. 5.3 If Buyer shall notify its Broker, Escrow Holder and Seller, in writing, within the time specified in paragraph 5.2 hereof, that Buyer has not obtained said New Loan, this Agreement shall be terminated, and Buyer shall be entitled to the prompt return of the Deposit, plus any interest earned thereon, less only Escrow Holder and Title Company cancellation fees and costs, which Buyer shall pay. Seller Financing. (Purchase Money Note). (Strike if not applicable) 6.1 If Seller approves Buyer's financials (see paragraph 6.5) the Purchase Money Note shall provide for interest on unpaid principal at the rate of % per annum, with principal and interest paid as follows: _. The Purchase Money Note and Purchase Money Deed of Trust shall be on the current forms nmonly used by Escrow Holder, and be junior and subordinate only to the Existing Note(s) and/or the New Loan expressly called for by this Agreement 6.2 The Purchase Money Note and/or the Purchase Money Deed of Trust shall contain provisions regarding the following (see also paragraph 10.3 (b)) Prepayment. Principal may be prepaid in whole or in part at any time without penalty, at the option of the Buyer. Late Charge. A late charge of 6% shall be payable with respect to any payment of principal, interest, or other charges, not made within 10 days after (b) it is due. Due On Sale. In the event the Buyer sells or transfers title to the Property or any portion thereof, then the Seller may, at Seller's option, require the (c) entire unpaid balance of said Note to be paid in full. 6.3 If the Purchase Money Deed of Trust is to be subordinate to other financing, Escrow Holder shall, at Buyer's expense prepare and record on Seller's behalf a request for notice of default and/or sale with regard to each mortgage or deed of trust to which it will be subordinate. 6.4 WARNING: CALIFORNIA LAW DOES NOT ALLOW DEFICIENCY JUDGEMENTS ON SELLER FINANCING. IF BUYER ULTIMATELY DEFAULTS ON THE LOAN, SELLER'S SOLE REMEDY IS TO FORECLOSE ON THE PROPERTY. 6.5 Seller's obligation to provide financing is contingent upon Seller's reasonable approval of Buyer's financial condition. Buyer to provide a current financial statement and copies of its Federal tax returns for the last 3 years to Seller within 10 days following the Date of Agreement. Seller has 10 days following receipt of such documentation to satisfy itself with regard to Buyer's financial condition and to notify Escrow Holder as to whether or not Buyer's financial condition is acceptable. If Seller fails to notify Escrow Holder, in writing, of the disapproval of this contingency within said time period, it shall be conclusively presumed that Seller has approved Buyer's financial condition. If Seller is not satisfied with Buyer's financial condition or if Buyer fails to deliver the required documentation then Seller may notify Escrow Holder in writing that Seller Financing will not be available, and Buyer shall have the option, within 10 days of the receipt of such notice, to either terminate this transaction or to purchase the Property without Seller financing. If Buyer fails to notify Escrow Holder within said time period of its election to terminate this transaction then Buyer shall be conclusively presumed to have elected to purchase the Property without Seller financing. If Buyer elects to terminate, Buyer's Deposit shall be refunded less Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. 7.1 Each Party acknowledges receiving a Disclosure Regarding Real Estate Agency Relationship, confirms and consents to the following agency relationships in $this \, transaction \, \, with \, the \, \, following \, real \, estate \, \, broker(s) \, ("\textbf{Brokers"}) \, \, and/or \, their \, agents \, ("Agent(s)"): \, \, ("\textbf{Agent}(s)"): \, \, ("\textbf{Agent}(s)"): \, \, ((\textbf{Agent}(s))"): \, \, ((\textbf{Agent}($ is the broker of (check one): Lathe Seller; or both the Buyer and Seller (dual agent). License No. Seller's Brokerage Firm the Seller's Agent (salesperson or broker associate); or both the Seller's Agent is (check one): License No. Seller's Agent and the Buyer's Agent (dual agent). is the broker of (check one): $\ \ \ \ \$ the Buyer; or $\ \ \ \ \$ both the Buyer and Seller (dual agent). Buyer's Brokerage Firm License No. the Buyer's Agent (salesperson or broker associate); or 🔲 both the Buyer's Agent is (check one): Buyer's Agent License No. and the Seller's Agent (dual agent). The Parties acknowledge that other than the Brokers and Agents listed above, there are no other brokers or agents representing the Parties or due any fees and/or commissions under this Agreement. Buyer shall use the services of Buyer's Broker exclusively in connection with any and all negotiations and offers with respect to the Property for a period of 1 year from the date inserted for reference purposes at the top of page 1. 7.2 Buyer and Seller each represent and warrant to the other that he/she/it has had no dealings with any person, firm, broker, agent or finder in connection with the negotiation of this Agreement and/or the consummation of the purchase and sale contemplated herein, other than the Brokers and Agents named in paragraph 7.1, and no broker, agent or other person, firm or entity, other than said Brokers and Agents is/are entitled to any commission or finder's fee in connection with this transaction as the result of any dealings or acts of such Party. Buyer and Seller do each hereby agree to indemnify, defend, protect and hold the other harmless from and against any costs, expenses or liability for compensation, commission or charges which may be claimed by any broker, agent, finder or other similar party, other than said named Brokers and Agents by reason of any dealings or act of the indemnifying Party. 8.1 Upon acceptance hereof by Seller, this Agreement, including any counteroffers incorporated herein by the Parties, shall constitute not only the agreement of purchase and sale between Buyer and Seller, but also instructions to Escrow Holder for the consummation of the Agreement through the Escrow. Escrow Holder shall not prepare any further escrow instructions restating or amending the Agreement unless specifically so instructed by the Parties or a Broker herein. Subject to the reasonable approval of the Parties, Escrow Holder may, however, include its standard general escrow provisions. In the event that there is any conflict between the provisions of the Agreement and the provisions of any additional escrow instructions the provisions of the Agreement shall prevail as to the Parties and the Escrow Holden 8.2 As soon as practical after the receipt of this Agreement and any relevant counteroffers, Escrow Holder shall ascertain the Date of Agreement as defined in paragraphs 1.2 and 20.2 and advise the Parties and Brokers, in writing, of the date ascertained. Escrow Holder is hereby authorized and instructed to conduct the Escrow in accordance with this Agreement, applicable law and custom and practice of the community in which Escrow Holder is located, including any reporting requirements of the Internal Revenue Code. In the event of a conflict between the law of the state where the Property is located and the law of the state where the Escrow Holder is located, the law of the state where the Property is located shall prevail. 8.4 Subject to satisfaction of the contingencies herein described, Escrow Holder shall close this escrow (the "Closing") by recording a general warranty deed (a grant deed in California) and the other documents required to be recorded, and by disbursing the funds and documents in accordance with this Agreement. 8.5 Buyer and Seller shall each pay one-half of the Escrow Holder's charges and Seller shall pay the usual recording fees and any required documentary transfer taxes. Seller shall pay the premium for a standard coverage owner's or joint protection policy of title insurance. (See also paragraph 11.) 8.6 Escrow Holder shall verify that all of Buyer's contingencies have been satisfied or waived prior to Closing. The matters contained in paragraphs 9.1 subparagraphs (b), (c), (d), (e), (g), (i), (n), and (o), 9.4, 12, 13, 14, 16, 18, 20, 21, 22, and 24 are, however, matters of agreement between the Parties only and are not instructions to Escrow Holder. 8.7 If this transaction is terminated for non-satisfaction and non-waiver of a Buyer's Contingency, as defined in Paragraph 9.2 or disapproval of any other matter subject to Buyer's approval, then neither of the Parties shall thereafter have any liability to the other under this Agreement, except to the extent of a breach of any affirmative covenant or warranty in this Agreement. In the event of such termination, Buyer shall, subject to the provisions of paragraph 8.10, be promptly

8.8 The Closing shall occur on the Expected Closing Date, or as soon thereafter as the Escrow is in condition for Closing; provided, however, that if the Closing

INITIALS

refunded all funds deposited by Buyer with Escrow Holder, less only the \$100 provided for in paragraph 4.4 and the Title Company and Escrow Holder cancellation fees and costs, all of which shall be Buyer's obligation. If this transaction is terminated as a result of Seller's breach of this Agreement then Seller shall pay the Title

Company and Escrow Holder cancellation fees and costs.

does not occur by the Expected Closing Date and said Date is not extended by mutual instructions of the Parties, a Party not then in default under this Agreement may notify the other Party, Escrow Holder, and Brokers, in writing that, unless the Closing occurs within 5 business days following said notice, the Escrow shall be deemed terminated without further notice or instructions.

8.9 Except as otherwise provided herein, the termination of Escrow shall not relieve or release either Party from any obligation to pay Escrow Holder's fees and costs or constitute a waiver, release or discharge of any breach or default that has occurred in the performance of the obligations, agreements, covenants or warranties contained therein.

8.10 If this Escrow is terminated for any reason other than Seller's breach or default, then as a condition to the return of Buyer's deposit, Buyer shall within 5 days after written request deliver to Seller, at no charge, copies of all surveys, engineering studies, soil reports, maps, master plans, feasibility studies and other similar items prepared by or for Buyer that pertain to the Property.

9.1 IF, BEFORE EXPIRATION OF THE APPLICABLE TIME, BUYER FAILS TO PROVIDE ESCROW HOLDER WRITTEN NOTICE OF BUYER'S DISAPPROVAL OF ANY OF

9. Contingencies to Closing.

BUYER'S CONTINGENCIES OR ANY OTHER MATTER THAT IS SUBJECT TO BUYER'S APPROVAL IN THIS AGREEMENT, THEN BUYER SHALL BE CONCLUSIVELY DEEMED
TO HAVE SATISFIED SUCH BUYER'S CONTINGENCIES AND/OR APPROVED OF SUCH OTHER MATTERS. If a number of days is completed in any of the optional spaces in subparagraphs 9.1 (a) through (m), then such number shall apply and override the pre-printed number, even if the pre-printed number is not stricken. The Closing
of this transaction is contingent upon the satisfaction or waiver of the following contingencies:
(a) Disclosure. Seller shall make to Buyer, through Escrow, all of the applicable disclosures required by law (See AIR CRE ("AIR") standard form entitled "Seller's Mandatory Disclosure Statement"; if the Property was built prior to 1978, then attach Lead-Based Paint Hazards Addendum to this Agreement) and provide
Buyer with a completed Property Information Sheet ("Property Information Sheet") concerning the Property, duly executed by or on behalf of Seller in the current
form or equivalent to that published by the AIR within 10 or days following the Date of Agreement. Buyer has 10 days from the receipt of said disclosures to approve or disapprove the matters disclosed.
(b) Physical Inspection. Buyer has 10 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever
is later, to satisfy itself with regard to the physical aspects and size of the Property. Seller recommends that the Buyer have the Property inspected with regard to the
possible presence of wood destroying pests and organisms by a licensed structural pest control company. Any such report shall be paid for by Buyer. (c) Hazardous Substance Conditions Report. Buyer has 30 or days following the receipt of the Property Information Sheet or the Date of
Agreement, whichever is later, to satisfy itself with regard to the environmental aspects of the Property. Seller recommends that Buyer obtain a Hazardous Substance
Conditions Report concerning the Property and relevant adjoining properties. Any such report shall be paid for by Buyer. A "Hazardous Substance" for purposes of this Agreement is defined as any substance whose nature and/or quantity of existence, use, manufacture, disposal or effect, render it subject to Federal, state or local regulation, investigation, remediation or removal as potentially injurious to public health or welfare. A "Hazardous Substance Condition" for purposes of this
Agreement is defined as the existence on, under or relevantly adjacent to the Property of a Hazardous Substance that would require remediation and/or removal under applicable Federal, state or local law.
(d) Soil Inspection. Buyer has 30 or days following the receipt of the Property Information Sheet or the Date of Agreement, whichever is
later, to satisfy itself with regard to the condition of the soils on the Property. Seller recommends that Buyer obtain a soil test report. Any such report shall be paid for
by Buyer. Seller shall provide Buyer copies of any soils report that Seller may have within 10 days following the Date of Agreement.
(e) Governmental Approvals. Buyer has 30 or days following the Date of Agreement to satisfy itself with regard to approvals and permits from governmental agencies or departments which have or may have jurisdiction over the Property and which Buyer deems necessary or desirable in connection with
its intended use of the Property, including, but not limited to, permits and approvals required with respect to zoning, planning, building and safety, fire, police,
handicapped and Americans with Disabilities Act requirements, transportation and environmental matters.
(f) Conditions of Title. Escrow Holder shall cause a current commitment for title insurance ("Title Commitment") concerning the Property issued by the Title Company, as well as legible copies of all documents referred to in the Title Commitment ("Underlying Documents"), and a scaled and dimensioned plot showing
the location of any easements to be delivered to Buyer within 10 or days following the Date of Agreement. Buyer has 10 days from the receipt of the
Title Commitment, the Underlying Documents and the plot plan to satisfy itself with regard to the condition of title. The disapproval by Buyer of any monetary
encumbrance, which by the terms of this Agreement is not to remain against the Property after the Closing, shall not be considered a failure of this contingency, as
Seller shall have the obligation, at Seller's expense, to satisfy and remove such disapproved monetary encumbrance at or before the Closing.
(g) Survey. Buyer has 30 or days following the receipt of the Title Commitment and Underlying Documents to satisfy itself with regard to any ALTA title supplement based upon a survey prepared to American Land Title Association ("ALTA") standards for an owner's policy by a licensed surveyor, showing
the legal description and boundary lines of the Property, any easements of record, and any improvements, poles, structures and things located within 10 feet of either
side of the Property boundary lines. Any such survey shall be prepared at Buyer's direction and expense. If Buyer has obtained a survey and approved the ALTA title
supplement, Buyer may elect within the period allowed for Buyer's approval of a survey to have an ALTA extended coverage owner's form of title policy, in which event
Buyer shall pay any additional premium attributable thereto.
(h) Existing Leases and Tenancy Statements. Seller shall within 10 or days following the Date of Agreement provide both Buyer and Escrow Holder with legible copies of all leases, subleases or rental arrangements (collectively, "Existing Leases") affecting the Property, and with a tenancy statement
(" Estoppel Certificate ") in the latest form or equivalent to that published by the AIR, executed by Seller and/or each tenant and subtenant of the Property. Seller shall use its best efforts to have each tenant complete and execute an Estoppel Certificate. If any tenant fails or refuses to provide an Estoppel Certificate then Seller shall
complete and execute an Estoppel Certificate for that tenancy. Buyer has 10 days from the receipt of said Existing Leases and Estoppel Certificates to satisfy itself with
regard to the Existing Leases and any other tenancy issues.
(i) Owner's Association. Seller shall within 10 or days following the Date of Agreement provide Buyer with a statement and transfer
package from any owner's association servicing the Property. Such transfer package shall at a minimum include: copies of the association's bylaws, articles of incorporation, current budget and financial statement. Buyer has 10 days from the receipt of such documents to satisfy itself with regard to the association.
(j) Other Agreements. Seller shall within 10 or days following the Date of Agreement provide Buyer with legible copies of all other
agreements ("Other Agreements") known to Seller that will affect the Property after Closing. Buyer has 10 days from the receipt of said Other Agreements to satisfy itself with regard to such Agreements.
(k) Financing. If paragraph 5 hereof dealing with a financing contingency has not been stricken, the satisfaction or waiver of such New Loan contingency.
(I) Existing and new Purchase Money Notes. If paragraph 3.1(c) has not been stricken, Seller shall within 10 or days following the Date of
Agreement provide Buyer with legible copies of the Existing Notes, Existing Deeds of Trust and related agreements (collectively, "Loan Documents") to which the Property will remain subject after the Closing. Escrow Holder shall promptly request from the holders of the Existing Notes a beneficiary statement ("Beneficiary
Statement") confirming: (1) the amount of the unpaid principal balance, the current interest rate, and the date to which interest is paid, and (2) the nature and
amount of any impounds held by the beneficiary in connection with such loan. Buyer has 10 or days following the receipt of the Loan Documents and
Beneficiary Statements to satisfy itself with regard to such financing. Buyer's obligation to close is conditioned upon Buyer being able to purchase the Property without acceleration or change in the terms of any Existing Notes or charges to Buyer except as otherwise provided in this Agreement or approved by Buyer, provided,
however, Buyer shall pay the transfer fee referred to in paragraph 3.2 hereof. Likewise if Seller is to carry back a Purchase Money Note then Seller shall within 10 or
days following the Date of Agreement provide Buyer with a copy of the proposed Purchase Money Note and Purchase Money Deed of Trust. Buyer has
days from the receipt of such documents to satisfy itself with regard to the form and content thereof.
(m) Personal Property. In the event that any personal property is included in the Purchase Price, Buyer has 10 or days following the Date of
Agreement to satisfy itself with regard to the title condition of such personal property. Seller recommends that Buyer obtain a UCC-1 report. Any such report shall be paid for by Buyer. Seller shall provide Buyer copies of any liens or encumbrances affecting such personal property that it is aware of within 10 or days
following the Date of Agreement.
(n) Destruction, Damage or Loss. Subsequent to the Date of Agreement and prior to Closing there shall not have occurred a destruction of, or damage or
loss to, the Property or any portion thereof, from any cause whatsoever, which would cost more than \$10,000.00 to repair or cure. If the cost of repair or cure is
\$10,000.00 or less, Seller shall repair or cure the loss prior to the Closing. Buyer shall have the option, within 10 days after receipt of written notice of a loss costing more than \$10,000.00 to repair or cure, to either terminate this Agreement or to purchase the Property notwithstanding such loss, but without deduction or offset
against the Purchase Price. If the cost to repair or cure is more than \$10,000.00, and Buyer does not elect to terminate this Agreement, Buyer shall be entitled to any
insurance proceeds applicable to such loss. Unless otherwise notified in writing, Escrow Holder shall assume no such destruction, damage or loss has occurred prior
to Closing. (a) Material Change Buyer shall have 10 days following receipt of written notice of a Material Change within which to satisfy itself with regard to such
(o) Material Change. Buyer shall have 10 days following receipt of written notice of a Material Change within which to satisfy itself with regard to such change. "Material Change" shall mean a substantial adverse change in the use, occupancy, tenants, title, or condition of the Property that occurs after the date of this offer and prior to the Closing. Unless otherwise notified in writing, Escrow Holder shall assume that no Material Change has occurred prior to the Closing.

INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
ROFA-11.30, Revised 10-12-2022

- (p) Seller Performance. The delivery of all documents and the due performance by Seller of each and every undertaking and agreement to be performed by Seller under this Agreement.
- (q) Brokerage Fee. Payment at the Closing of such brokerage fee as is specified in this Agreement or later written instructions to Escrow Holder executed by Seller and Brokers ("Brokerage Fee"). It is agreed by the Parties and Escrow Holder that Brokers are a third party beneficiary of this Agreement insofar as the Brokerage Fee is concerned, and that no change shall be made with respect to the payment of the Brokerage Fee specified in this Agreement, without the written consent of Brokers.
- 9.2 The contingencies specified in subparagraphs 9.1(a) through (m) are for the benefit of, and may be waived by, Buyer, and are referred to collectively as "Buyer's Contingencies" and individually as a "Buyer's Contingency."
- 9.3 Buyer's timely and written disapproval or conditional approval of a Buyer's Contingency or any other matter that is subject to Buyer's approval in this Agreement shall constitute disapproval thereof ("Disapproved Item(s)"). Concurrent with notice of a Disapproved Item, Buyer may make a request to Seller regarding such Disapproved Item ("Buyer's Request"). If Buyer fails to make a timely and written Buyer's Request, then this Agreement shall terminate due to the non-satisfaction and non-waiver of a contingency. Seller may respond to a Buyer's Request within 10 days following Seller's receipt thereof ("Seller's Response"). Seller's acceptance of a Buyer's Request shall amend this Agreement accordingly. If Seller fails to provide a timely and written Seller's Response, then Seller's Response shall be deemed to be a rejection of Buyer's Request. Buyer may, within 10 days following the earlier of Buyer's receipt of a Seller's Response (which is not an acceptance of Buyer's Request) or the date of Seller's deemed rejection of a Buyer's Request ("Buyer's Reply Period"), reply to a Seller's Response ("Buyer's Reply") and elect to (i) terminate this Agreement due to the non-satisfaction and non-waiver of the applicable contingency, (ii) accept the Seller's Response in which event this Agreement shall be amended accordingly, or (iii) withdraw Buyer's Request and waive the Disapproved Item in which event Buyer shall accept the Property subject to the Disapproved Item. If Buyer fails to provide a timely and written Buyer's Reply, then Buyer shall be deemed to have elected to terminate this Agreement as of the end of the Buyer's Reply Period. The date Buyer accepts a Seller's Response or withdraws a Buyer's Request and waives a Disapproved Item shall be the date of Buyer's approval of the Disapproved Item. A Party shall provide to Escrow Holder copy of all notices of a Disapproved Item, Buyer's Request, Seller's Response and Buyer's Reply and Escrow Holder shall promptly provide copies thereof to the other Party. Unless the Parties in writing agree otherwise, if the Expected Closing Date is a specific calendar date and a Buyer's Reply Period expires after such specific calendar date, then notwithstanding paragraph 1.1, the Expected Closing Date shall be extended to be 3 business days after the earlier of the date Buyer withdraws a Buyer's Request and waives the applicable Disapproved Item or Buyer accepts the applicable Seller's Response.
- 9.4 The Parties acknowledge that extensive local, state and Federal legislation establish broad liability upon owners and/or users of real property for the investigation and remediation of Hazardous Substances. The determination of the existence of a Hazardous Substance Condition and the evaluation of the impact of such a condition are highly technical and beyond the expertise of Brokers. The Parties acknowledge that they have been advised by Brokers to consult their own technical and legal experts with respect to the possible presence of Hazardous Substances on the Property or adjoining properties, and Buyer and Seller are not relying upon any investigation by or statement of Brokers with respect thereto. The Parties hereby assume all responsibility for the impact of such Hazardous Substances upon their respective interests herein.

10. Documents and Other Items Required at or Before Closing.

- 10.1 Five days prior to the Closing date Escrow Holder shall obtain an updated Title Commitment concerning the Property from the Title Company and provide copies thereof to each of the Parties.
 - 10.2 Seller shall deliver to Escrow Holder in time for delivery to Buyer at the Closing:
 - (a) Grant or general warranty deed, duly executed and in recordable form, conveying fee title to the Property to Buyer.
 - (b) If applicable, the Beneficiary Statements concerning Existing Note(s).
- (c) If applicable, the Existing Leases and Other Agreements together with duly executed assignments thereof by Seller and Buyer. The assignment of Existing Leases shall be on the most recent Assignment and Assumption of Lessor's Interest in Lease form published by the AIR or its equivalent.
- (d) An affidavit executed by Seller to the effect that Seller is not a "foreign person" within the meaning of Internal Revenue Code Section 1445 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Internal Revenue Service such sum as is required by applicable Federal law with respect to purchases from foreign sellers.
- (e) If the Property is located in California, an affidavit executed by Seller to the effect that Seller is not a "nonresident" within the meaning of California Revenue and Tax Code Section 18662 or successor statutes. If Seller does not provide such affidavit in form reasonably satisfactory to Buyer at least 3 business days prior to the Closing, Escrow Holder shall at the Closing deduct from Seller's proceeds and remit to the Franchise Tax Board such sum as is required by such statute.
 - f) If applicable, a bill of sale, duly executed, conveying title to any included personal property to Buyer.
 - (g) If the Seller is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the sale of the Property.
 - (h) Keys and/or other means to operate all locks, security systems, alarms, mailboxes and garage doors (collectively "Keys")
 - 10.3 Buyer shall deliver to Seller through Escrow:
- (a) The cash portion of the Purchase Price and such additional sums as are required of Buyer under this Agreement shall be deposited by Buyer with Escrow Holder, by federal funds wire transfer, or any other method acceptable to Escrow Holder in immediately collectable funds, no later than 2:00 P.M. on the business day prior to the Expected Closing Date provided, however, that Buyer shall not be required to deposit such monies into Escrow if at the time set for the deposit of such monies Seller is in default or has indicated that it will not perform any of its obligations hereunder. Instead, in such circumstances in order to reserve its rights to proceed Buyer need only provide Escrow with evidence establishing that the required monies were available.
- (b) If a Purchase Money Note and Purchase Money Deed of Trust are called for by this Agreement, the duly executed originals of those documents, the Purchase Money Deed of Trust being in recordable form, together with evidence of fire insurance on the improvements in the amount of the full replacement cost naming Seller as a mortgage loss payee, and a real estate tax service contract (at Buyer's expense), assuring Seller of notice of the status of payment of real property taxes during the life of the Purchase Money Note.
 - (c) The Assignment and Assumption of Lessor's Interest in Lease form specified in paragraph 10.2(c) above, duly executed by Buyer.
 - (d) Assumptions duly executed by Buyer of the obligations of Seller that accrue after Closing under any Other Agreements.
 - (e) If applicable, a written assumption duly executed by Buyer of the loan documents with respect to Existing Notes.
 - (f) If the Buyer is a corporation, a duly executed corporate resolution authorizing the execution of this Agreement and the purchase of the Property.
- 10.4 At Closing, Escrow Holder shall: (i) deliver all Keys to Buyer, and (ii) cause to be issued to Buyer a standard coverage (or ALTA extended, if elected pursuant to 9.1(g)) owner's form policy of title insurance effective as of the Closing, issued by the Title Company in the full amount of the Purchase Price, insuring title to the Property vested in Buyer, subject only to the exceptions approved by Buyer. In the event there is a Purchase Money Deed of Trust in this transaction, the policy of title insurance shall be a joint protection policy insuring both Buyer and Seller.

IMPORTANT: IN A PURCHASE OR EXCHANGE OF REAL PROPERTY, IT MAY BE ADVISABLE TO OBTAIN TITLE INSURANCE IN CONNECTION WITH THE CLOSE OF ESCROW SINCE THERE MAY BE PRIOR RECORDED LIENS AND ENCUMBRANCES WHICH AFFECT YOUR INTEREST IN THE PROPERTY BEING ACQUIRED. A NEW POLICY OF TITLE INSURANCE SHOULD BE OBTAINED IN ORDER TO ENSURE YOUR INTEREST IN THE PROPERTY THAT YOU ARE ACQUIRING.

11. Prorations and Adjustments.

- 11.1 Toxes. Applicable real property taxes and special assessment bonds shall be prorated through Escrow as of the date of the Closing, based upon the latest tax bill available. The Parties agree to prorate as of the Closing any taxes assessed against the Property by supplemental bill levied by reason of events occurring prior to the Closing. Payment of the prorated amount shall be made promptly in cash upon receipt of a copy of any supplemental bill.
- 11.2 Insurance. WARNING: Any insurance which Seller may have maintained will terminate on the Closing. Buyer is advised to obtain appropriate insurance to cover the Property.
- 11.3 Rentals, Interest and Expenses. Scheduled rentals, interest on Existing Notes, utilities, and operating expenses shall be prorated as of the date of Closing. The Parties agree to promptly adjust between themselves outside of Escrow any rents received after the Closing.
 - 11.4 Security Deposit. Security Deposits held by Seller shall be given to Buyer as a credit to the cash required of Buyer at the Closing.
- 11.5 Post Closing Matters. Any item to be prorated that is not determined or determinable at the Closing shall be promptly adjusted by the Parties by appropriate cash payment outside of the Escrow when the amount due is determined.
- 11.6 Variations in Existing Note Balances. In the event that Buyer is purchasing the Property subject to an Existing Deed of Trust(s), and in the event that a Beneficiary Statement as to the applicable Existing Note(s) discloses that the unpaid principal balance of such Existing Note(s) at the closing will be more or less than the amount set forth in paragraph 3.1(c) hereof ("Existing Note Variation"), then the Purchase Money Note(s) shall be reduced or increased by an amount equal to such Existing Note Variation. If there is to be no Purchase Money Note, the cash required at the Closing per paragraph 3.1(a) shall be reduced or increased by the amount of such Existing Note Variation.
- 11.7 Variations in New Loan Balance. In the event Buyer is obtaining a New Loan and the amount ultimately obtained exceeds the amount set forth in paragraph 5.1, then the amount of the Purchase Money Note, if any, shall be reduced by the amount of such excess.
 - 11.8 Owner's Association Fees. Escrow Holder shall: (i) bring Seller's account with the association current and pay any delinquencies or transfer fees from

INITIALS		INITIALS	
----------	--	----------	--

Seller's proceeds, and (ii) pay any up front fees required by the association from Buyer's funds.

12. Representations and Warranties of Seller and Disclaimers.

- 12.1 Seller's warranties and representations shall survive the Closing and delivery of the deed for a period of 3 years, and any lawsuit or action based upon them must be commenced within such time period. Seller's warranties and representations are true, material and relied upon by Buyer and Brokers in all respects. Seller hereby makes the following warranties and representations to Buyer and Brokers:
- (a) Authority of Seller. Seller is the owner of the Property and/or has the full right, power and authority to sell, convey and transfer the Property to Buyer as provided herein, and to perform Seller's obligations hereunder.
- (b) Maintenance During Escrow and Equipment Condition At Closing. Except as otherwise provided in paragraph 9.1(n) hereof, Seller shall maintain the Property until the Closing in its present condition, ordinary wear and tear excepted.
- (c) Hazardous Substances/Storage Tanks. Seller has no knowledge, except as otherwise disclosed to Buyer in writing, of the existence or prior existence on the Property of any Hazardous Substance, nor of the existence or prior existence of any above or below ground storage tank.
- (d) Compliance. Except as otherwise disclosed in writing, Seller has no knowledge of any aspect or condition of the Property which violates applicable laws, rules, regulations, codes or covenants, conditions or restrictions, or of improvements or alterations made to the Property without a permit where one was required, or of any unfulfilled order or directive of any applicable governmental agency or casualty insurance company requiring any investigation, remediation, repair, maintenance or improvement be performed on the Property.
- (e) Changes in Agreements. Prior to the Closing, Seller will not violate or modify any Existing Lease or Other Agreement, or create any new leases or other agreements affecting the Property, without Buyer's written approval, which approval will not be unreasonably withheld.
- (f) Possessory Rights. Seller has no knowledge that anyone will, at the Closing, have any right to possession of the Property, except as disclosed by this Agreement or otherwise in writing to Buyer.
 - (g) Mechanics' Liens. There are no unsatisfied mechanics' or materialmens' lien rights concerning the Property.
- (h) Actions, Suits or Proceedings. Seller has no knowledge of any actions, suits or proceedings pending or threatened before any commission, board, bureau, agency, arbitrator, court or tribunal that would affect the Property or the right to occupy or utilize same.
- (i) Notice of Changes. Seller will promptly notify Buyer and Brokers in writing of any Material Change (see paragraph 9.1(o)) affecting the Property that becomes known to Seller prior to the Closing.
- (j) No Tenant Bankruptcy Proceedings. Seller has no notice or knowledge that any tenant of the Property is the subject of a bankruptcy or insolvency proceeding.
 - (k) No Seller Bankruptcy Proceedings. Seller is not the subject of a bankruptcy, insolvency or probate proceeding.
- (I) Personal Property. Seller has no knowledge that anyone will, at the Closing, have any right to possession of any personal property included in the Purchase Price nor knowledge of any liens or encumbrances affecting such personal property, except as disclosed by this Agreement or otherwise in writing to Buyer.
- 12.2 Buyer hereby acknowledges that, except as otherwise stated in this Agreement, Buyer is purchasing the Property in its existing condition and will, by the time called for herein, make or have waived all inspections of the Property Buyer believes are necessary to protect its own interest in, and its contemplated use of, the Property. The Parties acknowledge that, except as otherwise stated in this Agreement, no representations, inducements, promises, agreements, assurances, oral or written, concerning the Property, or any aspect of the occupational safety and health laws, Hazardous Substance laws, or any other act, ordinance or law, have been made by either Party or Brokers, or relied upon by either Party hereto.
- 12.3 In the event that Buyer learns that a Seller representation or warranty might be untrue prior to the Closing, and Buyer elects to purchase the Property anyway then, and in that event, Buyer waives any right that it may have to bring an action or proceeding against Seller or Brokers regarding said representation or warranty.
- 12.4 Any environmental reports, soils reports, surveys, and other similar documents which were prepared by third party consultants and provided to Buyer by Seller or Seller's representatives, have been delivered as an accommodation to Buyer and without any representation or warranty as to the sufficiency, accuracy, completeness, and/or validity of said documents, all of which Buyer relies on at its own risk. Seller believes said documents to be accurate, but Buyer is advised to retain appropriate consultants to review said documents and investigate the Property.

13. Possession.

Possession of the Property shall be given to Buyer at the Closing subject to the rights of tenants under Existing Leases.

14. Buyer's Entry.

At any time during the Escrow period, Buyer, and its agents and representatives, shall have the right at reasonable times and subject to rights of tenants, to enter upon the Property for the purpose of making inspections and tests specified in this Agreement. No destructive testing shall be conducted, however, without Seller's prior approval which shall not be unreasonably withheld. Following any such entry or work, unless otherwise directed in writing by Seller, Buyer shall return the Property to the condition it was in prior to such entry or work, including the re-compaction or removal of any disrupted soil or material as Seller may reasonably direct. All such inspections and tests and any other work conducted or materials furnished with respect to the Property by or for Buyer shall be paid for by Buyer as and when due and Buyer shall indemnify, defend, protect and hold harmless Seller and the Property of and from any and all claims, liabilities, losses, expenses (including reasonable attorneys' fees), damages, including those for injury to person or property, arising out of or relating to any such work or materials or the acts or omissions of Buyer, its agents or employees in connection therewith.

15. Further Documents and Assurances.

The Parties shall each, diligently and in good faith, undertake all actions and procedures reasonably required to place the Escrow in condition for Closing as and when required by this Agreement. The Parties agree to provide all further information, and to execute and deliver all further documents, reasonably required by Escrow Holder or the Title Company.

16. Attorneys' Fees.

If any Party or Broker brings an action or proceeding (including arbitration) involving the Property whether founded in tort, contract or equity, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees and costs. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term "Prevailing Party" shall include, without limitation, a Party or Broker who substantially obtains or defeats the relief sought, as the case may be, whether by compromise, settlement, judgment, or the abandonment by the other Party or Broker of its claim or defense. The attorneys' fees award shall not be computed in accordance with any court fee schedule, but shall be such as to fully reimburse all attorneys' fees reasonably incurred.

17. Prior Agreements/Amendments.

- 17.1 This Agreement supersedes any and all prior agreements between Seller and Buyer regarding the Property.
- 17.2 Amendments to this Agreement are effective only if made in writing and executed by Buyer and Seller.

18. Broker's Rights.

18.1 If this sale is not consummated due to the default of either the Buyer or Seller, the defaulting Party shall be liable to and shall pay to Brokers the Brokerage Fee that Brokers would have received had the sale been consummated. If Buyer is the defaulting party, payment of said Brokerage Fee is in addition to any obligation with respect to liquidated or other damages.

18.2 Upon the Closing, Brokers are authorized to publicize the facts of this transaction.

19. Notices.

19.1 Whenever any Party, Escrow Holder or Brokers herein shall desire to give or serve any notice, demand, request, approval, disapproval or other communication, each such communication shall be in writing and shall be delivered personally, by messenger, or by mail, postage prepaid, to the address set forth in this agreement or by facsimile transmission, electronic signature, digital signature, or email.

19.2 Service of any such communication shall be deemed made on the date of actual receipt if personally delivered, or transmitted by facsimile transmission, electronic signature, digital signature, or email. Any such communication sent by regular mail shall be deemed given 48 hours after the same is mailed. Communications sent by United States Express Mail or overnight courier that guarantee next day delivery shall be deemed delivered 24 hours after delivery of the same to the Postal Service or courier. If such communication is received on a Saturday, Sunday or legal holiday, it shall be deemed received on the next business day.

19.3 Any Party or Broker hereto may from time to time, by notice in writing, designate a different address to which, or a different person or additional persons to whom, all communications are thereafter to be made.

20. Duration of Offer.

20.1 If this offer is not accepted by Seller on or before 5:00 P.M. according to the time standard applicable to the city of ______ on the date of

INITIALS
© 2019 AIR CRE. All Rights Reserved.

•	•		ment between the Parties as described in pa conditionally accepting the last outstanding	•
THE PARTIES AGREE THAT IT WOU WOULD BE SUFFERED BY SELLER I OF ALL CONTINGENCIES PROVIDE AMOUNT OF UPO	F BUYER FAILS TO PERFORM ITS OBLIGA D FOR THE BUYER'S BENEFIT, BUYER BR	DIFFICULT TO FIX, PRIOR TO S ATIONS UNDER THIS AGREEM EACHES THIS AGREEMENT, SE UYER SHALL BE RELEASED FRI	th Parties). IIGNING THIS AGREEMENT, THE ACTUAL DA ENT. THEREFORE, IF, AFTER THE SATISFACTI ELLER SHALL BE ENTITLED TO LIQUIDATED D OM ANY FURTHER LIABILITY TO SELLER, AN	ION OR WAIVER AMAGES IN THE
	Buyer's Initials		Seller's Initials	
22.1 ANY CONTROVERSY AS SHALL BE DETERMINED BY BINDIF ITS COMMERCIAL ARBITRATION RESUCH CONTROVERSY SHALL BE AS EXPERIENCE IN THE TYPE OF REAL ACCORDANCE WITH APPLICABLE AGREEMENT AND ANY AMENDM PERMITTED IN ACCORDANCE WITH AWARD WITHIN 30 DAYS AFTER TO PARAGRAPH 16 HEREOF AND SHALL AND LEGAL ARGUMENT AS THE ACCOMPETENT JURISDICTION NOTW 22.2 BUYER'S RESORT TO OF BUYER FOR DAMAGES AND/OR SPONICH EVENT SUCH AWARD SHALL 22.3 NOTICE: BY INITIALING OF DISPUTES" PROVISION DECIDED HAVE THE DISPUTE LITIGATED IN A APPEAL, UNLESS SUCH RIGHTS AR AGREEING TO THIS PROVISION, YOU TO THIS ARBITRATION PROVISION	ING ARBITRATION ADMINISTERED BY THE SULES ("COMMERCIAL RULES"). ARBITRATED BY A SINGLE ARBITRATOR, AND LESTATE THAT IS THE SUBJECT OF THIS A LAW OF THE JURISDICTION WHERE THE ENTS THERETO, AND UPON THE EVIDEN HIT THE COMMERCIAL RULES OR STATE LAME CONCLUSION OF THE HEARING, WHI LAW OF THE COMMERCIAL RULES OR STATE LAW OF COMPENSATION OR ADMINISTRATIVE OF COMPENSATION OR ADMINISTRATIVE OF COMPENSATION OR ADMINISTRATIVE OF A PARTY OF THE PROPERTY OF THE PROPER	IQUIDATED DAMAGES AND/O E JUDICIAL ARBITRATION & M ATION HEARINGS SHALL BE H PPOINTED UNDER THE COMM AGREEMENT. THE ARBITRATOI PROPERTY IS LOCATED, THE I ICE PRODUCED AT AN ARBITRAT ICH MAY INCLUDE ATTORNEY DPINION. THE FAILURE OR RE E CHARGES SHALL CONSTITU A DEFAULT JUDGMENT AGAI S AN AWARD. JUDGMENT MAY POCCEDINGS SHALL NOT BAR I'LL THE ARBITRATION RESULTS Y BUTCH FOR ANY DISPUTE ARIS ED BY CALIFORNIA LAW AND Y N THE SPACE BELOW YOU ARE RATION OF DISPUTES" PROVIS NDER THE AUTHORITY OF THE	OR BUYER IS ENTITLED TO THE RETURN OF THE BUYER IS ENTITLED TO THE RETURN OF THE BUYER IS ENTITLED TO THE RETURN OF THE BUYER IN ACCURED IN THE COUNTY WHERE THE PROPERTY MERCIAL RULES WHO HAS HAD AT LEAST 5 YER SHALL HEAR AND DETERMINE SAID CONTINITED OF THE PARTIES AS EXPRESSED MATION HEARING. PRE-ARBITRATION DISCOMINED PROCEEDINGS. THE ARBITRATOR SHALES OF A PARTY TO PAY SUCH PARTY'S RETE A WAIVER BY SUCH PARTY TO PRESENT IN THE ABSENTY OF THE NON-PAYING PARTY IN THE ABSENTY OF THE MATTERS INCLUDED IN THE OWN OF THE OWN OF THE MATTERS INCLUDED IN THE OWN OF THE OWN OF THE OWN OF THE OWN OWN OF THE OWN OF THE OWN	ORDANCE WITH Y IS LOCATED. YEARS OF ROVERSY IN IN THIS VERY SHALL BE LL RENDER AN RTY PER EQUIRED SHARE EVIDENCE OR NCE OF EVIDENCE JIST OF CTION BY THE ED DAMAGES, IN HE "ARBITRATION T POSSESS TO COVERY AND RATION AFTER YOUR AGREEMENT
WE HAVE READ AND UNDERSTAND PROVISION TO NEUTRAL ARBITRAT		IIT DISPUTES ARISING OUT OF	THE MATTERS INCLUDED IN THE "ARBITRAT	TON OF DISPUTES
			*	
	Buyer's Initials	<u> </u>	Seller's Initials	
Paragraphs 21 and 22 are each inc Agreement accomplished by mear 23.2 Applicable Law; Venue	orporated into this Agreement only if inings of electronic signature or similar technors. This Agreement shall be governed by, a	tialed by both Parties at the ti nology shall be legal and bindi and paragraph 22.3 is amende	not paragraphs 21 and 22 are initialed by bo me that the Agreement is executed. Signati ng. ed to refer to, the laws of the state in which is iated in the county in which the Property is	ures to this the Property is

- - 23.3 **Time of Essence**. Time is of the essence of this Agreement.

. it shall be deemed automatically revoked.

- 23.4 Counterparts. This Agreement may be executed by Buyer and Seller in counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument. Escrow Holder, after verifying that the counterparts are identical except for the signatures, is authorized and instructed to combine the signed signature pages on one of the counterparts, which shall then constitute the Agreement.
- 23.5 Waiver of Jury Trial. THE PARTIES HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING INVOLVING THE PROPERTY OR ARISING OUT OF THIS AGREEMENT.
- 23.6 Conflict. Any conflict between the printed provisions of this Agreement and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions. Seller and Buyer must initial any and all handwritten provisions.
- 23.7 1031 Exchange. Both Seller and Buyer agree to cooperate with each other in the event that either or both wish to participate in a 1031 exchange. Any party initiating an exchange shall bear all costs of such exchange. The cooperating Party shall not have any liability (special or otherwise) for damages to the exchanging Party in the event that the sale is delayed and/or that the sale otherwise fails to qualify as a 1031 exchange.
- 23.8 Equal Housing Opportunity. The Property is being sold in compliance with local, state and federal anti-discrimination laws.
 23.9 NOTICE REGARDING GAS AND HAZARDOUS LIQUID TRANSMISSION PIPELINES. This notice is being provided simply to inform you that information about the general location of gas and hazardous liquid transmission pipelines is available to the public via the National Pipeline Mapping System (NPMS) Internet Web site maintained by the United States Department of Transportation at http://www.npms.phmsa.dot.gov/. To seek further information about possible transmission pipelines near the property, you may contact your local gas utility or other pipeline operators in the area. Contact information for pipeline operators is searchable by ZIP Code and county on the NPMS Internet Web site.
- Disclosures Regarding The Nature of a Real Estate Agency Relationship.
- 24.1 The Parties and Brokers agree that their relationship(s) shall be governed by the principles set forth in the applicable sections of the California Civil Code, as summarized in paragraph 24.2.
- 24.2 When entering into a discussion with a real estate agent regarding a real estate transaction, a Buyer or Seller should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Buyer and Seller acknowledge being advised by the Brokers in this transaction, as follows:
- (a) Seller's Agent. A Seller's agent under a listing agreement with the Seller acts as the agent for the Seller only. A Seller's agent or subagent has the following affirmative obligations: (1) To the Seller: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Seller. (2) To the Buyer and the Seller: a. Diligent exercise of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith. c. A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the affirmative duties set
- Buyer's Agent. A selling agent can, with a Buyer's consent, agree to act as agent for the Buyer only. In these situations, the agent is not the Seller's agent, even if by agreement the agent may receive compensation for services rendered, either in full or in part from the Seller. An agent acting only for a Buyer has the following affirmative obligations. (1) To the Buyer: A fiduciary duty of utmost care, integrity, honesty, and loyalty in dealings with the Buyer. (2) To the Buyer and the Seller: a. Diligent exercise of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith. c. A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the

INITIALS	INITIALS
© 2019 AIR CRE. All Rights Reserved.	

affirmative duties set forth above.

- (c) Agent Representing Both Seller and Buyer. A real estate agent, either acting directly or through one or more associate licensees, can legally be the agent of both the Seller and the Buyer in a transaction, but only with the knowledge and consent of both the Seller and the Buyer. (1) In a dual agency situation, the agent has the following affirmative obligations to both the Seller and the Buyer: a. A fiduciary duty of utmost care, integrity, honesty and loyalty in the dealings with either Seller or the Buyer. b. Other duties to the Seller and the Buyer as stated above in their respective sections (a) or (b) of this paragraph 24.2. (2) In representing both Seller and Buyer, the agent may not, without the express permission of the respective Party, disclose to the other Party confidential information, including, but not limited to, facts relating to either Buyer's or Seller's financial position, motivations, bargaining position, or other personal information that may impact price, including Seller's willingness to accept a price less than the listing price or Buyer's willingness to pay a price greater than the price offered. (3) The above duties of the agent in a real estate transaction do not relieve a Seller or Buyer from the responsibility to protect their own interests. Buyer and Seller should carefully read all agreements to assure that they adequately express their understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, consult a competent professional. Buyer has the duty to exercise reasonable care to protect Buyer, including as to those facts about the Property which are known to Buyer or within Buyer's diligent attention and observation. Both Seller and Buyer should strongly consider obtaining tax advice from a competent professional because the federal and state tax consequences of a transaction can be complex and subject to change.
- (d) Further Disclosures. Throughout this transaction Buyer and Seller may receive more than one disclosure, depending upon the number of agents assisting in the transaction. Buyer and Seller should each read its contents each time it is presented, considering the relationship between them and the real estate agent in this transaction and that disclosure. Buyer and Seller each acknowledge receipt of a disclosure of the possibility of multiple representation by the Broker representing that principal. This disclosure may be part of a listing agreement, buyer representation agreement or separate document. Buyer understands that Broker representing Buyer may also represent other potential buyers, who may consider, make offers on or ultimately acquire the Property. Seller understands that Broker representing Seller may also represent other sellers with competing properties that may be of interest to this Buyer. Brokers have no responsibility with respect to any default or breach hereof by either Party. The Parties agree that no lawsuit or other legal proceeding involving any breach of duty, error or omission relating to this transaction may be brought against Broker more than one year after the Date of Agreement and that the liability (including court costs and attorneys' fees), of any Broker with respect to any breach of duty, error or omission relating to this Agreement shall not exceed the fee received by such Broker pursuant to this Agreement; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker
- 24.3 Confidential Information. Buyer and Seller agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party to be confidential.
- 25. Construction of Agreement. In construing this Agreement, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Agreement. Whenever required by the context, the singular shall include the plural and vice versa. This Agreement shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.

26.	^ 44:4:4	Provisions.

Additional provisions of this offer, if any, are as follows or are attached hereto by an addendum or addenda consisting of paragraphs ______ through ______ (If there are no additional provisions write "NONE".)

ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS AGREEMENT OR THE TRANSACTION TO WHICH IT RELATES. THE PARTIES ARE URGED TO:

- SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS AGREEMENT.
- 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PROPERTY. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF WOOD DESTROYING PESTS OR ORGANISMS AND/OR HAZARDOUS SUBSTANCES, THE ZONING OF THE PROPERTY, THE INTEGRITY AND CONDITION OF ANY STRUCTURES AND OPERATING SYSTEMS, WHETHER OR NOT THE PROPERTY MEETS CURRENT CODES AND WHETHER OR NOT THE PROPERTY IS SUBJECT TO RENT CONTROL OR OTHER LOCAL RENTAL PROPERTY REGULATIONS.

Date:

WARNING: IF THE PROPERTY IS LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THIS AGREEMENT MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PROPERTY IS LOCATED.

NOTE:

- 1. THIS FORM IS NOT TO BE USED WITH PROPERTIESS CONTAINING LESS THAN 5 RESIDENTIAL UNITS.
- 2. IF EITHER PARTY IS A CORPORATION, IT IS RECOMMENDED THAT THIS AGREEMENT BE SIGNED BY TWO CORPORATE OFFICERS.

The undersigned Buyer offers and agrees to buy the Property on the terms and conditions stated and acknowledges receipt of a copy hereof.

BROKER	BUYER
Attn:	Ву:
Title:	Name Printed:
	Title:
Address:	Phone:
Phone:	Fax:
Fax:	Email:
Email:	
Federal ID No.:	By:
Broker DRE License #:	By: Name Printed:
Agent DRE License #:	Title:
	Phone:
	Fax:
\ \	Email:
	Address:
V	Federal ID No.:
27. Acceptance.	
•	and hereby agrees to sell the Property to Buyer on the terms and conditions therein specified.
27.2 In consideration of real estate brokerage service rendered b	y Brokers, Seller agrees to pay Brokers a real estate Brokerage Fee in a sum equal to
% of the Purchase Price to be divided between the Broke	rs as follows: Seller's Broker % and Buyer's Broker %. This
•	p pay such Brokerage Fee to Brokers out of the proceeds accruing to the account of Seller at the
Closing.	
27.3 Seller acknowledges receipt of a copy hereof and authorizes	Brokers to deliver a signed copy to Buyer.
NOTE: A DRODERTY INCORMATION SHEET IS REQUIRED TO BE DELIVER	RED TO RUYER BY SELLER LINDER THIS AGREEMENT

INITIALS

INITIALS
© 2019 AIR CRE. All Rights Reserved.
ROFA-11.30, Revised 10-12-2022

	Date:
BROKER	SELLER
Attn: Title:	By:
nue	Name Printed: Title:
Address:	Phone:
Phone:	Fax:
Fax: Email:	Email:
Federal ID No.:	
Broker DRE License #:	By:
Agent's DRE License #:	Name Printed: Title:
	Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
	* 213-687-8777 * contracts@aircre.com
NOTICE: No part of these works may be repr	oduced in any form without permission in writing.
	<) \
	() '
	X
() \	
	,
() '	
X	
() \	
'	
1	

Limited liability company: Manager(s)

Husband and wife: Community Property or Separate property

Joint tenants: All joint tenants

Tenants in common: All tenants in common

Corporation:

Board chairman, president or any vice president

and

Secretary, asst. secretary, CFO or asst. treasurer

General partnership: General partner with authority

Limited Partnership: General partner

Para. 9.1	Contingency topic	Time Seller has to deliver	Time Buyer has to approve
(a)	Seller's Mandatory Disclosure Statement and Property Information Sheet	to Escrow within 10 days after Date of Agreement ("DOA")	10 days after Buyer's receipt
(b)	Physical Inspection		10 days after the later of Buyer's receipt of Property Information Sheet or DOA
(c)	Hazardous Substance Condition Report		30 days after later of receipt of Property Information Sheet or DOA
(d)	Soil Inspection	within 10 days after DOA	30 days after later of receipt of Property Information Sheet or DOA
(e)	Governmental approvals		30 days after DOA
(f)	Title	within 10 days of DOA, Escrow to deliver Title Commitment, Underlying Documents and scaled and dimensioned plot of easements	10 days after Buyer's receipt of Title Commitment, Underlying Documents and plot plan
(g)	ALTA Survey		30 days after Buyer's receipt of Title Commitment and Underlying Documents
(h)	Leases and Estoppels	within 10 days after DOA	10 days after Buyer's receipt
(i)	Owner's Association	within 10 days after DOA	10 days after Buyer's receipt
(j)	Other Agreements	within 10 days after DOA	10 days after Buyer's receipt
(k)	New Loan Financing		[] after DOA
(1)	Existing Notes	within 10 days after DOA provide copies of loan documents	10 days after Buyer's receipt of loan documents and beneficiary statement
(m)	Personal Property	within 10 days after DOA provide copies of liens	10 days after DOA



BUYER'S NOTICE DISAPPROVING OF CONTINGENCIES

Date of Notice:
Buyer's Disapproval Notice Number:
Escrow Holder:
In accordance with the terms, conditions and provisions of the Standard Offer, Agreement And Escrow Instructions For Purchase Of Real Estate dated for reference purposes only as of (as amended) ("Agreement") by and between ("Buyer") and ("Seller") regarding the property commonly known as ("Property"), the defined terms of which Agreement shall have the same meaning in this Notice, Buyer disapproves of the following contingencies (check all that apply):
Disclosures (Para. 9.1(a))
Physical Inspection (Para. 9.1(b))
Hazardous Substance Conditions Report (Para. 9.1(c))
Soil Inspection (Para. 9.1(d))
Governmental approvals (Para. 9.1(e))
Conditions of Title (Para. 9.1(f))
ALTA Survey (Para. 9.1(g))
Existing Leases and Tenancy Statements (Para. 9.1(h))
Owner's Association (Para. 9.1(i))
Other Agreements (Para. 9.1(j))
Financing (Para. 5 and 9.1(k))
Existing and New Purchase Money Notes (Para. 9.1(I))
Personal Property (Para. 9.1(m))
All of the above contingencies
All of the above contingencies except (add description):
Other matter that is subject to Buyer's approval (add description):
Buyer requests ("Buyer's Request(s)") Seller to (check and complete as appropriate):
At Seller's cost, make the following repairs, replacements and/or take the following actions by the following dates (describe):
Reduce the Purchase Price by the amount of (insert amount):
At the Closing, credit Buyer the amount of (insert amount):
Extend time for the following contingencies by the following periods of time (describe contingencies and extended periods of time):
Extend the Expected Closing Date as follows (insert):
Other (add description):
BUYER
Dug.
By: Name Printed:
Title:
AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com
NOTICE: No part of these works may be reproduced in any form without permission in writing.

NITIALS
© 2022 AIR CRE. All Rights Reserved.
BNDC-1.00, Revised 10-12-2022

INITIALS



SELLER'S RESPONSE TO BUYER'S REQUESTS

BUYER'S REQUESTS
Responding to Buyer's Request Number: Date of Seller's Response: Escrow Holder:
In accordance with the terms, conditions and provisions of the Standard Offer, Agreement And Escrow Instructions For Purchase Of Real Estate dated for reference purposes only as of (as amended) ("Agreement") by and between ("Buyer") and ("Seller") regarding the property commonly known as ("Property"), the defined terms of which Agreement shall have the same meaning in this Notice, Seller responds to the above-referenced Buyer's Request as follows (check and complete as appropriate): Seller Denies Buyer's Request. Seller agrees to the entirety of Buyer's Request; the Agreement is amended accordingly. Seller does not agree to Buyer's Request except the following parts (check all that apply): At Seller's cost, make the following repairs, replacements and/or take the following actions by the following dates (describe): Reduce the Purchase Price by the amount of (insert amount) At the Closing, credit Buyer the amount of (insert amount) Extend time for Buyer's approval of the following contingency by the following period of time (describe contingency and period of time): Extend Expected Closing Date as follows (insert): Other (add description):
SELLER
By:Name Printed:
Title:
AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2022 AIR CRE. All Rights Reserved.
SRBR-1.00, Revised 10-12-2022



BUYER'S REPLY TO SELLER'S RESPONSE

IO SELLER 5 RESPONSE
Buyer's Request Number: Date of Buyer's Reply: Escrow Holder:
In accordance with the terms, conditions and provisions of the Standard Offer, Agreement And Escrow Instructions For Purchase Of Real Estate dated for reference purposes only as of (as amended) ("Agreement") by and between ("Buyer") and ("Seller") regarding the property commonly known as ("Property"), the defined terms of which Agreement shall have the same meaning in this Notice, Buyer replies ("Buyer's Reply Seller's Response to the above-referenced Buyer's Request as follows (select one only):
Buyer accepts Seller's Response; the Agreement is amended accordingly.
Buyer withdraws Buyer's Request and waives the applicable Disapproved Item; Buyer accepts the Property subject to such waived Disapproved Item. Buyer elects to terminate the Agreement due to the non-satisfaction and non-waiver of the applicable contingency.
BUYER
By: Name Printed:
Title:
AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS



CONFIDENTIALITY AGREEMENT

This Agreement ("Agreement") is made and entered into by and between	("Owner") and ("Potential Buyer") and shall be effective as of
WHEREAS, Owner owns certain real property commonly known as (street address, city,	state, zip) ("Property"), and
WHEREAS, Potential Buyer is considering acquiring the Property from Owner, and	
WHEREAS, in order to evaluate whether or not to acquire the Property Potential Buyer documents, $ \\$	wants Owner to provide Potential Buyer with certain information and/or
NOW THEREFORE, in consideration of the promises contained herein and other good at acknowledged Owner and Potential Buyer agree as follows:	nd valuable consideration, the receipt and sufficiency of which is hereby
1. Agreement to Provide Information. Owner agrees to make available to Potential concerning the Property. As a condition to furnishing such information and documents Buyer agrees to treat such information in accordance with the provisions of this Agreemany person or entity, collectively all of such person's or entity's officers, directors, empladvisors, including, without limitation, attorneys, accountants, consultants, bankers and	to Potential Buyer and its "Representatives" (as defined below), Potential ment. As used herein the term "Representatives" shall mean with respect to oyees, affiliates, partners, members, representatives, agents, brokers or
2. Definition of Evaluation Material. The term "Evaluation Material" means all infor representatives provides to Potential Buyer or to its Representatives and all notes, anal Potential Buyer or its Representatives that contain, reflect or are based upon, in whole pursuant hereto; provided, however, that the foregoing shall not require Potential Buyer compilations, studies, interpretations or other documents prepared by Potential Buyer information which is generally available to the public.	yses, compilations, studies, interpretations or other documents prepared by or in part, the information furnished to Potential Buyer or its Representatives er to reveal or disclose to Owner in any manner any such notes, analyses,
3. Use of Evaluation Material and Confidentiality.	< _ \
(a) Potential Buyer hereby agrees that it and its Representatives shall use the Event Property, that the Evaluation Material shall be kept confidential and that Potential Buyer any manner whatsoever and, that such information shall be disclosed only to such of Posole purpose of evaluating the purchase of the Property and who are informed that such for any disclosure by any of its Representatives of any Evaluation Materials which if managrees, at its sole expense, to take all reasonable measures to prevent or restrain its Re Evaluation Material. In no event shall Potential Buyer or its Representatives use the Evaluation Material is strictly prohibited.	otential Buyer's Representatives who need to know such information for the chinformation is confidential. In any event, Potential Buyer shall be responsible de by Potential Buyer would be a breach of this Agreement, and Potential Buyer presentatives from prohibited or unauthorized disclosure or use of the
(b) Potential Buyer agrees that, without the prior written consent of Owner, it a Evaluation Material has been made available to Potential Buyer, the fact that discussion any of the terms, conditions or other matters then being discussed with respect thereto Property's tenants, partners, members, employees, contractors, managers, insurers, managers, mana	o (including the status thereof), and (ii) shall not contact any of Owner's or the
(c) If Potential Buyer decides that it does not wish to proceed with the purchase upon the request of Owner for any reason, Potential Buyer and shall promptly deliver to Representatives by or on behalf of Owner pursuant hereto. At such time Potential Buyer Potential Buyer or its Representatives and no copy thereof shall be retained. Notwithst and its Representatives shall continue to be bound by Potential Buyer's obligations of continue to be bound by Potential Buyer's obligations.	er shall also destroy or deliver to Owner all Evaluation Material prepared by tanding the return or destruction of the Evaluation Material, Potential Buyer
4. Accuracy of Evaluation Material . Potential Buyer understands and acknowledges Seller and those representing Potential Buyer) are not making any representation or wa Material. Potential Buyer agrees that Owner, its Representatives and/or the Brokers sh relating to or resulting from the use of the Evaluation Material or any errors therein or made in a final definitive agreement regarding the purchase, as, when and if executed a	arranty, express or implied, as to the accuracy or completeness of the Evaluation all not have any liability to Potential Buyer or to any of its Representatives omissions therefrom. Only those representations or warranties which are
5. Remedies . It is understood and agreed that money damages would not be a suffi Representatives and that Owner shall be entitled to equitable relief, including injunctio shall not be deemed to be the exclusive remedies for a breach by Potential Buyer or an remedies available at law or equity to Owner.	ns and specific performance, as a remedy for any such breach. Such remedies
6. Miscellaneous.	
(a) Waivers and Amendments. No failure or delay by Owner in exercising any ri any single or partial exercise thereof preclude any other or future exercise thereof or the may be amended only with the written consent of Owner and Potential Buyer.	
(b) Effect of Agreement. Neither Potential Buyer nor Owner nor any of their respect to conducting negotiations relating to or consummating the purchase of the Prodiscretion, to reject any and all proposals made by Potential Buyer with regard to purch into a definitive agreement, with regard to any other transaction, with any other person discussions and negotiations with Potential Buyer at any time.	operty by virtue of this Agreement. Owner reserves the right, in its sole hasing the Property and to engage in discussions and negotiations, and to enter
(c) Governing Law. The parties hereby agree that this Agreement shall be gove Property is located. The parties hereby irrevocably consent to the exclusive jurisdiction actions, suits or proceedings arising out of or relating to this Agreement.	rned by and construed in accordance with the laws of the State in which the of the state courts located in the State in which the Property is located for any
(d) Severability. If any provision of this Agreement is found to violate any statut agency or exchange, such invalidity shall not be deemed to affect any other provision h provision shall be deemed deleted herefrom to the minimum extent necessary to cure	ereof or the validity of the remainder of this Agreement, and such invalid
(e) Counterparts. This Agreement may be executed in multiple counterparts, eagreement. Telecopied and PDF signatures may be used in place of original signatures such telecopied document or PDF copies. Signatures to this Agreement accomplished binding.	on this Agreement, and the parties intend to be bound by the signatures on
(f) Entire and Sole Agreement. This Agreement constitutes the entire understa hereof and supersedes any and all prior or contemporaneous oral or written communic	nding and agreement between the parties with respect to the subject matter cations with respect hereto, all of which are merged herein.
WHEREFORE, by their signature below, the parties hereto acknowledge that they have legally binding Agreement.	reviewed carefully the terms of this Agreement and they understand it is a

INITIALS

© 2017 AIR CRE. All Rights Reserved. CA-1.02, Revised 10-22-2020

POTENTIAL BUYER:	OWNER:
Ву:	Ву:
Name Printed:	Name Printed:
Title:	Title:

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS



COUNTER OFFER - SELLER

Dated:			
By and Between			
Seller:	<u></u>		
Buyer:	<u></u>		
Property Address:	_		
rioperty Address.	(street address, city, state, zip)		
APN:	(
Arn.			
his is a Counter Offer to the "Standard Off hereinafter the "Offer"). A full and comple	=		, regarding the above-referenced Property and Parties by reference.
Seller hereby accepts the Offer provide	ded the Buyer agrees to the following ch	anges in the Offer:	
(Please check the appropriate box or	boxes)		
Initialed pen and ink changes have	ve been made to the Offer.		
Seller shall be deemed to have in	nitialed paragraph 21 regarding liquidate	ed damages and paragra	ph 22 regarding arbitration of disputes
The changes are detailed in the a	addendum attached hereto consisting of	paragraphs	through
The changes are set forth below.	·		~ X
-			
			
NOTE: PARAGRAPHS IN THE OFFER WHICH JNLESS SPECIFICALLY REFERENCED FOR INC		ARE NOT INITIALED BY	ALL PARTIES, ARE EXCLUDED FROM THE FINAL AGREEMENT
	and conditions contained in the Offer w	hich have not been spe	cifically modified or deleted by this Counter Offer are
nereby approved by Seller.			
(Seller's Initials:) (If checked and initialed) MUITIPI	E COLINTER OFFER: Co.	ncurrently herewith, Seller is also making a Counter Offer to
	may or may not be the same as in this C	Counter Offer. Acceptan	ice of this Counter Offer by Buyer shall not, therefore, be
RIGHT TO ACCEPT OTHER OFFERS: Se	eller reserves the right to continue to off	fer the Property for sale	, and to accept any other offer at any time prior to actual
eceipt by Seller's authorized agent,			ed by Buyer and, if applicable, the re-signing of this Counter er Offer.
EXPIRATION: This Counter Offer shall	ll expire and be revoked, without further	notice, at 5:00 pm on	, unless it is accepted and signed by Buyer and
	y Seller's agent (as specified in paragraph	// h	Upon expiration, Buyer's Deposit, if any, shall be returned.
5. TIME : Time is of the essence.			
RINDING FFFFCT: This Counter Offer	when fully executed by both Buyer and	Seller, and delivered an	d received as specified above, shall be a binding contract.
ignatures to this Counter Offer accomplish RECOMMENDATION IS MADE BY AIR CRE O	hed by means of electronic signature or sort BY ANY BROKER AS TO THE LEGAL SUF	similar technology shall FFICIENCY, LEGAL EFFEC	be legal and binding. NO REPRESENTATION OR T, OR TAX CONSEQUENCES OF THIS AGREEMENT OR THE LEGAL AND TAX CONSEQUENCES OF THIS AGREEMENT.
NOTE:			
1. THIS FORM IS NOT FOR USE IN O	CONNECTION WITH THE SALE OF RESIDE TION, IT IS RECOMMENDED THAT THIS AG		BY TWO OF ITS CORPORATE OFFICERS.
The undersigned Seller agrees to sell	the Property on the terms and condition	ns stated herein.	
ELLER'S BROKER		SELLER	
ELLEN 3 BROKEN		JELLEN	
Attn:		By:	
Title:		Name Printed:	
Address:		Title:	
Phone:		Phone:	_
ax:		Fax:	
mail:		Email:	
ederal ID No.:			
roker DRE License #:		By:	
Agent DRE License #:		Name Printed:	
		Title:	
)		Phone:	_
		Fax:	
\		Email:	
		Address:	
		Federal ID No.:	
ACCEPTANCE: Buyer accepts the fore leliver a signed copy to Seller.	egoing Counter Offer and agrees to purch	nase the Property on the	e terms and conditions specified, and authorizes Brokers to
BUYER'S BROKER		BUYER	
O I LN 3 DNORLN		DOILN	
Attn:		By:	
ītle:		Name Printed:	
NITIALS		INITIALS	
© 2017 AIR CRE. All Rights Reserved.			Last Edited: 9/30/2022 2:03 PM

COS-3.02, Revised 10-22-2020

Address:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Federal ID No.:	
Broker DRE License #:	By: Name Printed:
Agent Bite Election II.	Title:
	Phone:
	Fax:
	Email:
	Address:
	Federal ID No.:
10. ACKNOWLEDGMENT OF RECEIPT am/pm.	: Seller's agent hereby acknowledges that a copy of the Counter Offer executed by Buyer was received on at
	By: Name Printed:
	Name Finiteu.
	ATURE LINE: (Applicable only if paragraph 3 is checked and initialed.) By signing below, Seller accepts this Multiple Counter Note to Seller: Do NOT sign in this paragraph until after Buyer has accepted the Counter Offer by signing in paragraph 9).
SELLER	
- -	
	Ву:
	Name Printed:
	Title:
	Phone: Fax:
	Email:
	Ву:
	Name Printed:
	Title:
	Phone:
	Fax: Email:
NOT	AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com ICE: No part of these works may be reproduced in any form without permission in writing.



COUNTER OFFER - BUYER

Dated:			
By and Between Seller:			
Buyer:			
Property Address:			
APN:	(street address, city, state, zip)		
APN.			
	Thereafter, Seller executed and deliv	chase of Real Estate" dated, regarding the ered to Buyer the "Counter Offer - Seller" dated rated herein by reference.	
Buyer hereby accepts Seller's Counter Of	fer provided the Seller agrees to the f	ollowing changes in said Counter Offer:	
(Please check the appropriate box or box	es)		./ \
Initialed pen and ink changes have b	een made to the Offer.		
Seller shall be deemed to have initia	led paragraph 21 regarding liquidated	d damages and paragraph 22 regarding arbitration of di	sputes
The changes are detailed in the adde	endum attached hereto consisting of	paragraphs through	X
The changes are set forth below.			
-			
	OURSE INITIALS BY ALL BARTIES BUT I		
UNLESS SPECIFICALLY REFERENCED FOR INCLU		ARE NOT INITIALED BY ALL PARTIES, ARE EXCLUDED FRO	JM THE FINAL AGREEMENT
2. REMAINING TERMS : All of the terms an Counter Offer are hereby approved by Buyer.	d conditions contained in the Seller's	Counter Offer which have not been specifically modified	ed or deleted by this
3. EXPIRATION : This Counter Offer shall exp			red and signed by Seller and
a fully executed copy is actually received by Bu Counter Offer may be executed in counterparts		d time. Upon expiration, Buyer's Deposit, if any, shall b	e promptly returned. This
4. TIME : Time is of the essence.			
5. BINDING EFFECT : This Counter Offer, wh	en fully executed by both Buyer and S	Seller, and delivered and received as specified above, sl	hall be a binding contract.
RECOMMENDATION IS MADE BY AIR CRE OR B	Y ANY BROKER AS TO THE LEGAL SUFF	imilar technology shall be legal and binding. NO REPRE FICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF TH	HIS AGREEMENT OR THE
TRANSACTION TO WHICH IT RELATES. THE PAR	RTIES ARE ADVISED TO SEEK ADVICE O	F COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCE	S OF THIS AGREEMENT.
NOTE:	INECTION WITH THE SALE OF RESIDEN	ITIAL DEODERTY	
2. IF EITHER PARTY IS A CORPORATION	N, IT IS RECOMMENDED THAT THIS AG	REEMENT BE SIGNED BY TWO OF ITS CORPORATE OFF	ICERS.
6. The undersigned Buyer agrees to purchase	se the Property on the terms and con-	ditions stated herein.	
BUYER'S BROKER		BUYER	
Attn:		By:	
Title:		Name Printed:	
Address:	•	Title:	
Phone:		Phone:	
Fax:		Fax: Email:	
Email:			
Federal ID No.: Broker DRE License #:		Ву:	
Agent DRE License #:		Name Printed:	
		Title:	
		Phone: Fax:	
		Email:	
//'			
\nearrow		Address:	
		Federal ID No.:	
7. ACCEPTANCE : Seller accepts the foregoin a signed copy to Buyer.	ng Counter Offer and agrees to sell the	e Property on the terms and conditions specified, and a	authorizes Brokers to deliver
SELLER'S BROKER		SELLER	
Atta			
Attn: Title:		By:	
		Name Printed: Title:	
Address:		Phone:	
Phone:		Fax:	
Fax: Email:		Email:	
INITIALS		INITIALS	
HINITIALS		HINTHALA	

© 2017 AIR CRE. All Rights Reserved. COB-2.02, Revised 10-22-2020

By:
Name Printed:
Title:
Phone:
Fax:
Email:
Address:
Federal ID No.:
a copy of the Counter Offer executed by Seller was received on at
By: Name Printed:
Numer integral
* 212 CO7 C777 * contract Onlines
* 213-687-8777 * contracts@aircre.com oduced in any form without permission in writing.
•



ADDENDUM TO THE STANDARD OFFER, AGREEMENT AND ESCROW INSTRUCTIONS FOR PURCHASE OF REAL ESTATE

By and Between	
Buyer:	
Seller:	_
	_
Property Address:	(street address situ state 7in)
	(street address, city, state, zip)
"Agreement"). In the event of any conflict	of the above-referenced Agreement (said Agreement and the Addendum are hereinafter collectively referred to as the between the provisions of this Addendum shall control. AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2017 AIR CRE. All Rights Reserved.
APA-1.03, Revised 10-22-2020



_____ AMENDMENT TO PURCHASE AND SALE AGREEMENT

THIS AMENDMENT TO PURCHASE AND SALE AGREEMENT AND ESCROW INSTRUCTI between ("Buyer") and ("Seller") (collectively, the "Prate of the collection	•
WHEREAS, Buyer and Seller are parties to that certain Standard Offer, Agreement as ("Agreement") with respect to the property commonly known as (street address, ci	
WHEREAS, the Agreement has not been previously amended except (if there are no	exceptions write "NONE"):
WHEREAS, the Parties have opened an escrow with("Escrow Holder");
WHEREAS, the Parties now desire to amend the Agreement and their instructions to	e Escrow Holder ("Escrow Instructions");
NOW, THEREFORE, for Buyer's payment of ten dollars (\$10) to Seller and other good acknowledged, the Parties hereby amend the Agreement and the Escrow Instructio	
The defined terms of the Agreement shall have the same meaning in this Amendment or similar technology shall be legal and binding. This Amendment may be executed together shall constitute one and the same instrument. This Amendment shall not as a whole, as if both Parties had prepared it. Except as specifically amended by this and continue in full force and effect.	in counterparts, each of which shall be deemed an original, and all of which be construed as if prepared by one of the Parties, but according to its fair meaning
ATTENTION: NO REPRESENTATION OR RECOMMENDATION IS MADE BY AIR CRE O CONSEQUENCES OF THIS AMENDMENT. THE PARTIES ARE URGED TO SEEK ADVICE AMENDMENT.	
EXECUTED as of the date first written above. By Seller:	By Buyer:
Ву:	Ву:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax: Email:	Fax: Email:
Ву:	By:
Name Printed:	Name Printed:
Title:	Title:
Phone:	Phone:
Fax:	Fax:
Email:	Email:
Address:	Address:
Federal ID No.:	Federal ID No.:
AIR CRE * https://www.aircra.com *	212-687-8777 * contracts@aircra.com

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS

© 2020 AIR CRE All Rights Reserved



SELLER'S MANDATORY DISCLOSURE STATEMENT

(DO NOT USE IN SALE OF RESIDENTIAL PROPERTIES WITH 1-4 UNITS)

This Seller's Mandatory Disclosure Statement ("Disclosure") is made pursuant to the	e requirements of and shall be a part of the STANDARD OFFER, AG	REEMENT
AND ESCROW INSTRUCTIONS FOR PURCHASE OF REAL ESTATE (See paragraph 9.1(a)	·	
purposes as of, regarding that certain real property commonly known is the Seller and is the Buyer. Note: This Disclosure does	in as (street address, city, state, zip): (" Property ") wherein ses not replace the Property Information Sheet published by AIR CRE ("A	
unless not required by the Purchase Agreement, Seller should also provide Buyer th		
1. NATURAL HAZARDS DISCLOSURES. To make the disclosures required by Califor Government Code Section 8589.4 (dam failure inundation area), California Government cone), California Public Resources Code Sections 4125, 4142 and 4291 (wildlands fire California Public Resources Code Section 2696 (seismic hazard zone) ("Natural Hazar Report") prepared by:	ment Code Sections 51178 or 51179, and 51182 (very high fire hazard se re area), California Public Resources Code Section 2622 (earthquake fau	everity ılt zone), and
myNHD Inc., 800-814-2922, https://www.myNHD.com., or	,, and the Natural Hazards Report is atta	ached.
2. EARTHQUAKE SAFETY. If the improvements on the Property were constructed or reinforced masonry walls together with wood frame floors or roofs, or (ii) unreinf Government Code §§8875.6, 8875.9, 8893.2, and 8893.3; and California Civil Code § Owner's Guide to Earthquake Safety (the "Booklet") published by the California Seis	forced masonry walls, then California Business & Professions Code §10: §2079.9 require Seller to provide Buyer a copy of The Commercial Prop	147;
https://ssc.ca.gov/forms_pubs/cog. Seller is not required to provide the Bookle Seller is hereby providing to Buyer.	let to Buyer. Seller is required to provide the Booklet to Buyer, whi	ich Booklet
3. FLOOD DISASTER INSURANCE. If Seller or Seller's predecessor-in-interest has perconditioned upon obtaining and maintaining flood insurance on the Property, Federathat they will be required to obtain and maintain such insurance on the Property and by a flood disaster, the purchaser may be required to reimburse the Federal Government.	ral law, i.e., 42 U.S.C. §5154a, mandates that prospective purchasers be not that if such insurance is not maintained and the Property is thereafte	e advised er damaged
Seller's knowledge, Federal flood disaster assistance has or has not been put has been received, the law specifies that the required notice be "contained in document of the contained in docu	previously received with regard to the Property. Note: if such disaster as ments evidencing the transfer of ownership".	ssistance
4. WATER HEATER BRACING. The Property does or does not contain or or more water heaters of 120 gallons or less, then as required by California Health a have or have not been braced, strapped and/or anchored in accordance wi		
5. TITLE INSURANCE. If the Purchase Agreement does not provide that Buyer wil title insurance. In accordance with California Civil Code §1057.6, Buyer is advised as	ill be provided with title insurance, Buyer is strongly urged to consider p	ourchasing
IMPORTANT: IN A PURCHASE OR EXCHANGE OF REAL PROPERTY, IT MAY BE ADVISAL SINCE THERE MAY BE PRIOR RECORDED LIENS AND ENCUMBRANCES WHICH AFFECT INSURANCE SHOULD BE OBTAINED IN ORDER TO ENSURE YOUR INTEREST IN THE PR	T YOUR INTEREST IN THE PROPERTY BEING ACQUIRED. A NEW POLICY	
6. METHAMPHETAMINE LABORATORY ACTIVITY. As required by California Health	n and Safety Code §25400.28, Seller notifies Buyer that a government o	rder
identifying the Property as contaminated by methamphetamine laboratory activity been received by Seller.	has been received by Seller (copy of such order is attached), or	has not
7. HAZARDOUS SUBSTANCES. If the Property is non-residential property, then as	s required by California Health and Safety Code §25359.7 Seller notifies	Buyer that:
Seller neither knows nor has reasonable cause to believe that any release of ha	azardous substance (as such terms are defined by California Health and	Safety Code
§§25320 and 25316) has come to be located on or beneath the Property, or Se substance has come to be located on or beneath the Property.	eller knows or has reasonable cause to believe that a release of hazardo	ous
8. WATER CONSERVING PLUMBING FIXTURES. For multi-family residential (conta California Civil Code §1101.5 requires all noncompliant plumbing fixtures to be repla	• • • • • • • • • • • • • • • • • • • •	
by California Civil Code §1101.5(e), Seller discloses that the Property was built a	after January 1, 1994 and is exempt from this statute; U does include	е
noncompliant plumbing fixtures; or — does not include any noncompliant plumbi	ing fixtures.	
9. DEATH ON THE PROPERTY. As required by California Civil Code §1710.2, Seller the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer to purchase the Property in the three years before the date of Buyer's offer the purchase the p		cupant of
10. OTHER.		
Seller believes the information provided in this Disclosure by Seller is true and corre that Seller has exercised good faith in the selection of a Natural Hazards Expert, Sell-information contained in the Natural Hazards Report, and makes no representation Hazards Report. Buyer agrees the Natural Hazards Report satisfies Seller's legal duty Report shall be deemed to be an expert dealing with matters within the scope of its Natural Hazards Disclosures, and California Civil Code §1103.4 which deals with a se knowledge shall be deemed to apply to Seller.	ler has not independently verified, and Seller is not obligated to verify, to warranty as to the truth or accuracy of any information contained in ty to make the Natural Hazards Disclosures, the provider of the Natural sexpertise with respect to the examination and written report regarding	the n the Natura Hazards g the
The hazards referenced in this Disclosure may limit the ability of the owner of the Pradisaster. The maps on which the Natural Hazards Report are based estimate when property will be affected by a natural disaster. Buyer is advised to obtain profession	re natural hazards exist and are not definitive indicators of whether or r	not a
Applicable laws and the areas covered by the various natural hazard zones, etc. may the Natural Hazards Report becomes inaccurate as a result of any governmental acti has actual knowledge that the information has become inaccurate. Prior to the clos is advised to verify with appropriate local governmental agencies, such as the city ar established natural hazard zones in addition to those listed above. Signatures to this shall be legal and binding. This Disclosure may be executed in counterparts.	tion, map revision, changed information, or other act or occurrence, unl se of escrow, Buyer may wish to again check the status of the Property. and/or county in which the Property is located, to determine if they have	less Seller Also, Buyer e
The statements herein are not intended to be full or complete disclosure of all post the Property, Buyer is advised to: 1. Review the entirety of all laws applicable to such disclosed matter.	ssible impacts on Buyer and/or the Property. If a matter is disclosed r	egarding
INITIALS	INITIALS	
© 2017 AIR CRE. All Rights Reserved.	Last Edited: 9/30/2022 2	2:03 PM

SMD-7.01, Revised 01-18-2022

- 2. Seek advice of counsel as to the legal consequence of any disclosed matters.
- 3. Retain appropriate consultants to review and investigate the impact of all such disclosed matters.

Neither AIR CRE nor any of the brokers make any representation, warranty or recommendation regarding the legal sufficiency, legal effect or any other consequences of this Disclosure.

Date:			
SELLER			
Ву:			
Name Printed:			
Title:			

Buyer acknowledges receipt of this Disclosure together with the Natural Hazards Report, the Booklet and any other documents referenced herein (as applicable) as being provided to Buyer, and agrees to the above terms and conditions.

Date: _____
BUYER

By: _____
Name Printed: _____
Title: ____

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2017 AIR CRE. All Rights Reserved.
SMD-7.01, Revised 01-18-2022



PROPERTY INFORMATION SHEET

(For the sale or leasing of non-residential properties)

PREFACE:

Purpose: This Statement is NOT a warranty as to the actual condition of the Property/Premises. The purpose is, instead, to provide the brokers and the potential buyer/lessee with important information about the Property/Premises which is currently in the actual knowledge of the Owner and which the Owner is required by law to disclose.

Actual Knowledge: For purposes of this Statement the phrase 'actual knowledge' means: the awareness of a fact, or the awareness of sufficient information and circumstances so as to cause one to believe that a certain situation or condition probably exists. TO WHOM IT MAY CONCERN: ("Owner"), owns the Property/Premises commonly known as (street address, city, state, zip) located in the County of generally described as (describe briefly the nature of the Premises or Property) (hereinafter "Property"), and certifies that: Material Physical Defects. Owner has no actual knowledge of any material physical defects in the Property or any improvements and structures thereon, including, but not limited to the roof, except (if there are no exceptions write "NONE"): Equipment. Owner has no actual knowledge that the heating, ventilating, air conditioning, plumbing, loading doors, electrical and lighting systems, life safety systems, security systems and mechanical equipment existing on the Property as of the date hereof, if any, are not in good operating order and condition, except (if there are no exceptions write "NONE"): Owner has no actual knowledge of any leases, financing agreements, liens or other agreements affecting any equipment which is being included with the Property, except (if there are no exceptions write "NONE"): Soil Conditions. Owner has no actual knowledge that the Property has any slipping, sliding, settling, flooding, ponding or any other grading, drainage or soil problems, except (if there are no exceptions write "NONE"): **Utilities.** Owner represents and warrants that the Property is served by the following utilities (check the appropriate boxes) public sewer system and the cost of installation thereof has been fully paid, \square private septic system, \square electricity, \square natural gas, \square domestic water, telephone, and U other: Insurance. Owner has no actual knowledge of any insurance claims filed regarding the Property during the preceding 3 years, except (if there are no exceptions write "NONE"): Compliance With Laws. Owner has no actual knowledge of any aspect or condition of the Property which violates applicable laws, rules, regulations, codes, or covenants, conditions or restrictions, or of improvements or alterations made to the Property without a permit where one was required, or of any unfulfilled order or directive of any applicable government agency or of any casualty insurance company that any work of investigation, remediation, repair, maintenance or improvement is to be performed on the Property, except (if there are no exceptions write "NONE"): Hazardous Substances and Mold. A. Owner has no actual knowledge of the Property ever having been used as a waste dump, of the past or present existence of any above or below ground storage tanks on the Property, or of the current existence on the Property of asbestos, transformers containing PCB's or any hazardous, toxic or infectious substance whose nature and/or quantity of existence, use, manufacture or effect, render it subject to Federal, state or local regulation, investigation, remediation or removal as potentially injurious to public health or welfare, except (if there are no exceptions write "NONE"): Owner represents and warrants that it is not currently, and never has been engaged in the business of hauling waste, and never stored hazardous substances on the Property, except (if there are no exceptions write "NONE").

C. Owner has no actual knowledge of the existence on the Property of hazardous levels of any mold or fungi defined as toxic under applicable state or Federal law, except (if there are no exceptions write "NONE"); Fire Damage. Owner has no actual knowledge of any structure on the Property having suffered material fire damage, except (if there are no exceptions write "NONE"): Actions, Suits or Proceedings. Owner has no actual knowledge that any actions, suits or proceedings are pending or threatened before any court, arbitration tribunal, governmental department, commission, board, bureau, agency or instrumentality that would affect the Property or the right or ability of an owner or tenant to convey, occupy or utilize the Property, except (if there are no exceptions write "NONE"): Owner has not served any Notices of Default on any of the tenants of the Property which have not been resolved except (if there are no exceptions write "NONE"): 10. Governmental Proceedings. Owner has no actual knowledge of any existing or contemplated condemnation, environmental, zoning, redevelopment agency plan or other land use regulation proceedings which could detrimentally affect the value, use and operation of the Property, except (if there are no exceptions write "NONE"): 11. Unrecorded Title Matters. Owner has no actual knowledge of any encumbrances, covenants, conditions, restrictions, easements, licenses, liens, charges or other matters which affect the title of the Property that are not recorded in the official records of the county recorder where the Property is located, except (if there are no exceptions write "NONE"): Leases. Owner has no actual knowledge of any leases, subleases or other tenancy agreements affecting the Property, except (if there are no exceptions write "NONF"): 13. Options. Owner has no actual knowledge of any options to purchase, rights of first refusal, rights of first offer or other similar agreements affecting the Property, except (if there are no exceptions write "NONE"): 14. Short Sale/Foreclosure. The ability of the Owner to complete a sale of the Property is contingent is not contingent upon obtaining the consent of one or more lenders to conduct a 'short sale', ie. a sale for less that the amount owing on the Property. (This paragraph only needs to be completed if this Property Information Sheet is being completed in connection with the proposed sale of the Property) One or more of any loans secured by the Property is is is not in 15. **Energy Efficiency**. The Property has has not been granted an energy efficiency rating or certification such as one from the U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED) or Seller/Lessor does not know if the Property has been granted such a rating or certificate. If such a rating or certification has been obtained please describe the rating or certification and provide the name of the organization that granted it: 16. Other. (It will be presumed that there are no additional items which warrant disclosure unless they are set forth herein):

INITIALS

INITIALS
© 2017 AIR CRE. All Rights Reserved.
PI-8.03, Revised 10-22-2020

The statements herein will be relied upon by brokers, buyers, lessees, lenders and others. Therefore, Owner and/or the Owner's Property Manager has reviewed and modified this printed statement as necessary to accurately and completely state all the known material facts concerning the Property. To the extent such modifications are not made, this statement may be relied upon as printed. This statement, however, shall not relieve a buyer or lessee of responsibility for independent investigation of the Property. Owner agrees to promptly notify, in writing, all appropriate parties of any material changes which may occur in the statements contained herein from the date this statement is signed until title to the Property is transferred, or the lease is executed. Signatures to this Statement accomplished by means of electronic signature or similar technology shall be legal and binding.

Date: (fill in date of execution)	OWNER
	By: Name Printed: Title:
Buyer/lessee hereby acknowledges receipt of a copy of this Property Information	Sheet on (Fill in date received) BUYER/LESSEE
	By: Name Printed: Title:
	213-687-8777 * contracts@aircre.com duced in any form without permission in writing.
QY.	



LEAD-BASED PAINT HAZARDS ADDENDUM TO STANDARD OFFER, AGREEMENT, AND ESCROW INSTRUCTIONS FOR PURCHASE OF RESIDENTIAL INCOME PROPERTY

(For residential property built prior to 1978)

Date:	
By and Between	
Seller:	
Buyer:	
Property Address:	
	(street address, city, state, zip)
aragraph:	
his Addendum is attached to and mad greement, this Addendum shall contr	e part of the above-referenced Agreement. In the event of any conflict between the provisions of this Addendum and the ol.
the Property is residential property b	uilt prior to 1978, then as required by federal law (42 USC §4852d):
. LEAD WARNING STATEMENT. Th	e following is included in the Agreement:
nay present exposure to lead from hildren may produce permanent i mpaired memory. Lead poisoning o provide the buyer with any infor	residential real property on which a residential dwelling was built prior to 1978 is notified that such property in lead-based paint that may place young children at risk of developing lead poisoning. Lead poisoning in young neurological damage, including learning disabilities, reduced intelligence quotient, behavioral problems, and also poses a particular risk to pregnant women. The seller of any interest in residential real property is required mation on lead-based paint hazards from risk assessments or inspections in the seller's possession and notify the int hazards. A risk assessment or inspection for possible lead-based paint hazards is recommended prior to
. SELLER'S DISCLOSURES.	
(a) Seller must provide Buyer a	lead hazard information pamphlet, as prescribed by the Administrator of the Environmental Protection Agency. (This pamphlet
an be obtained via the following link:	https://www.epa.gov/lead/protect-your-family-lead-your-home.) Seller has provided to Buyer this pamphlet (or an in the state where the Property is located).
(b) Seller discloses that Seller hoo exceptions, write "NONE"):	as no knowledge of the presence in the Property of any lead-based paint or any lead-based paint hazards, except (if there are
(c) Seller discloses that Seller h xceptions, write "NONE"):	as no lead hazard evaluation report available to Seller, except the following which Seller has provided Buyer (if there are no
	t Buyer a 10-day period (unless the parties mutually agree upon a different period of time) to conduct a risk assessment or ed paint and/or lead-based paint hazards.
. BUYER'S ACKNOWLEDGMENTS.	Buyer certifies that Buyer has:
(a) Read the Lead Warning Stat	ement above and understands its contents;
(b) Received from Seller a lead	hazard information pamphlet described above (or an equivalent pamphlet approved for use in the state where the Property is n reports, if any, described above; and
(c) Either (<i>check the applicable</i>	box):
i. A 10-day opportu	nity (unless the parties mutually agreed upon a different period of time) before becoming obligated under the Agreement to assessment or inspection for the presence of lead-based paint and/or lead-based paint hazards; or
ii. Waived the oppo	rtunity to conduct a risk assessment or inspection for the presence of lead-based paint and/or lead-based paint hazards.
NOT	AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com ICE: No part of these works may be reproduced in any form without permission in writing.
	ice. No part of these works may be reproduced in any form without permission in writing.

INITIALS

INITIALS
© 2022 AIR CRE. All Rights Reserved.
PSADD-1.00, Revised 10-13-2022



STANDARD ESTOPPEL CERTIFICATE - BY LESSEE

TO W	HOM IT MAY CONCERN:				
(stree		("Premises"	as Lessor, and "), which Lease was amendedare specified above).		
Lesse	e hereby certifies as follows:				
amer	ndments and guarantees.) Other	than the documents included	he guarantees, if any, are attached herel I in Exhibit 1 there are no oral or writter ons, write "NONE")	•	• •
2.	The Lease term commenced or	and expires or	·		
3.	The current monthly rent and e	expense pass-through, if any, a	re as follows:		
		Amount	Day of Month Due	Paid Up Through	<u>Year</u>
No re	Through ints or pass-throughs have been above.)	prepaid except as reflected in	n the Lease. (It will be presumed that no	o expense pass-throughs are curr	ently required unless set
4.	The current amount of security	deposit held by Lessor is		7	
			s execution, except as hereinabove iden no exceptions, write "NONE"):		nd effect and contains the
6. of an	The improvements and space r inducement nature by Lessor h	equired to be provided by Less ave been fulfilled except (if th	sor have been furnished and completed ere are no exceptions, write "NONE"): _	in all respects to the satisfaction	of Lessee, and all promises
7.	Lessee has no knowledge of an	y uncured defaults by Lessor o	r Lessee under the Lease, except (if the	re are no exceptions, write "NON	IE"):
	There are no disputes between otions, write "NONE"):		the Lease, the Premises or the improve	ements therein or thereon, excep	ot (if there are no
	Lessee is in full and complete p	ossession of the Premises and	has not assigned or sublet any portion	of the Premises, except (if there	are no exceptions, write
	Lessee has no knowledge of an IE"):	y prior sale, transfer, assignme	ent or encumbrance of the Lessor's inter	rest in the Lease, except (if there	are no exceptions, write
been in cor	11. Lessee has made no alterations or additions to the Premises, except (if there are no exceptions, write "NONE"): If alterations or additions have been made by Lessee, Lessee represents that to the best of its knowledge, all such alterations and additions were done in accordance with the terms of the Lease and in compliance with all applicable laws, rules and regulations, except (if there are no exceptions, write "NONE"): 12. The guarantees of the Guarantors named above are still in full force and effect, except (if there are no exceptions, write "NONE"):				
	13. Lessee is not currently the subject of a bankruptcy proceeding and to the best of its knowledge neither Lessor nor any Guarantor is involved in such a proceeding, except (if there are no exceptions, write "NONE"):				
as ne state	14. Lessee is aware that buyers, lenders and others will rely upon the statements made in this Estoppel Certificate, and has therefore adjusted the language hereof as necessary to make it an accurate statement of the current facts concerning the Lease. If no such adjustments have been made, said parties may rely upon the statements in this form as printed. Signatures to this Certificate accomplished by means of electronic signature or similar technology shall be legal and binding. 15. Additional items (if there are no additional items, write "NONE"):				
DATE	:(Fill in date of execution)	<u> </u>	 By:		
	Name Printed: Title: Phone: Fax: Email:				
			By: Name Printed:		
			Title: Phone: Fax:		
			Email:		
			Address:	_	

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

ESTLE-3.02, Revised 10-22-2020



STANDARD ESTOPPEL CERTIFICATE - BY LESSOR

TO W	HOM IT MAY CONCERN:					
conc	ease ("Lease") dated erning the real property known ("Guarantor(s)") (i	as (street address, city, state, z	rip):	("Premises"), wh	ich Lease was amended	and guaranteed
In lig	ht of the fact that the Lessee ha	as failed to provide an Estoppel	Certificate, Lessor herek	y certifies as follows	:	
	True copies of the above referendments and guarantees.) Othere with respect to the Premises		d in Exhibit 1 there are n	o oral or written agre	, , , ,	•
2.	The Lease term commenced o	n and expires on	·			
3.	The current monthly rent and	expense pass-through, if any, a	re as follows:			
		<u>Amount</u>	Day of Month Due		Paid Up Through	<u>Year</u>
Rent						
No re	Through ents or pass-throughs have been above.)	n prepaid except as reflected in	the Lease. (It will be pro	esumed that no expe	ense pass-throughs are current	ly required unless set
4.	The current amount of security	y deposit held by Lessor is				
5. entir	The Lease has not been modifi e agreement between Lessor a	ied, orally or in writing, since its nd Lessee, except (if there are r				effect and contains the
6. Lesso	The improvements and space or have been fulfilled except (if	required to be provided by Less there are no exceptions, write	•	•	respects, and all promises of a	n inducement nature by
7.	Lessor has no knowledge of an	ny uncured defaults by Lessor o	r Lessee under the Lease	, except (if there are	no exceptions, write "NONE")	:
8. exce	There are no disputes between ptions, write "NONE"):	n Lessor and Lessee concerning	the Lease, the Premises	or the improvement	s therein or thereon, except (i	f there are no
	To the best of Lessor's knowled e are no exceptions, write "NON	-	ete possession of the Pre	mises and has not a	ssigned or sublet any portion o	of the Premises, except (if
	Lessor has no knowledge of ar NE"):	ny prior sale, transfer, assignme	nt or encumbrance of the	e Lessee's interest in	the Lease, except (if there are	e no exceptions, write
If alte	To the best of Lessor's knowled erations or additions have been the terms of the Lease and in c	made by Lessee, Lessor repres	ents that to the best of i	ts knowledge, all suc	ch alterations and additions we	ere done in accordance
12.	The guarantees of the Guarant	tors named above are still in ful	ll force and effect, except	(if there are no exce	eptions, write "NONE"):	
	Lessor is not currently the subjections, except (if there are no e		g and to the best of its kr	nowledge neither Les	ssee nor any Guarantor is invol	ved in such a
as ne	Lessor is aware that buyers, le cessary to make it an accurate ments in this form as printed.	statement of the current facts	concerning the Lease. If	no such adjustments	s have been made, said parties	may rely upon the
15.	The Lessee may be contacted a Mailing address: Telephone: Facsimile: Email:	at:				
16.	Additional items (if there are n	no additional items, write "NON	IE"):			
DATE	(fill in date of execution)			rinted:	-	
•						
	//					
) `		Email: _			
			Ву:			
	7		Name P	rinted:		
						
		AIR CRE * https://ww	w.aircre.com * 213-687	'-8777 * contracts@	Daircre.com	

AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com NOTICE: No part of these works may be reproduced in any form without permission in writing.

INITIALS
© 2017 AIR CRE. All Rights Reserved.



RELEASE OF FUNDS STANDARD SALE ADDENDUM

Date:	
By and Between	
Seller:	_
Buyer:	-
Property Address:	(street address, city, state, zip)
Paragraph:	
Release of Funds: Escrow Holder is authoriz without any further written or verbal instruc	ed and instructed to immediately release Buyer's opening deposit to Seller upon removal of Buyer's contingency perio tions from either party.
portion of the purchase price at the close of release of these funds is authorized prior to	e to Buyer for any reason other than a default by Seller under this Agreement. Said funds being released will apply as escrow and the Seller's demand for cash at close of escrow is reduced accordingly. The Buyer acknowledges that the the recording of any document as required under the Buyer's previous instructions, and regardless of the current or outcome of this escrow. Payment of the funds prior to the close of escrow as herein authorized is made without
Α	NR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com No part of these works may be reproduced in any form without permission in writing.



, by and between, as Lessor, and, as Lesse, concerning the real property commonly known as (street address, city, state, zip) (the "Premises"). Signatures to this Assignment accomplished by means of electronic signature or similar technology shall be legal and binding. 2. Assignor also assigns to Assignee the security deposit under said Lease in the sum of and the interest of Lessor in any applicable guaranty of said Lease. 3. This Assignment shall be effective from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Dated: ASSIGNOR By:	ASSIGNMENT AND ASSUMP	TION OF LESSOR'S INTEREST IN LEASE
(the "Premises"). Signatures to this Assignment accomplished by means of electronic signature or similar technology shall be legal and binding. 2. Assignor also assigns to Assignee the security deposit under said Lease in the sum of		
Lease. 3. This Assignment shall be effective from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Dated:	, by and between, as Lessor, and zip) (the "Premises"). Signatures to this Assignment accomp	, as Lessee, concerning the real property commonly known as (street address, city, state, lished by means of electronic signature or similar technology shall be legal and binding.
Dated:	2. Assignor also assigns to Assignee the security deposit under said Lease Lease.	e in the sum of and the interest of Lessor in any applicable guaranty of said
ASSIGNOR By: Name Printed: Title: Phone: By: Name Printed: Title: Phone: ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indehnifty, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By: Name Printed: Title: Phone:	3. This Assignment shall be effective from and after the recordation, in the Assignee.	he county where the Premises are located, of the deed passing fee title to the Premises to
ASSIGNOR By: Name Printed: Title: Phone: By: Name Printed: Title: Phone: ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indehnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By: Name Printed: Title: Phone: By: Name Printed: Title: Phone: By: Name Printed: Title: Phone:		Dated:
By: Name Printed: Title: Phone: By: Name Printed: Title: Phone: ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indefinity, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By: Name Printed: Title: Phone: By: Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		
Name Printed: Phone: By: Name Printed: Title: Phone: ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indemnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By: Name Printed: fitle: Phone: By: Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		Assidnon
Name Printed: Phone: By: Name Printed: Title: Phone: ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indemnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By: Name Printed: Fitle: Phone: By: Name Printed: Title: Phone: By: Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		
Phone: By: Name Printed: Title: Phone: ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee shall defend, indehnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By: Name Printed: fitle: Phone: By: Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		
By: Name Printed: Title: Phone: ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indemnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By: Name Printed: Title: Phone: Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		
Name Printed: Title: Phone: ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee shall defend, indemnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By:		
Title: Phone: ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indemnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By: Name Printed: Title: Phone: By: Name Printed: Title: Phone:		
ACCEPTANCE AND ASSUMPTION Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indemnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated:		
Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indemnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE Phone: Phone: Title: Title: Phone: Title: Phone: Title: Phone: Phone: Title: Phone: Title: Phone: Phone: Title: Phone: Phone: Title: Phone: Phone: Phone: Title: Phone: Phone:		Phone:
where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indemnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date. Dated: ASSIGNEE By:	ACCEPTA	NCE AND ASSUMPTION
ASSIGNEE By: Name Printed: Fitle:	Assignee hereby accepts the above Assignment and assumes all of the rights and obligations of the Lessor accruing from and after the recordation, in the county where the Premises are located, of the deed passing fee title to the Premises to Assignee. Assignee shall defend, indemnify, and hold Assignor harmless with respect to the obligations of Lessor under said Lease accruing from and after said date.	
By:		Dated:
Name Printed: Title: Phone: By: Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		ASSIGNEE
Name Printed: Title: Phone: By: Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		
Name Printed: Title: Phone: By: Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		Bv:
Phone: By: Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		Name Printed:
Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		
Name Printed: Title: Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com	()	Dur
Phone: AIR CRE * https://www.aircre.com * 213-687-8777 * contracts@aircre.com		Name Printed:
NOTICE: No part of these works may be reproduced in any form without permission in writing.		.com * 213-687-8777 * contracts@aircre.com
	NOTICE: No part of these works may be	e reproduced in any form without permission in writing.
	/ \	
	</th <th></th>	
	K X	
·	*	

INITIALS

INITIALS © 2017 AIR CRE. All Rights Reserved. AA-2.03, Revised 10-22-2020